STATE OF VERMONT GREEN MOUNTAIN CARE BOARD

IN RE: PROPOSED PURCHASE OF)	GMCB 014-17con
BIRCHWOOD TERRACE)	

APPLICANTS' RESPONSES TO THE GREEN MOUNTAIN CARE BOARD'S DECEMBER 8, 2017 FIRST SET OF REQUESTS FOR INFORMATION

NOW COME the Applicants and submit the following responses to the December 8, 2017 requests for information from the Green Mountain Care Board.

1. Confirm whether any of the three proposed buyers have other personal, professional, non-health care or health care interests under any other personal name or business name not reflected on the Personal Financial Statements submitted with the application. If so, revise and resubmit the Personal Financial Statements to reflect these interests and associated assets and liabilities.

Response:

Except for the clarification set forth in #2 below to Milton Ostreicher's Personal Financial Statement, no other changes are necessary - as the Personal Financial Statements were otherwise accurate. None of the three proposed individual buyers has other personal, professional, non-health care or health care interests under any other personal name or business name, except as reflected on the Personal Financial Statements submitted with the application, or as hereby amended.

2. Revise and resubmit Mr. Ostreicher's Personal Financial Statement to reflect the assets and liabilities associated with the other facilities in which he has interests. Provide separate Verifications Under Oath from each of the three buyers to accompany their Personal Financial Statements.

Response:

Applicants will be submitting a revised <u>Attachment 22</u> under separate cover as a confidential document. This is Mr. Ostreicher's updated Personal Financial Statement with an accompanying verification for the same. Verifications Under Oath will be submitted with that same submission for the Personal Financial Statements of Messrs. Rubin and Erlichman.

3. Provide the annual fee the current owner of Birchwood Terrace pays to Ventas Realty, LP to lease the property on which the facility is built, the annual fee the new owners will pay, and the terms of the assignment of the leasehold interest. Explain whether and where these costs are included in the total project cost in Tab 11, Table 1; if they are not included, explain and revise the table, if needed. Also verify whether the annual fee is included in the projected financials in Tab 35.

Response:

The current owner of the real property (under a long term leasehold interest) is KND Real Estate Escrow Holdings, LLC., a Delaware limited liability company ("KND") and an affiliate of Kindred Nursing Centers East LLC, the current operator of the Facility. The land under the Facility is divided into three parcels that are addressed in two leases. The duration and rental terms of the leases are as follows:

Ground Lease Term:

Lease A (two parcels): commenced on July 5, 1963 and expires on June 30, 2062 Lease B (one parcel): commenced on July 5, 1963 and expires on June 30, 2062

Ground Lease Rent:

Lease A: \$1,800.00 per annum payable in one installment on January 15th of each year **Lease B:** \$200.00 per annum payable in one installment on January 15th of each year

The Applicants have been informed that, on December 21, 2017, Ventas L.P., the previous owner of the leasehold interest, assigned its rights under the Facility's ground lease to KND. KND currently leases the property to Kindred Nursing Centers East LLC which is the current licensed operator of the Facility. Except for the change to the lessee entity (Ventas L.P. to KND), there were no other changes to the terms of the current leases. At closing, KND will assign its leasehold interests to the Applicants who will pay the same annual rent for the duration of the leasehold terms. **Attachment 11**, Table 1 and **Attachment 35** have been updated to reflect these rent payments.

The Applicants are in negotiations with the ground lessor for an extension of the term of the ground lease, but have not finalized the same. If and when such terms are finalized, the Applicants will update the Board.

4. Explain whether any of the three proposed buyers also manage or have management agreements with any other facilities. If yes, provide the name and address of each facility.

Response:

The proposed buyers do not manage or have management agreements with any other facilities except those indicated in the Revised Certificate of Need Narrative.

5. Explain what funding options, other than the financing by HHC Finance at a 7.24% interest rate, were explored by the applicants and why they were not selected.

Response:

The Applicants reached out to numerous financing sources before selecting their term sheet, as discussed below. However, as the Applicants will only own a leasehold interest, as the Facility leases the land via a ground lease, many banks are hesitant to provide financing. Additionally, the relatively small size of the loan limited the financing options as well. The Applicants will continue to explore the potential for more favorable financing options and will inform the GMCB if they are successful.

As set forth in the Revised Certificate of Need Narrative, the Applicants have recently secured a new term sheet from Customers Bank ("Customers"), submitted herewith and as <u>Attachment 34</u> to the Revised Certificate of Need Narrative. Customers has agreed to finance the proposed transaction with an interest rate of LIBOR plus 3.25%, which, at the time of the projections, equated to 4.81%.

6. Identify the contingency plan if HHC Finance does not provide the loan you are seeking. Fully explain the financial stability of the facility if the applicants cannot obtain HUD financing in five years when payments balloon.

Response:

If Customers (the Applicants have secured a new term sheet from Customers) does not fund the loan at closing and the Applicants are not able to procure an alternative lender, the Applicants will fund the purchase price with cash. Although the Applicants intend to refinance their loan via HUD financing, the Applicants will obtain a traditional loan if necessary. Due to the relatively small size of the loan in comparison to the projected income of the Facility, the Applicants do not anticipate this being an issue. Moreover, the projected income of the Facility can easily fund the debt service on either type of loan.

7. Provide the documents withheld in Attachments 32 (Operations Transfer Agreement) and 33 (Amended and Restated Assignment and Assumption Agreement). The Board's review of the project cannot be completed without these documents.

Response:

Copies of the Operations Transfer Agreement and the Amended and Restated Assignment and Assumption Agreement are submitted herewith and as <u>Attachments 32 and 33</u>, respectively, to the Revised Certificate of Need Application.

8. Provide a readable e-copy and readable hard copy of the 2017 actuals in Attachment 15.

Response:

The Applicants have replaced Attachments 13, 14 and 15, with audited financials for the years in question.

9. Revise Attachments 25-28 to cover the most recent 6-month period for Achieve Rehabilitation and Nursing Center (Achieve), Beacon Nursing and Rehabilitation (Beacon), Highland Care Center (Highland) and Birchwood.

Response:

Attachments 25, 26, 27, and 28 are submitted herewith and have been revised to include the most recent 6-month period for Achieve Rehabilitation and Nursing Center (Achieve), Beacon Nursing and Rehabilitation (Beacon), Highland Care Center (Highland), and Birchwood.

10. Explain the reasons for low CMS star ratings for Overall Staffing and RN Staffing at Achieve, Beacon and Highland.

Response:

Importantly, and as set forth in the Revised CON Narrative, the Applicants do not anticipate making any material staffing changes at the Facility. The Applicants are committed to raising the overall quality of care at the Facility and, after consultation with the current administrator, feel that the current staffing levels are appropriate for the resident population at the facility.

In regards to the star ratings for staffing at Achieve, Beacon and Highland, the Applicants do not believe that this category properly reflects the quality of care that is delivered at these facilities. As set forth in the star ratings data, these facilities score between 3 and 5 stars for Health Inspection and Quality of Care; clearly indicating a sufficient level of staffing to ensure the delivery of stellar care to their respective resident populations. Moreover and as indicated through their surveys, these facilities have had excellent annual surveys including deficiency free surveys.

Instead, these facilities might have received lower staffing scores due to the fact that they service a higher acuity resident population than most facilities; leading to a less accurate analysis due to CMS's current scoring methodology for this category. The Applicants also believe that the low scores may be due to the fact that the staffing ratings are currently based on self-reported data not always accurately reported by some facilities as opposed to the Applicants' related facilities which always reported their data accurately. Indeed, CMS has announced plans to modify their current methodology and reporting structure partially in response to the foregoing.

- 11. Provide detailed information, implementation timeline and costs (including the costs and whether and where such costs are included in budget and staffing projections), for each of the following facility improvements and program expansions:
 - a. expanded programs for behavioral health, renal failure and parenteral nutrition;
 - b. expanded on-site psychological services;
 - c. more aggressive program for the Alzheimer Unit; and
 - d. revised admission policy to accept residents now ineligible for admission.

Response:

As mentioned in the CON narrative, our goal is to do whatever is necessary to raise the quality of care in the Facility leading to positive outcomes and satisfied residents and family members. This may include the implementation of new initiatives (like those mentioned in a-d above) and the hiring of additional and appropriate qualified staff to support the same. However, because we are not yet at the Facility, we have not been able to conduct a proper analysis of which of these are necessary and feasible to implement at the Facility. Accordingly, we did not include the cost of implementation of these services in our financial projections. Still, based on our current projections, the Facility profits should allow us to roll out these and/or other initiatives at the Facility.

Based on preliminary consultations with the Facility, we have provided below some more detail on the above-mentioned initiatives and why we specifically included these as potential areas we focused on for the Facility.

<u>Behavioral/Mental Health</u>: This is the third most significant diagnosis in the facility. As such, we need to ensure that the Facility is able to provide these residents with the highest level of care. If necessary and feasible, we will expand the current program to include additional onsite services and/or tele-medicine.

Renal: The Facility currently services a high volume of dialysis patients (approx. 6-8), despite the cost of transportation to off-site dialysis centers (\$70-\$90/day). We will explore the possibility of peritoneal dialysis. However, we will need to first ensure that all regulatory requirements, staff education, and new policies and procedures are in place prior to implementation.

<u>Total Parenteral Nutrition ("TPN")</u>: Birchwood is one of only two facilities in Vermont that accept TPN patients. The Facility, however, can only admit TPN patients who are fairly stable, as its pharmacy can only service labs twice a week (in the event formula changes are required). The Applicants will meet with the facility's pharmacy to explore the potential to accommodate additional TPN patients.

Expansive Programming for the Alzheimer's Unit: The Applicants hope to create a sensory space with better suited furniture and tactile stimulation.

<u>Revised Admission Policy</u>: We will explore whether it is feasible for the facility to admit bariatric residents. We will procure the necessary equipment and ensure the facility has adequate staff to meet the needs of this population prior to admission.

12. Explain how the applicants will achieve these facility improvements and program expansions without adding additional staff and/or training in projected years 1-3, as shown in Attachment 11, Table 9.

Response:

As clarified above, the above-mentioned improvements and program expansions are areas the Applicants have begun to focus on after consultation with the Facility but because they are not operating the Facility they are not able to complete a proper analysis of its feasibility and cannot commit to its implementation. As such, they did not include the costs for the foregoing in their projections.

It should be noted, however, that the costs of some of these initiatives are limited where there is no new equipment to purchase and the only expense is the education and training of staff. Moreover, some initiatives will even save the Facility money (on-site dialysis versus off-site will save the facility the cost of daily transportation for these residents).

13. Birchwood Terrace falls below the state average in 12 of 24 CMS quality measures. See Attachment 29. In a table format, identify the quality measures that fall below the state average and explain how the applicants will seek to improve each, and the timetable for improvement.

Response:

See Attachment 39.

14. Provide the three most recent surveys conducted for Birchwood Terrace and the Highland, Achieve and Beacon facilities.

Response:

The three most recent surveys conducted for Birchwood Terrace and the Highland, Achieve and Beacon facilities are submitted as follows:

Birchwood, <u>Attachment 40</u>; Highland, <u>Attachment 41</u>; Achieve, <u>Attachment 42</u>; and Beacon, <u>Attachment 43</u>. 15. Isaac Rubin's resume states that he "excels at leading 'troubled' facilities to excellence." Provide specific information regarding his leadership experience at other facilities, his title and the time period in which he was involved with each facility, and the facility's name and location.

Response:

Isaac Rubin is the Vice President of Business Development and Managed Care Initiatives as well as Corporate Director of the Delivery System Reform Incentive Payment (DSRIP) Program for a leading post-acute care provider in the Northeast. In this capacity, Mr. Rubin is at the forefront of the rapidly evolving healthcare landscape and the national trends towards quality and value. Mr. Rubin is driven by the belief that high quality care is not only a provider's duty, but also an opportunity. Accordingly, Rubin appreciates the challenges of preparing a skilled nursing facility for the post-fee-for-service environment. While quality is by definition 'qualitative,' he understands that outcomes must be benchmarked against his peers, and he will accept nothing short of excellence in this regard.

Rubin has leveraged high quality care to secure innovative partnerships across the care continuum with hospitals, physicians and managed care plans. He sits at the forefront of healthcare reform initiatives involving Accountable Care Organizations and bundled payment programs; always leading the discussion with documented outcomes and resident satisfaction surveys. These initiatives have driven referrals and validated his approach.

Rubin excels at leading 'troubled' facilities to excellence and has directed historically 'one star' providers to market leading status. Indeed, many of the regional administrators from Centers Health Care report to Rubin. He is a fixture 'on the floors' of the facilities in his charge, and epitomizes the qualities required to succeed in the new world order of healthcare reform. He has successfully implemented value base programs such as the BPCI Model 2, New York State's DSRIP (Delivery System Redesign Incentive Payment) program, and CMS's "Initiative to Reduce Avoidable Hospitalizations among Nursing Facility Residents". These programs have helped reduce unnecessary hospitalizations and led to a better quality of life. Additionally, while acting in his role at Centers Health Care from 2013 to the present, Rubin helped raise the overall census at Centers Health Care affiliated facilities from approximately 92% to 94% and the overall return to hospital rate decreased from 22% to 16%. A list of facilities that Rubin has been involved with since June 2013 is submitted as **Attachment 44**.

16. In a table format, provide the occupancy rates for the most recent 12-month period by month for Birchwood Terrace and for Highland, Achieve and Beacon.

Response:

The occupancy rates for the most recent 12-month period, by month, for Birchwood, Highland, Achieve, and Beacon are submitted in <u>Attachment 45</u>.

17. Provide a copy of the admission policy, as referred to on page 10 of the application.

Response:

The current admissions policy is submitted to the Revised Certificate of Need Narrative as **Attachment 46.**

18. Correct the error on page 21 that references "Silverbrook Corporation" and explain, relative to Birchwood Terrace, why no less expensive alternatives exist, would be unsatisfactory or are not feasible or appropriate for the proposed project.

Response:

The correction has been made to the Revised Certificate of Need Narrative.

19. Provide a letter from DAIL to satisfy HRAP Standards 5.2 and 5.3.

Response:

The Applicants hope to submit such letter in the next several weeks.

20. Explain why, relative to Tabs 16 and 17, there are no members of the LLC.

Response:

The members of Birchwood Operations LLC and Birchwood Prop LLC are as set forth in the Applicants' Revised CON Narrative. At the time the LLCs were incorporated, the respective LLC memberships were not finalized. The Vermont LLC statute does not require the Articles to disclose who the members of an LLC are, so the Applicants have not gone back to revise the Articles.

STATE OF VERMONT GREEN MOUNTAIN CARE BOARD

IN RE: PROPOSED PURCHASE OF)	GMCB 014-17con
BIRCHWOOD TERRACE)	

VERIFICATION UNDER OATH APPLICANTS' RESPONSES TO THE GREEN MOUNTAIN CARE BOARD'S DECEMBER 8, 2017 REQUESTS FOR INFORMATION

Ariel Erlichman, being duly sworn, states on oath as follows:

- 1. My name is Ariel Erlichman. I have reviewed the Applicants' Responses to the Green Mountain Care Board's December 8, 2017 Requests for Information (the "Submission").
- 2. Based on my personal knowledge, after diligent inquiry, I attest that the information contained in the Submission is true, accurate and complete, does not contain any untrue statement of a material fact, and does not omit to state a material fact necessary to make the statement made therein not misleading, except as specifically noted herein or as otherwise specifically noted in the Submission.
- 3. My personal knowledge of the truth, accuracy and completeness of the information contained in the Submission is based upon either my actual knowledge of the subject information or, where identified below, upon information reasonably believed by me to be reliable and provided to me by the individuals identified below who have certified that the information they have provided is true, accurate, and complete, does not contain any untrue statement of a material fact, and does not omit to state a material fact necessary to make the statement made therein not misleading.
- 4. The following certifying individuals have provided information or documents to me in connection with the Submission, and such individuals have certified, based on their actual knowledge of the subject information or, where specifically identified in such certification, based on information reasonably believed by the certifying individual to be reliable, that the information or documents provided are true, accurate and complete, do not contain any untrue statement of a material fact, and do not omit to state a material fact necessary to make the statement made therein not misleading:

Milton Ostreicher, BIRCHWOOD PROP LLC and BIRCHWOOD OPERATIONS LLC Isaac Rubin, BIRCHWOOD PROP LLC and BIRCHWOOD OPERATIONS LLC Andrew Bachand CPA, Tomas Depoy, Sharon Martin, Alecia DiMario LNHA.

5. In the event that the information contained in the Submission becomes untrue, inaccurate or incomplete in any material respect, I acknowledge my obligation to notify the Green Mountain Care Board, and to supplement the Submission, as soon as I know, or reasonably should know, that any information or document has become untrue, inaccurate or incomplete in any material respect.

STATE OF VERMONT GREEN MOUNTAIN CARE BOARD

IN RE: PROPOSED PURCHASE OF)	GMCB 014-17con
BIRCHWOOD TERRACE)	

<u>VERIFICATION UNDER OATH</u> REVISED CERTIFICATE OF NEED NARRATIVE

Ariel Erlichman, being duly sworn, states on oath as follows:

- 1. My name is Ariel Erlichman. I have reviewed the REVISED CERTIFICATE OF NEED NARRATIVE and attachments to be filed on January 29, 2018 (the "Submission").
- 2. Based on my personal knowledge, after diligent inquiry, I attest that the information contained in the Submission is true, accurate and complete, does not contain any untrue statement of a material fact, and does not omit to state a material fact necessary to make the statement made therein not misleading, except as specifically noted herein or as otherwise specifically noted in the Submission.
- 3. My personal knowledge of the truth, accuracy and completeness of the information contained in the Submission is based upon either my actual knowledge of the subject information or, where identified below, upon information reasonably believed by me to be reliable and provided to me by the individuals identified below who have certified that the information they have provided is true, accurate and complete, does not contain any untrue statement of a material fact, and does not omit to state a material fact necessary to make the statement made therein not misleading.
- 4. The following certifying individuals have provided information or documents to me in connection with the Submission, and such individuals have certified, based on their actual knowledge of the subject information or, where specifically identified in such certification, based on information reasonably believed by the certifying individual to be reliable, that the information or documents provided are true, accurate and complete, do not contain any untrue statement of a material fact, and do not omit to state a material fact necessary to make the statement made therein not misleading:

Milton Ostreicher, BIRCHWOOD PROP LLC and BIRCHWOOD OPERATIONS LLC Isaac Rubin, BIRCHWOOD PROP LLC and BIRCHWOOD OPERATIONS LLC Andrew Bachand CPA, Tomas Depoy, Sharon Martin, Alecia DiMario LNHA.

5. In the event that the information contained in the Submission becomes untrue, inaccurate or incomplete in any material respect, I acknowledge my obligation to notify the Green Mountain Care Board, and to supplement the Submission, as soon as I know, or reasonably should know, that any information or document has become untrue, inaccurate or incomplete in any material respect.

ATTACHMENT 11

Birchwood Operations, LLC and Birchwood Prop, LLC TABLE 1 PROJECT COSTS

Construction Costs		
New Construction	\$	
2. Renovation		-
3. Site Work		
Fixed Equipment		
Design/Bidding Contingency		
6. Construction Contingency		
Construction Manager Fee		
8. Other (please specify)		
Subtotal	_\$	- 3 ,⊻
Related Project Costs		
Major Moveable Equipment	\$	
2. Furnishings, Fixtures & Other Equip.		300,000
3. Architectural/Engineering Fees		
Land Acquisition		- 1
5. Purchase of Buildings		3,038,785
Administrative Expenses & Permits		75,000
7. Debt Financing Expenses (see below)		113,421
8 Debt Service Reserve Fund		- I
9. Working Capital		-
10. Other (please specify)		
Subtotal	\$	3,527,206
otal Project Costs	¢	3,527,206

Debt Financing Expenses		
Capital Interest	\$	
2. Bond Discount or Placement Fee		
3. Misc. Financing Fees & Exp. (issuance costs)		113,421
4. Other		
Subtotal	\$	113,421
Less Interest Earnings on Funds		
Debt Service Reserve Funds	\$	
Capitalized Interest Account		
3. Construction Fund		
4. Other	لسالح	
Subtotal	\$	
Total Debt Financing Expenses	\$	113,421
feeds to line 7 above		

Birchwood Operations, LLC and Birchwood Prop, LLC

TABLE 2

DEBT FINANCING ARRANGEMENT, SOURCES & USES OF FUNDS

4.8%		
	2023	
	\$	2,671,028
		856,178
		100
	\$	3,527,206
	4.8% Jan 2018 To: Dec 2	Jan 2018 To: Dec 2023

Uses of Funds		
Project Costs (feeds from Table 1)		
New Construction	\$	2
2. Renovation		-
3. Site Work		-
4 Fixed Equipment		9
Design/Bidding Contingency		~
6. Construction Contingency		-
Construction Manager Fee		
8. Major Moveable Equipment		-
Furnishings, Fixtures & Other Equip.	30	000,00
10. Architectural/Engineering Fees		-
11. Land Acquisition		-
12. Purchase of Buildings	3,03	88,785
Administrative Expenses & Permits	7	'5,000
14. Debt Financing Expenses	11	3,421
Debt Service Reserve Fund		-
16. Working Capital		*
17. Other (please specify)	-	
Total Uses of Funds	\$ 3,52	27,206

Total sources should equal total uses of funds.

Birchwood Operations, LLC and Birchwood Prop, LLC TABLE 6A REVENUE SOURCE PROJECTIONS WITHOUT PROJECT

							Р	roposed		P	roposed		P	roposed	
	Li	atest Actual	% of		Budget	% of		Year 1	% of		Year 2	% of		Year 3	% of
		2016	Total		2017	Total		2018	Total		2019	Total		2020	Tota
Gross Inpatient Revenue															
Medicare	\$	3,222,856	16.7%	5	3,287,313	16.7%			#DIV/01			#D V/O1			#DIV/Q
Medicaid		13,662,464	70.0%	\$	13,935,713	70.9%			#DIV/01			#DIV/01			WDIVA
Commercial		1,316,657	6.8%	\$	1,342,990	6.8%			#01V/01			#DIV/O'			#DIV/
Self Pay		1,141,650	5 9%	\$	1,164,483	5.9%			יכועום#			#DIVIO			#DIV
Free Care / Bad Debt		(64,825)	-0.3%	5	(66,122)	-0.3%			#DIV/D!			#CIV/01			DOIVE
Other		(- (,,	0.0%	5		0.0%			#DIV/01			#DIV/01			#DIVI
	s	19,278,802	100 0%	s	19,664,378	100 05%	\$	(6)	#01V/01	\$		#DIV/01	\$	22.5	#DIV/
Gross Outpatient Revenu	e														
Medicare	S		#DIV/O			#DIV/0			יוס/עוט#			#DIV/0!			#DIV/
Medicaid			#0:V/0"			#DIV/01			#DIV/01			#DIV/O			#DIV
Commercial			#DIV/01			#DIV/01			#DIV/0"			#DIV/0			#OIV
Self Pay			#DIV/O			#DIV/0			#DIV/01			#DIV/01			#DIV
Free Care / Bad Debt			#017/01			#D1V/01			#DIV/01			#DIV/01			#DIV
Other			#0/V/01			#DIV/O			#DIV/0!			#DIV/01			NOIV
Ottlei	<u>s</u>	*	#DIV/01	\$		#DIA/0,	s	*	#DIV/0!	S		#DIV/01	\$	390	#DIV/
Gross Other Revenue	-			_											
Medicare	S	328,442	79.2%	3	335,011	79.2%			#DIV/01	S	18	#DIVIO	S		#DIV/
Medicard	9	020,442	0.0%	5	-	0.0%			#OIV/0!	\$		#DIV/01	s		#DIV/
Commercial			19 3%	5	81,697	19 3%			#DIV/O!	\$	2	#DIV/O	\$		#DIV
		80,095								5			3		#DIV
Self Pay		5,904	1 4%	\$	6,022	1.4%			#DIV/0!	6.5		#DIV/01	S		
Free Care / Bad Debt			0.0%	S		0.0%			#DIV/O!	\$	1.1	#DIV/01			NDIA:
Other	_		0.0%	- \$		0.0%	- \$	-	#DIV/0!	_ S		#DIV/0!	\$		#DIV.
	\$	414,441	100 0%	S	422 730	100.0%	\$	*)	#DIV/0!	s	*	#DIV/01	\$	1.04	#DIV
Gross Patient Revenue															
Medicare	S	3,551,298	18 0%	\$	3,622,324	18 0%	\$	-	#DIV/0!	5	-	#DIV/01	\$	-	#DIV
Medicaid		13,662,464	69 4%		13,935,713	69 4%		-	#DIV/01		-	יט/עומי			#DIV
Commercial		1,396,752	7 1%		1,424,687	7 1%		*	#DIVIO!			#DIV/0!		-	#DIVI
Self Pay		1,147,554	5 8%		1,170,505	5.6%			#D,V/01			#DIV/01		-	#DIV
Free Care / Bad Debt		(64,825)	-0 3%		(66, 122)	-0 3%			#DIV/0!		-	#DIVIO!		-	#DIV
Other			0.0%		-	0.0%			#DIVIO:			#DIV/01			#DIV
	S	19,693,243	100 0%	\$	20,087,108	100 0%	\$	-	#DIV/0!	S		#DIV/0!	\$		HOIVE
Deductions from Revenue	е														
Medicare	5	(540,395)	-7.9%	\$	(551,203)	-7.9%			#DIV/01			NDIA\0			#DIV/
Medicaid		6,858,088	100 3%	\$	6,995,250	100 3%			#DIV/0!			#DIV/0			#DIV/
Commercial		487,951	7.1%	S	497,710	7 192			#DIV/01			#DIV/0'			#DIV/
Self Pay		30,662	0.4%	\$	31,275	0 4%			#DIV/0!			#DIV/0!			#DIV/
Free Care / Bad Debt			0.059	\$	- C -	0.0%			#DIV/01			#DIV/01			#DIV/
Other			0.0%	\$		0.0%		14	#DIV/0!			#DIV/01			#DIV
	S	6,835,306	100 0%	5	6,973,032	100 0%	\$	**	#DIV/0°	S	1.5	#DIV/CI	s	3*3	#DIV/
Net Patient Revenue															
Medicare	\$	4,091,693	318%	5	4,173,527	31.6%	\$	153	#DIV/01	\$	*5	#DIVIO'	S	850	HOIV
Medicaid		6,804,376	52.9%		6,940,464	52.9%			#DIV/01		ĕ	#DIV/01			#DIV
Commercial		908,801	7 1%		926,977	7 1%		lk:	#DIV/O!		: €	#DIV/01		9.5	#DIV
Self Pay		1,116,892	8 7%		1,139,230	9.7%		6	#DIV/O		63	#DIV/01		39.7	NOIV
Free Care / Bad Debt		(64,825)	-0.5%		(66, 122)	-0 5%			#DIV/01			#DIV/01			#DIV
Other		(04,020)	0.0%		(00,122)	0.0%		100	#DIV/0!		17.	#DIV/01		36	#DIV
DSP*			0.0%			0.0%		100	#DIV/0!			ימעוסיי			#DIV
			M M 52			V: V / 1								The second second second	

Latest actual numbers should tie to the hospital budget process

^{*} Disproportionate share payments

Birchwood Operations, LLC and Birchwood Prop, LLC TABLE 6B

REVENUE SOURCE PROJECTIONS
PROJECT ONLY

	Latest Actual	% of		udget	% of		Proposed Year 1 2018	% of Total		Proposed Year 2 2019	% of Total	I	Proposed Year 3 2020	% of
	2016	Total		2017	Total		2018	Total	_	2013	TOTAL	_	2020	1010
Gross Inpatient Revenue	111111111111111111111111111111111111111				D 51		2 650 924	41.00	\$	2,591,084	14.2%	\$	2,625,344	14.2%
Medicare			S	7.90	#DIV/01	\$	2,556,824	14.2%						
Medicaid	11/1/1991////				*D:0/0'	\$	13,344,963	74.3%	S	13,523,778	74.3%	S	13,702,593	74 39
Commercia				(40)	#DIA/01	\$	1,204,155	6.7%	5	1,220,290	6.7%	\$	1,236,425	G 7%
Self Pay	1////998/////			(*)	#D:V:01	\$	993,698	5 5%	\$	1,007,013	5.5%	\$	1,020,328	5.5%
Free Care / Bad Debt	1//////////////////////////////////////			(8)	NDIA/01	\$	(140,774)	-0 8%	\$	(143,524)	-0.8%	\$	(148,326)	-Q B5
Other	////868////				#DIV/O	\$	(*)	0.0%	\$	387	0.0%	\$		0.04
	1///88////.		5	*	#DIV/0	\$	17,958,867	100 0%	s	18,198,642	100 0%	\$	18,438,365	100 0
ross Outpatient Revenu	e													
Medicare	1////68////		5		#D(V/O)	\$		#DIVIO!	\$		#DIV:01	\$		#DIV
Medicaid	1////8/8/////			140	#DIV/O		0.0	#DIV/01			#DIV/01		14.1	NCIV.
Commercial	1//////////////////////////////////////				#DIV/D			#DIVIO		240	POIVIO			#DIV
Self Pay	1111186811111				#DIV/O			#DJV/DI			#DIV/O		(2)	WOIV
Free Care / Bad Debt	//////////////////////////////////////			1,5	WD:V/O			€DIV/3°			יפועום			#DIVI
								#DIV/O!			#DIVIO			#DIV.
Other	444444			`*.	#DIV/O	_			_			S		
	1111199911111		\$	*	#C!V.C'	S		#DIV:01	\$	700	#DIV 01	5		#DIV
Gross Other Revenue										212515		2	255 540	
Medicare	1///899////		\$	1.5	#DIV/GI	\$	341,711	79.2%	\$	348,545	79.2%	5	355,518	79.2
Medicaid	////N99/////			2.0	aD!A16.	S	367	0.0%	5		0.0%	S		0.0
Commercial	1///898////			130	#DIV/01	\$	83,331	19.3%	\$	84,997	19.3%	\$	86,697	19 3
Self Pay	1////886////			- 41	#DIVIO	\$	6,143	1.4%	\$	6,265	1.4%	5	6,391	1.49
Free Care / Bad Debt	1////8/6////				NDIVIOL.	\$	(4)	0.0%	\$	390	0.0%	\$		0.0
Other	1//////////////////////////////////////				#DIV/01	5		0.0%	S		0.0%	S		0.0
Strict	7////		S	¥	■DIV:01	\$	431,184	100 0%	Ş	439,808	100 0%	S	448,604	100 0
Gross Patient Revenue						_								
Medicare	1////868////		5	2)	#DIVIO	S	2.898.535	15.8%	\$	2,939,629	15.8%	S	2,980,860	15.8
Medicard	11/1/88/1///		•		#QiViCI		13,344,963	72.6%		13,523,778	72 6%		13,702,593	72.6
	//////////////////////////////////////						1,287,486	7.0%		1:305,288	7.0%		1,323,122	7.0
Commercial	4//////////////////////////////////////				#DIV/CI									
Self Pay	4//////////////////////////////////////			*	#O:A:0.		999,841	5.4%		1,013,279	5.4%		1,026,719	5.45
Free Care / Bad Debt	//// <i>?</i>			**	NDIA:0,		(140,774)	0.8%		(143,524)	·0 8%		(146,326)	-0.8
Other	4//////////////////////////////////////			- 1	#DIV-01	_		0.015	_		0.0%			0.03
	V///888////.		\$		#DIV/01	S	18,390,051	100 0%	\$	18,638,450	100 0%	\$	18,886,969	100 (
eductions from Revenue	e													
Medicare	1/////99/////		\$	2	WO!AIG.	\$	(1,111,257)	25.0%	\$	(1,150,359)	-26 0%	\$	(1,190,928)	-27 1
Medicaid	1///898////				#DIV/Q!	\$	5,307,858	119 2%	\$	5,325,931	120.2%	\$	5,340,789	121 4
Commercial	1////8/8/////			1.5	#OIVIO	5	256,830	\$ 8%	\$	254,018	5.7%	\$	250,828	5.75
Self Pay	1////868////			1.0	#DIV/01			0.0%	\$	140	0.0%	\$	74.1	0.08
Free Care / Bad Debt	1//////////////////////////////////////			140	#@1V/01	S		0.0%	S	/*	D 0 %	S		0.0%
Olher	1////66////				#DIV/OT	5		0.0%	\$		0.0%	\$	- 1	0.03
Skiel	7///88////		\$		#DIVIO#	\$	4,453,431	100 0%	\$	4,429,590	100 0%	\$	4,400,689	100 0
let Patient Revenue			_			_								
Medicare	1////6/6/////		\$	25	#DIMO:	s	4,009,792	28 8%	s	4,089,988	28.8%	\$	4,171,788	28.8
	1//////////////////////////////////////		Ψ.				8,037,105	57.7%	_	8,197,847	57.7%	•	8,361,804	57 7
Medicaid	1///2011//				#BIV/01									
Commercial	1//////////////////////////////////////				ROIA)G		1,030,656	7.4%		1,051,269	7.4%		1,072,294	7.4
Self Pay	1//////////////////////////////////////			F.3	#DIVION		999,841	7.2%		1,013,279	7 1%		1,026,719	7 1
Free Care / Bad Debt	1///898////				a DIA IG.		(140,774)	-1.0%		(143,524)	-10%		(146,326)	-10
Other DSP*			11111	1811/11/1	MDINIO	111	118681111	0.0%	111	11/8/8/1/1/	00%	111	1/166/1//	0.09
			21111			111	ATTITITITE.		-11	cerrettiii.		-11		

Latest actual numbers should lie to the hospital budget process

¹ Disproportionate share payments

Birchwood Operations, LLC and Birchwood Prop, LLC TABLE 6C REVENUE SOURCE PROJECTIONS WITH PROJECT

	La	test Actual	% of		Budget	% of	ı	Proposed Year 1	% of		Proposed Year 2 2019	% of	!	Proposed Year 3 2020	% of
	_	2016	Total		2017	Total		2018	Total	-	2013	Total	_	AVAV	. 010
iross Inpatient Revenue		5.5					_			_	0 504 094	14.2%	s	2,625,344	14.25
Medicare	\$	3,222,856	18 7%	\$	3,287,313	16 7%	\$	2,556,824	14.2%	\$	2,591,084		3		
Medicaid		13,662,464	70 9%		13,935,713	70 9%		13,344,963	74 3%		13,523,778	74.3%		13,702,593	74 31
Commercial		1,316,857	6.8%		1,342,990	68%		1,204,155	6.7%		1,220,290	6.7%		1,236,425	6 71
Self Pay		1,141,650	5 9%		1,164,483	5 9%		993,698	5 5%		1,007,013	5 5%		1.020.328	5 59
Free Care / Bad Debt		(64,825)	-0 3%		(66, 122)	-0 3%		(140,774)	-0.8%		(143,524)	-0 8%		(146,328)	-0 89
Other			0 0%			0 0%			0.0%			0 0%			0.09
	\$	19,278,802	100 0%	\$	19,664,378	100 0%	\$	17,958,867	100,0%	S	18,198,642	100 0%	\$	18,438,365	100 0
Gross Outpatient Revenue	-														
Medicare	\$		#DIV/01	\$	(3)	#DIV/0!	\$	-	■DIV/0!	\$	-	#DIV/01	\$	3.€3	₽ DŧV
Medicaid			#DIV/0!		3963	#DIV/01			#DIV/01			#DIV/01		•	#DIV
Commercial		*	#DIV/0!		1.0	#DIV/0!			SDIV/0			€DIV/0!			PDIV
Self Pay		2	#DIV/0!		192	#DIV/01		5 -	#DIV/O		*	#DIV/0I		3.€3	#DIV
Free Care / Bad Debt			#DIV/Q!		5#3	#DIV/01			#DIV/O		ŝ	#DIV/31		7-	øDIV.
Other -		-	#DIV/OI		1/4	#DIV/01			#DIV/O			DIV/01		3.02	OUV
Ollidi	\$		#DIV/0!	\$		■DIV/01	\$	3₹11	#DIV/0	\$	•	#DIV/01	\$	•	#DIV
Gross Other Revenue													_		
Medicare	5	328,442	79 2%	\$	335,011	79 2%	\$	341,711	79 2%	\$	348,545	79 2%	\$	355,516	79 2
Medicaid	-	-	0.0%		¥	0.0%		(*)	0.0%		*	0.0%			0.09
Commercial		80.095	19 3%		81.697	19.3%		83,331	19 3%		84,997	19 3%		86,697	19 3
=		5,904	1:4%		6.022	1.4%		6,143	1.4%		6,265	1.4%		6,391	1 45
Self Pay		•			0,022			141	0.0%		0,200	0.0%			0.09
Free Care / Bad Debt			0 0%			0.0%			0.0%		- 2	0.0%		-	0.01
Other	\$	414,441	100.0%	\$	422,730	100.0%	\$	431,184	100 0%	\$	439,808	100 0%	\$	448,804	100 (
Gross Patient Revenue	_			_		_				-	45.00				
Medicare	s	3,551,298	18 0%	s	3,622,324	18 0%	\$	2,898,535	15.8%	\$	2.939,629	15 8%	\$	2,980,860	158
	3	13,682,464	69.4%	•	13,935,713	69.4%	-	13,344,963	72 6%		13,523,778	72 6%		13,702,593	72 6
Medicaid						7.1%		1.287.486	7.0%		1,305,288	7.0%		1,323,122	70
Commercial		1,398,752	7.1%		1,424,687				5.4%		1,013,279	5 4%		1,026,719	54
Self Pay		1,147,554	5 8%		1,170,505	5 8%		999,841							-08
Free Care / Bad Debt		(64,825)	-0 3%		(66,122)	-0 3%		(140,774)	-0 8%		(143,524)	-0.6%		(146,326)	
Other	<u> </u>	19,693,243	100 0%	\$	20,087,108	100.0%	S	18,390,051	100.0%	\$	18,638,450	100 0%	\$	18,886,969	100
				_			_			_			_		
Deductions from Revenue		1540 005	• • • •	\$	(551,203)	-7.9%	s	(1,111,257)	-25 0%	s	(1,150,359)	-26 0%	S	(1,190,928)	-27 1
Medicare	\$	(540,395)	-7 9%	9	,,		3	5,307,858	119 2%	~	5,325,931	120 2%	•	5,340,789	121.
Medicaid		6,858,088	100 3%		6,995,250	100 3%					254.018	5 7%		250,828	57
Commercial		487,951	7 1%		497,710	7.1%		256,830	5 8%		•-			230,020	
Self Pay		30,662	0.4%		31,275	0.4%		•	0.0%		*	0.0%		₹	0.0
Free Care / Bad Debt		•	0 0%		-	0 0%		(<u>*</u>)	0 0%			0 0%		5	0.0
Other	_	5.48	0 0%			0.0%			0.0%	_	540	0.0%	_	×	0.0
	\$	6,836,306	100 0%	\$	6,973,032	100 0%	\$	4,453,431	100 0%	\$	4,429,590	100 0%	\$	4,400,689	100
Net Patient Revenue											4.000.000		_	4 474 700	
Medicare	\$	4,091,693	31 8%	\$	4,173,527	31 8%	\$	4,009,792	28 8%	\$	-	28 8%	\$		26 (
Medicaid		6,804,376	52.9%		6,940,464	52 9%		8,037,105	57.7%		8,197,847	57 7%		8,361,804	57.:
Commercial		908,801	7 1%		926,977	7 1%		1,030,656	7.4%		1,051,269	7 4%		1.072,294	7.4
Self Pay		1,116,892	8 7%		1,139,230	8 7%		999,841	7 2%		1,013,279	7_1%		1,026,719	7
Free Care / Bad Debt		(64,825)	-0.5%		(86,122)	-0 5%		(140,774)	-1.0%		(143,524)	-1.0%		(148,326)	-1
Other		(0 ()==0)	0.0%			0.0%		*	0.0%			0.0%		-	0 (
DSP*			0.0%		7.	0.0%			0.0%		7 .	0.0%			0.0

Latest actual numbers should tie to the hospital budget process.

^{*} Disproportionate share payments

Birchwood Operations, LLC and Birchwood Prop, LLC

TABLE 7
UTILIZATION PROJECTIONS
TOTALS

A: WITHOUT PROJECT			Proposed	Proposed	Proposed
8407	Latest Actual 2016	Budget 2017	Year 1 2018	Year 2 2019	Year 3 2020
Inpatient Utilization					
Staffed Beds	144	144			
Admissions	407	407			
Patient Days	48,452	48,452			
Average Length of Stay	117.18	117,18			
Outpatient Utilization					
All Outpatient Visits					
OR Procedures					
Observation Units					
Physician Office Visits					
Ancillary					
All OR Procedures					
Emergency Room Visits					
Adjusted Statistics					
Adjusted Admissions					
Adjusted Patient Days					

B: PROJECT ONLY			Proposed	Proposed	Proposed
	Latest Actual 2016	Budget 2017	Year 1 2018	Year 2 2019	Year 3 2020
Inpatient Utilization				4.14	444
Staffed Beds	/////NSON/////		144	144	144
Admissions	/////NW//////		407	407	407
Patient Days	/////8868/////		48,505	48,505	48,505
Average Length of Stay	/////808/////		117.18	117.18	117.18
Outpatient Utilization	/////888/////				
All Outpatient Visits	/////898/////				
OR Procedures	///////////////////////////////////////				
Observation Units	/////856/////				
Physician Office Visits	/////868/////				
Ancillary	///////////////////////////////////////				
All OR Procedures	/////8888/////				
Emergency Room Visits	///////////////////////////////////////	-		200	
Adjusted Statistics	/////888/////		*	S#7	
Adjusted Admissions	/////s(s/////				
Adjusted Patient Days	///////////////////////////////////////		* 1		

C: WITH PROJECT			Proposed	Proposed	Proposed
	Latest Actual 2016	Budget 2017	Year 1 2018	Year 2 2019	Year 3 2020
Inpatient Utilization				4.4	444
Staffed Beds	144	144	144	144	144
Admissions	407	407	407	407	407
Patient Days	48,452	48,452	48,505	48,505	48,505
Average Length of Stay	117.18	117-18	117.18	117.18	117.18
Outpatient Utilization					
All Outpatient Visits	*	3.5	3	-	3.
OR Procedures	¥	190	₩	*	N7 1
Observation Units	×	1.51	9	2	*
Physician Office Visits	2	(34)	3	-	
Ancillary					
All OR Procedures	8	S#3		*	55.0
Emergency Room Visits	(*)	S.# ?			
Adjusted Statistics					
Adjusted Admissions	€				
Adjusted Patient Days					

NOTE: When completing this table make entries in the shaded fields only.

Birchwood Operations, LLC and Birchwood Prop, LLC

TABLE 9 STAFFING PROJECTIONS TOTALS

A: WITHOUT PROJECT			Proposed	Proposed	Proposed
	Latest Actual	Budget	Year 1	Year 2	Year 3
	2016	2017	2018	2019	2020
Non-MD FTEs					
Total General Services	14.9	14.9			
Total Inpatient Routine Services	85.1	85.1			
Total Outpatient Routine Services	0.0	0.0			
Total Ancillary Services	10.4	10.4			
Total Other Services	34.8	34.8			
Total Non-MD FTEs	145.2	145.2	0.0	0.0	0.0
Physician FTEs	0.3	0.3			
Direct Service Nurse FTEs	37.1	37.1			

B: PROJECT ONLY			Proposed	Proposed	Proposed
	Latest Actual 2016	Budget 2017	Year 1 2018	Year 2 2019	Year 3 2020
Non-MD FTEs	Daniel and San				440
Total General Services	<i>/////8908/////</i>	0.0	14.9	14.9	14.9
Total Inpatient Routine Services	/////NUX/////	0.0	85.1	85.1	85.1
Total Outpatient Routine Services	//////NXX/////	0.0	0.0	0.0	0.0
Total Ancillary Services	/////ks/k/////	0.0	10.4	10.4	10.4
Total Other Services	/////kgk/////	0.0	34.8	34.8	34.8
Total Non-MD FTEs	7//////////////////////////////////////	0.0	145.2	145.2	145.2
Physician Services	/////xix/////	0.0	0.3	0.3	0.3
Direct Service Nurse FTEs	/////898/////	0.0	37.1	37.1	37.1

C: WITH PROJECT			Proposed	Proposed	Proposed
	Latest Actual 2016	Budget 2017	Year 1 2018	Year 2 2019	Year 3 2020
Non-MD FTEs			44.0	14.0	14.9
Total General Services	#VALUE!	14.9	14.9	14.9	
Total Inpatient Routine Services	#VALUE!	85.1	85.1	85.1	85.1
Total Outpatient Routine Services	#VALUE!	0.0	0.0	0.0	0.0
Total Ancillary Services	#VALUE!	10.4	10.4	10.4	10.4
Total Other Services	#VALUE!	34.8	34.8	34.8	34.8
Total Non-MD FTEs	#VALUE!	145.2	145.2	145.2	145.2
Physician Services	#VALUE!	0.3 37 .1	0.3 37.1	0.3 37.1	0.3 37.1
Direct Service Nurse FTEs	#VALUEI	31.1	37.1	0711	••••

ATTACHMENT 15

Kindred Nursing Centers East, LLC d/b/a Birchwood Terrace Healthcare

Report on Audit of Financial Statements for the year ended December 31, 2016

KINDRED NURSING CENTERS EAST, LLC d/b/a BIRCHWOOD TERRACE HEALTHCARE INDEX

CONTENTS

	Page
Report of Independent Auditors	1
Financial Statements:	
Statement of Operations and Accumulated Deficit	2
Balance Sheet	3
Statement of Cash Flows	4
Notes to Financial Statements	5



Report of Independent Auditors

To the Management of Kindred Nursing Centers East, LLC d/b/a Birchwood Terrace Healthcare:

We have audited the accompanying financial statements of Kindred Nursing Centers East LLC d/b/a Birchwood Terrace Healthcare, a wholly-owned operating component of Kindred Healthcare Inc., which comprise the balance sheet as of December 31, 2016, and the related statement of operations and accumulated deficit and cash flows for the year then ended.

Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

Auditors' Responsibility

Our responsibility is to express an opinion on the financial statements based on our audits. We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free from material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on our judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, we consider internal control relevant to the Company's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Opinion

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Kindred Nursing Centers East LLC d/b/a Birchwood Terrace Healthcare as of December 31, 2016, and the results of its operations and its cash flows for the year then ended in accordance with accounting principles generally accepted in the United States of America.

Pricewaterhouse Coopers LLP

May 31, 2017

KINDRED NURSING CENTERS EAST, LLC d/b/a BIRCHWOOD TERRACE HEALTHCARE STATEMENT OF OPERATIONS AND ACCUMULATED DEFICIT

For the year ended December 31, 2016

Revenues	\$	12,922,343
Salaries, wages and benefits		7,136,562
Supplies		662,822
Rent		1,485,544
Other operating expenses		4,887,769
Depreciation		152,558
Investment income		(368)
Total operating expenses		14,324,887
Net loss		(1,402,544)
Accumulated deficit at beginning of year		(3,787,137)
Accumulated deficit at end of year	_\$_	(5,189,681)

KINDRED NURSING CENTERS EAST, LLC d/b/a BIRCHWOOD TERRACE HEALTHCARE BALANCE SHEET

As of December 31, 2016

ASSETS

Current assets:		
Cash and cash equivalents	\$	57,833
Accounts receivable less allowance of \$39,595		1,646,392
Inventories		30,765
Insurance recoverables		481,596
Other current assets		4,228
		2,220,814
Property and equipment:		
Land and land improvements		12,260
Leasehold improvements		2,716,674
Equipment		1,340,381
		4,069,315
Accumulated depreciation		(2,900,820)
		1,168,495
Insurance recoverables		939,646
Patient fund accounts		37,193
	\$	4,366,148
	,	
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Accounts payable	\$	241,376
Salaries, wages and other compensation		222,496
Patient credit balances		76,970
Professional liability and workers compensation		481,596
Other accrued liabilities		3,187
		1,025,625
Patient fund accounts		37,193
Deferred rent		324,083
Professional liability and workers compensation		939,646
Commitments and contingencies (Note 4)		
Stockholders' equity:		(5.100.601)
Accumulated deficit		(5,189,681)
Net contributions from Kindred Healthcare, Inc.		7,229,282
		2,039,601
	\$	4,366,148

See accompanying notes.

KINDRED NURSING CENTERS EAST, LLC d/b/a BIRCHWOOD TERRACE HEALTHCARE STATEMENT OF CASH FLOWS

For the year ended December 31, 2016

Cash flows from operating activities:	
Net loss	\$ (1,402,544)
Adjustments to reconcile net loss to net cash used in operating activities:	
Depreciation	152,558
Provision for doubtful accounts	64,825
Change in operating assets and liabilities:	
Accounts receivable	(295,671)
Inventories and other assets	(391,470)
Accounts payable	41,592
Salaries, wages and other compensation	2,820
Patient credit balances and other accrued liabilities	 387,505
Net cash used in operating activities	(1,440,385)
Cash flows from investing activities: Purchase of property and equipment	(592,163)
Cash flows from financing activities:	
Net increase of contributions due from Kindred Healthcare, Inc.	 2,084,408
Change in cash and cash equivalents	51,860
Cash and cash equivalents at beginning of year	 5,973
Cash and cash equivalents at end of year	\$ 57,833
Supplemental information:	
Transfers of property and equipment from Kindred	\$ 5,380
Property and equipment purchases payable	10,801

NOTE 1 – ACCOUNTING POLICIES

Reporting Entity

Kindred Nursing Centers East, LLC d/b/a Birchwood Terrace Healthcare (the "Facility") is a wholly-owned operating component of Kindred Healthcare, Inc. ("Kindred" or the "Company") and has no separate legal status or existence. The Facility owns and operates a 144-bed skilled nursing facility located in Burlington, Vermont.

Basis of Presentation

As a wholly-owned operating component of Kindred with no separate legal status or existence, the Facility is subject to the accounting policies of Kindred. The accompanying financial statements have been prepared in accordance with generally accepted accounting principles ("GAAP") and include amounts based upon the estimates and judgments of management. Actual amounts may differ from those estimates.

Recently Issued Accounting Requirements

In January 2017, the Financial Accounting Standards Board (the "FASB") issued authoritative guidance that revises the definition of a business, which affects accounting for acquisitions, disposals, goodwill impairment, and consolidation. The guidance is intended to help entities evaluate whether transactions should be accounted for as acquisitions (or disposals) of assets or businesses. The revision provides a more robust framework to use in determining when a set of assets and activities is a business. The new guidance is effective for annual and interim periods beginning after December 15, 2017 and early adoption is permitted. The adoption of this standard is not expected to have a material impact on the Facility's business, financial position, results of operations or liquidity.

In November 2016, the FASB issued authoritative guidance that simplifies the disclosure of restricted cash within the statement of cash flows. The guidance is intended to reduce diversity when reporting restricted cash and requires entities to explain changes in the combined total of restricted and unrestricted balances in the statement of cash flows. The new guidance is effective for annual and interim periods beginning after December 15, 2017 and early adoption is permitted. The adoption of this standard is not expected to have a material impact on the Facility's statement of cash flows.

In August 2016, the FASB issued authoritative guidance to eliminate diversity in practice related to the cash flow statement classification of eight specific cash flow issues, which include debt prepayment or extinguishment costs, maturity of a zero coupon bond, settlement of contingent consideration liabilities after a business combination, proceeds from insurance settlements and distribution from certain equity method investees. The new guidance is effective for annual and interim periods beginning after December 15, 2017 and early adoption is permitted. The adoption of this standard is not expected to have a material impact on the Facility's statement of cash flows.

NOTE 1 – ACCOUNTING POLICIES (continued)

Recently Issued Accounting Requirements (continued)

In February 2016, the FASB issued amended authoritative guidance on accounting for leases. The new provisions require that a lessee of operating leases recognize in the statement of financial position a liability to make lease payments (the lease liability) and a right-of-use asset representing its right to use the underlying asset for the lease term. The lease liability will be equal to the present value of lease payments, with the right-of-use asset based upon the lease liability. The classification criteria for distinguishing between finance (or capital) leases and operating leases are substantially similar to the previous lease guidance, but with no explicit bright lines. As such, operating leases will result in straight-line rent expense similar to current practice. For short term leases (term of 12 months or less), a lessee is permitted to make an accounting election not to recognize lease assets and lease liabilities, which would generally result in lease expense being recognized on a straightline basis over the lease term. The guidance is effective for annual and interim periods beginning after December 15, 2018, and will require application of the new guidance at the beginning of the earliest comparable period presented. We will not elect early adoption and will apply the modified retrospective approach as required. The adoption of this standard is expected to have a material impact on the Facility's financial position. The Facility is still evaluating the impact on its results of operations and there is no impact on liquidity.

In August 2014, the FASB issued authoritative guidance requiring management to evaluate whether there are conditions and events that raise substantial doubt about the entity's ability to continue as a going concern and to provide disclosures in certain circumstances. The guidance is effective for annual and interim periods ending after December 15, 2016. This guidance did not have a material impact on the Facility's financial statements.

In May 2014, the FASB issued authoritative guidance which changes the requirements for recognizing revenue when entities enter into contracts with customers. Under the new provisions, an entity will recognize revenue when it transfers promised goods or services to customers in an amount that reflects what it expects in exchange for the goods or services. It also requires more detailed disclosures to enable users of financial statements to understand the nature, amount, timing, and uncertainty of revenue and cash flows arising from contracts with customers.

- In July 2015, the FASB finalized a one year deferral of the new revenue standard with an updated effective date for interim and annual periods beginning on or after December 15, 2017. Entities are not permitted to adopt the standard earlier than the original effective date, which was on or after December 15, 2016.
- In March 2016, the FASB finalized its amendments to the guidance in the new revenue standard on assessing whether an entity is a principal or an agent in a revenue transaction. Under the new amendments, the FASB confirmed that a principal in an arrangement controls a good or service before it is transferred to a customer but revised the structure of indicators when an entity is the principal. The amendments have the same effective date and transition requirements as the new revenue standard.

NOTE 1 - ACCOUNTING POLICIES (continued)

Recently Issued Accounting Requirements (continued)

• In May 2016, the FASB finalized its amendments to the guidance in the new revenue standard on contracts with customers and specifically, collectability, non-cash consideration, presentation of sales taxes, and completed contracts. The amendments are intended to reduce the risk of diversity in practice and the cost and complexity of applying certain aspects of the revenue standard. The amendments have the same effective date and transition requirements as the new revenue standard, which is effective for interim and annual periods beginning on or after December 15, 2017, with early adoption permitted on or after December 15, 2016.

The Facility will not elect early adoption but will apply the modified retrospective approach upon the required effective date. The Facility is still evaluating the impact of the adoption of the new revenue standard on its business, financial position, results of operations, and liquidity.

Net Patient Service Revenue

Net patient service revenues are recorded based upon estimated amounts due from patients and third-party payors for healthcare services provided, including anticipated settlements under reimbursement agreements with Medicare, Medicaid, Medicare Advantage and other third-party payors.

A summary of revenues by payor type for the year ended December 31, 2016 follows:

Medicare	\$ 3,849,656
Medicaid	6,804,376
Medicare Advantage	229,141
Private and other	2,039,170
	\$ 12,922,343

Revenues under third-party agreements are subject to examination and retroactive adjustment. Provisions for estimated third-party adjustments are provided in the period the related services are rendered. Differences between the amounts accrued and subsequent settlements are recorded in the periods the interim or final settlements are determined.

Cash and Cash Equivalents

Cash and cash equivalents include unrestricted highly-liquid investments with an original maturity of three months or less when purchased. Cash restricted relates to patient trust accounts. The carrying value of cash and cash equivalents approximates fair market value.

Accounts Receivable

Accounts receivable consist primarily of amounts due from the Medicare and Medicaid programs, other government programs, managed care health plans, commercial insurance companies and individual patients. Estimated provisions for doubtful accounts are recorded to the extent it is probable that a portion or all of a particular account will not be collected.

In evaluating the collectibility of accounts receivable, the Facility considers a number of factors, including the age of the accounts, changes in collection patterns, the composition of patient accounts by payor type, the status of ongoing disputes with third-party payors and general industry conditions.

NOTE 1 – ACCOUNTING POLICIES (continued)

Due to third-party payors

The Facility is required to submit cost reports at least annually to various state and federal agencies administering the respective reimbursement programs. In many instances, interim cash payments to the Facility are only an estimate of the amount due for services provided. Any overpayment to the Facility arising from the completion of a cost report is recorded as a liability.

Inventories

Inventories consist primarily of medical supplies and have been reflected in the accompanying balance sheet at the lower of cost (first-in, first-out) or market. Inventory carrying value was \$30,765 at December 31, 2016.

Property and Equipment

Property and equipment is carried at cost less accumulated depreciation. Depreciation expense for the Facility, computed by the straight-line method, was \$152,558 for the year ended December 31, 2016. Leasehold improvements are depreciated over their estimated useful lives or the remaining lease term, whichever is shorter. Estimated useful lives of equipment vary from 5 to 10 years. Repairs and maintenance are expensed as incurred.

Net Contributions from Kindred

Net contributions from Kindred Healthcare, Inc. are classified as an increase to equity on the accompanying balance sheet. For the year ended December 31, 2016, transfers of property and equipment from Kindred to the Facility increased amounts due from Kindred by \$5,380, in addition to other various interdivisional transactions such as cash management, accounts receivable processing, property and equipment record keeping, accounts payable processing, payroll and general bookkeeping, as further discussed in note 6, which increased amounts due from Kindred by \$2,084,408.

Comprehensive Income

The Facility has no components of other comprehensive income or loss and as a result, comprehensive income or loss is equal to net income or loss as presented in the accompanying statement of operations and accumulated deficit.

Other Information

On November 7, 2016, Kindred Healthcare Operating, Inc., the owner of the Facility, announced plans to exit its skilled nursing facility business as an owner and operator. Accordingly, while Kindred is unable at this time to determine an expected completion date, Kindred is targeting to complete the exit from the skilled nursing facility business by the end of 2017. At this time, there has been no determination of the implications to the ownership of the Facility.

The company has performed an evaluation of subsequent events through May 31, 2017, the date on which the financial statements were issued.

NOTE 2 – INCOME TAXES

The Facility is included in the consolidated federal and state income tax returns filed by the Company. The Company allocates the consolidated federal and state income tax liabilities among the members of the consolidated return group (including the Facility) using a separate return method. Amounts determined to be a payable or receivable under the separate return method are classified as an element of net contributions to or from Kindred on the accompanying balance sheet at December 31, 2016.

The provision or benefit for income taxes is based upon the Facility's annual reported income or loss for each respective accounting period. The Facility recognizes an asset or liability for the deferred tax consequences of temporary differences between the tax bases of assets and liabilities and their reported amounts in the financial statements. These temporary differences will result in taxable or deductible amounts in future years when the reported amounts of the assets are recovered or liabilities are settled and represent amounts due to or due from the Company in lieu of taxes. A valuation allowance is provided for these deferred tax assets if it is more likely than not that some portion or all of the net deferred tax assets will not be realized. Net operating losses are recognized on a benefits-for-loss basis which modifies the separate return method so that net operating losses that would not be realized (or realizable) by the Facility are recognized as a benefit for the Facility, given that they are realized (or realizable) by the Company.

The Facility had no income tax benefit or expense for the year ended December 31, 2016.

Reconciliation of federal statutory tax benefit to the benefit for income taxes for the year ended December 31, 2016 follows:

Income tax benefit at federal rate	\$(490,890)
Valuation allowance	488,468
Other items, net	2,422
	\$ -

A summary of deferred income taxes by source included in the balance sheet at December 31, 2016 follows:

	Assets	_Liabilities_
Property and Equipment	\$ 124,286	\$ -
Accounts receivable allowances	-	53,839
Compensation	17,085	:=:
Professional liability and worker's compensation risks	39,923	
Other	113,429	
Net operating losses	566,461	*
	861,184	\$ 53,839
Reclassification of deferred tax liabilities	(53,839)	
	807,345	
Valuation allowance	(807,345)	
Net deferred tax assets	\$ -	

NOTE 3 – LEASES

The Facility leases real estate and equipment under non-cancelable arrangements. The real estate under a non-cancelable operating lease is part of a master lease agreement along with other Kindred facilities. The real estate lease agreement to which the Facility is a party expires on April 30, 2025. Additionally, the master lease agreement provides for rent escalation annually on May 1. All annual rent escalators are payable in cash. The contingent annual rent escalator for the master lease agreement is based upon annual increases in the Consumer Price Index, subject to a ceiling of 4%. In 2016, the Facility recorded \$9,867 of contingent rent for the year ended December 31, 2016. Contingent rent is included in rent expense in the year incurred.

Rent expense related to non-cancelable operating leases amounted to \$1,485,544 for the year ended December 31, 2016. Future minimum payments under the real estate non-cancelable operating lease are as follows:

2017	\$ 1,468,987
2018	1,468,987
2019	1,468,987
2020	1,468,987
2021	1,468,987
Thereafter	4,896,623

On November 11, 2016, as part of the Company's strategic decision to exit the skilled nursing facility business, the Company entered into an agreement with Ventas which provides it with the option to acquire the real estate for all 36 skilled nursing facilities (the "Ventas SNFs") currently leased under the Master Lease Agreements for an aggregate consideration of \$700 million. The agreement also provides that, through October 31, 2018, the Company has the right to find one or more purchasers of the Ventas SNFs. As the Company locates new owners/operators for the Ventas SNFs, in exchange for the Company's payment to Ventas of the allocable portion of the \$700 million purchase price, Ventas has agreed to convey the real estate for the applicable Ventas SNF to the new owner/operator. The Company, at its option, may also elect to renew the leases for any of the Ventas SNFs through April 30, 2025, and transfer them into Master Lease Agreement No. 5. The Ventas SNFs will remain leased under their current Master Lease Agreements until the Company exercises its purchase option or April 30, 2018, whichever comes first. If the Company does not complete the acquisition of the Ventas SNFs by April 30, 2018, the lease for any remaining Ventas SNFs will be automatically renewed through April 30, 2025, and transferred into Master Lease Agreement No. 5. Since all of the Ventas SNFs will either be sold or transferred into Master Lease Agreement No. 5, Kindred's other Master Lease Agreements with Ventas will be effectively terminated and only Master Lease Agreement No. 5 will remain.

NOTE 4 – CONTINGENCIES

Laws and regulations governing the Medicare and Medicaid programs are complex and subject to interpretation. The Facility believes it is in compliance with all applicable laws and regulations and is not aware of any pending or threatened investigations involving allegations or potential wrongdoing. While no such regulatory inquiries have been made, compliance with such laws and regulations can be subject to future government review and interpretation as well as significant regulatory action including fines, penalties, and exclusion from the Medicare and Medicaid programs. Management is unaware of any commitments or contingencies that would result in losses to the Facility.

The Facility insures its professional liability and workers compensation risks through a wholly-owned, limited purpose insurance subsidiary of the Company (the "Insurance Subsidiary"). The following is a summary of the insurance recoverables and reserves under the policies as of December 31, 2016:

	9	Professional liability		Workers compensation		Total	
Assets:							
	Current	\$	94,682	\$	386,914	\$	481,596
	Long term		145,140		794,506		939,646
	_	\$	239,822	\$	1,181,420	\$	1,421,242
Liabilities:							
	Current	\$	94,682	\$	386,914	\$	481,596
	Long term		145,140		794,506		939,646
		\$	239,822	\$	1,181,420	\$	1,421,242

NOTE 5 - CONCENTRATION OF CREDIT RISK

The Facility derives a majority of its revenue through provider agreements with the Centers for Medicare and Medicaid Services and the Vermont Department of Health and Human Services. Accordingly, receivables from these third parties constitute the majority of the Center's patient accounts receivable which consisted of the following at December 31, 2016:

Private and other	41 %
Medicaid	33
Medicare	26
	100 %

Management monitors and evaluates the allowance for doubtful accounts to ensure that receivables are stated at their net realizable value.

NOTE 6 – RELATED-PARTY TRANSACTIONS

As a wholly-owned operating component of the Company, the Facility is subject to the accounting policies of the Company and is a party to numerous transactions with the Company. The Company is jointly liable for the obligations of the Facility and has pledged substantially all of the assets of the Facility to collateralize the Company's ABL Facility, Term Loan Facility and Notes (all as defined below).

NOTE 6 – RELATED PARTY TRANSACTIONS (Continued)

All obligations under the ABL Facility and Term Loan Facility are fully and unconditionally guaranteed, subject to certain customary release provisions, by substantially all of the Company's existing and future direct and indirect domestic 100% owned subsidiaries, as well as certain non-100% owned domestic subsidiaries as the Company may determine from time to time in its sole discretion.

The Notes are fully and unconditionally guaranteed, subject to certain customary release provisions, by substantially all of the Company's domestic 100% owned subsidiaries.

ABL Facility and Incremental ABL Joinder

On April 9, 2014, the Company entered into a second amendment and restatement agreement (the "Second ABL Amendment Agreement") among the Company, the other credit parties party thereto, JPMorgan Chase Bank, N.A. as administrative agent and collateral agent, and the lenders party thereto. The Second ABL Amendment Agreement, among other items, (1) extends the maturity date of the prior ABL Facility from June 1, 2018 to April 9, 2019, (2) provides for the replacement of all revolving commitments outstanding under the prior ABL Facility with new revolving commitments in the same principal amount, (3) increases the amounts available for incremental commitments and (4) amends certain provisions related to the incurrence of debt and liens and the making of acquisitions, investments and restricted payments.

On October 31, 2014, the Company entered into a third amendment and restatement agreement (the "ABL Amendment Agreement") among the Company, the consenting lenders party thereto and JPMorgan Chase Bank, N.A., as administrative agent. The ABL Amendment Agreement amended and restated the ABL Credit Agreement dated as of June 1, 2011, as amended by that certain Amendment No. 1 to the ABL Credit Agreement dated as of October 4, 2012 and as further amended and restated by that certain Amendment and Restatement Agreement dated as of August 21, 2013 and that certain Second Amendment and Restatement Agreement dated as of April 9, 2014 (the "ABL Facility"), among the Company, the lenders from time to time party thereto and JPMorgan Chase Bank, N.A., as administrative agent and collateral agent.

The ABL Amendment Agreement, among other items, modified certain provisions related to the issuance of Notes into the Escrow Accounts. Upon the consummation of Kindred's acquisition of Gentiva Health Services, Inc. on February 2, 2015, and the satisfaction of certain other conditions, the ABL Amendment Agreement further amended and restated the ABL Facility to, among other items, modify certain provisions related to the incurrence of debt and the making of acquisitions, investments and restricted payments.

The ABL Amendment Agreement did not modify the maturity date of the revolving commitments thereunder or the applicable interest rate margins applicable to any borrowings thereunder.

In addition, on December 12, 2014, the Company entered into an incremental joinder agreement (the "Incremental ABL Joinder") among the Company, JPMorgan Chase Bank, N.A., as administrative agent and collateral agent, the incremental lenders party thereto and the other credit parties party thereto. Upon the consummation of the Gentiva Merger and the satisfaction of certain other conditions, the Incremental ABL Joinder provided for additional revolving commitments in an aggregate principal amount of \$150 million under the ABL Facility.

NOTE 6 – RELATED PARTY TRANSACTIONS (Continued)

ABL Amendment No. 2 to the ABL Facility

On June 3, 2015, the Company entered into an amendment agreement to the ABL Amendment Agreement (the "ABL Amendment No. 2"), which modified the restrictions on the amount of cash and temporary cash investments that may be held outside of certain deposit accounts subject to control agreements. As used herein, the "ABL Facility" refers to the ABL Amendment Agreement, as amended by the Incremental ABL Joinder and the ABL Amendment No. 2.

Aside from the foregoing changes, the terms and conditions of the ABL Amendment Agreement were substantially similar to the terms and conditions before the effectiveness of the ABL Amendment No. 2.

Term Loan Facility

On April 9, 2014, the Company entered into a third amendment and restatement agreement (the "Third Amended and Restated Term Loan Facility") among the Company, the other credit parties party thereto, JPMorgan Chase Bank, N.A., as administrative agent and collateral agent, and the lenders party thereto. The Third Amended and Restated Term Loan Facility, among other items, (1) extends the maturity date of the prior Term Loan Facility from June 1, 2018 to April 9, 2021, (2) provides for the replacement of all term loans outstanding under the prior Term Loan Facility with new term loans in a principal amount of \$1 billion, (3) reduces the interest rate margins applicable to the term loans, (4) increases the available capacity for incremental term loans and (5) amends certain provisions related to the incurrence of debt and liens and the making of acquisitions, investments and restricted payments.

On November 25, 2014, the Company entered into a fourth amendment and restatement agreement (the "Fourth Amended and Restated Term Loan Facility") among the Company, the consenting lenders party thereto and JPMorgan Chase Bank, N.A., as administrative agent. The Fourth Amended and Restated Term Loan Facility amended and restated the Term Loan Credit Agreement dated as of June 1, 2011, as amended by that certain Incremental Amendment No. 1 to the Term Loan Credit Agreement dated as of October 4, 2012 and as further amended and restated by that certain Amendment and Restatement Agreement dated as of May 30, 2013, that certain Second Amendment and Restatement Agreement dated as of August 21, 2013 and that certain Third Amendment and Restatement Agreement dated as of April 9, 2014, among the Company, the lenders party thereto and JPMorgan Chase Bank, N.A., as administrative agent and collateral agent (the "Term Loan Facility").

The Fourth Amended and Restated Term Loan Facility amended and restated the Term Loan Facility to, among other items, (i) increase the applicable interest rate margins for London Interbank Offered Rate borrowings from 3.00% to 3.25% and for base rate borrowings from 2.00% to 2.25%, (ii) temporarily increase the maximum total leverage ratio permitted under the financial maintenance covenants, (iii) include soft-call protection at a prepayment premium of 1.00% for twelve months starting from November 25, 2014 and (iv) modify certain provisions related to the incurrence of debt and the making of acquisitions, investments and restricted payments. The Fourth Amended and Restated Term Loan Facility did not modify the maturity date of the loans made thereunder.

NOTE 6 – RELATED PARTY TRANSACTIONS (Continued)

Incremental Term Loan Amendment to Term Loan Facility

On March 10, 2015, the Company entered into an incremental amendment agreement to the Fourth Amended and Restated Term Loan Facility (the "Incremental Term Loan Agreement"), which provided for an incremental term loan in an aggregate principal amount of \$200 million under its Fourth Amended and Restated Term Loan Facility. The Company used the net proceeds of the incremental term loan to repay outstanding borrowings under the Third Amended and Restated ABL Facility. The incremental term loan was issued with 50 basis points of original issue discount and has the same terms as, and is fungible with, all other term loans outstanding under the Fourth Amended and Restated Term Loan Facility. As used herein, the "Term Loan Facility" refers to the Fourth Amended and Restated Term Loan Facility, as amended by the Incremental Term Loan Agreement.

On June 14, 2016, the Company entered into the Term Loan Credit Agreement that amended and restated the Term Loan Facility to provide for, among other things, (1) additional joint venture flexibility, including an increased ability to enter into and make investments in joint ventures that are non-guarantor restricted subsidiaries and to incur debt and liens of such joint ventures and other non-guarantor restricted subsidiaries, (2) an increase in the size of a basket for asset sales from 15% to 25% of consolidated total assets, (3) maintaining a maximum total leverage ratio of 6.00:1.00 for each quarterly measurement date after the date of such amendment, and (4) an incremental term loan in an aggregate principal amount of \$200 million. The incremental term loan was issued with 95 basis points of original issue discount ("OID") and has the same terms as, and is fungible with, the \$1.18 billion in aggregate principal amount of term loans that were outstanding under the Term Loan Facility immediately prior to the effectiveness of the Term Loan Credit Agreement. The net proceeds from the incremental term loan were used to repay a portion of the outstanding borrowings under the Company's ABL Facility.

On March 14, 2017, the Company entered into the Term Loan Amendment Agreement that amended and restated the Term Loan Facility to, among other things, (1) make adjustments to certain covenants and definitions to better accommodate the Company's previously announced plan to sell its skilled nursing division, (2) provide the Company with increased leverage covenant flexibility for an interim period, (3) increase the applicable margin on the outstanding borrowings from 3.25% to 3.50% for LIBOR borrowings and from 2.25% to 2.50% for base rate borrowings, (4) require a maximum leverage ratio of no more than 5.00 to 1.00 for use of the \$50 million annual dividend basket, and (5) provide for a prepayment premium of 1.00% in connection with any repricing transaction within six months of the closing date. In accordance with the authoritative guidance on debt, the Company accounted for the amendment as a debt modification.

Aside from the foregoing changes, the terms and conditions of the Fourth Amended and Restated Term Loan Facility were substantially similar to the terms and conditions before the effectiveness of the Incremental Term Loan Agreement.

NOTE 6 – RELATED PARTY TRANSACTIONS (Continued)

Notes

On April 9, 2014, the Company completed a private placement of \$500 million aggregate principal amount of 6.375% senior notes due 2022. On December 18, 2014, Kindred Escrow Corp. II, one of the Company's subsidiaries, completed a private placement of \$750 million aggregate principal amount of 8.00% senior notes due 2020 and \$600 million aggregate principal amount of 8.75% senior notes due 2023. The senior notes due 2022, 2020 and 2023 are collectively referred to as the "Notes."

Other Related Party Transactions

A significant portion of transactions are processed by the Company on the Facility's behalf including cash management, accounts receivable processing, property and equipment record keeping, accounts payable processing, payroll and general bookkeeping. Additionally, the Company manages general business functions on behalf of the Facility including cost reimbursement reporting, human resources, financial reporting, income taxes and legal services. Expenses incurred by the Company related to the operations of the Facility are allocated to the Facility based on a percentage of net revenues. The Company has allocated expenses of \$752,517 for the year ended December 31, 2016 to the Facility which are included in other operating expenses on the accompanying statement of operations and accumulated deficit.

The Company provides certain rehabilitation services to the Facility. The amount for rehabilitation services included in other operating expenses on the accompanying statement of operations and accumulated deficit for the year ended December 31, 2016 was \$1,042,058. This amount was based upon contracted rates for rehabilitation services.

The Facility insures the primary layer of its professional and general liability risks and its workers compensation risks through the Insurance Subsidiary. Risks in excess of those retained by the Insurance Subsidiary are reinsured with unaffiliated commercial insurance carriers. Liability premiums paid to the Insurance Subsidiary totaled \$35,042 in 2016 and are included in other operating expenses on the accompanying statement of operations and accumulated deficit. Workers compensation premiums paid to the Insurance Subsidiary totaled \$176,345 in 2016 and are included in salaries, wages and benefits on the accompanying statement of operations and accumulated deficit. See Note 4 for insurance recoverables and liabilities.

The Facility participates in an employee medical benefits plan sponsored by the Company. The plan provides medical benefits to participating employees and their qualifying dependents who meet certain eligibility requirements. Medical plan expense totaled \$321,013 in 2016 and is included in salaries, wages and benefits on the accompanying statement of operations and accumulated deficit.

FILED UNDER SEAL

Milton Ostreicher Personal Financial Statement

Attachment 25 - Overall Star Rating	Aug-17	Sep-17	Oct-17	Nov-17	Dec-17	Jan-17
Birchwood Terrace	4	4	3	3	3	3 Avg = 3.33
Achieve Rehabilitation and Nursing Center	3	3	3	3	3	3 Avg = 3
Beacon Nursing and Rehabiliation	5	5	5	5	5	5 Avg = 5
Highland Care Center	3	3	3	3	3	4 Avg = 3.16
Kindred Transitional Care And Rehab- Smith Ranch	3	3	3	3	3	3 Avg = 3

Attachment 26 - Health and Fire	Aug-17	Sep-17	Oct-17	Nov-17	Dec-17	Jan-17
Birchwood Terrace	3	3	3	3	3	3 Avg = 3
Achieve Rehabilitation and Nursing Center	4	4	4	3	4	4 Avg = 3.83
Beacon Nursing and Rehabiliation	5	5	5	5	5	5 Avg = 5
Highland Care Center	4	4	4	4	4	4 Avg = 4
Kindred Transitional Care And Rehab- Smith Ranch	2	2	2	2	2	2 Avg = 2

Attachment 27 - Staffing	Aug-17	Sep-17	Oct-17	Nov-17	Dec-17	Jan-17
Birchwood Terrace	4	4	3	3	3	3 Avg = 3.33
Achieve Rehabilitation and Nursing Center	1	1	1	1	1	1 Avg = 1
Beacon Nursing and Rehabiliation	1	1	1	1	1	3 Avg = 1.33
Highland Care Center	1	1	1	1	1	2 Avg = 1.16
Kindred Transitional Care And Rehab- Smith Ranch	4	4	4	4	4	4 Avg = 4

Attachment 28 - RN Staffing	Aug-17	Sep-17	Oct-17	Nov-17	Dec-17	Jan-17
Birchwood Terrace	4	4	4	4	4	4 Avg = 4
Achieve Rehabilitation and Nursing Center	2	2	2	2	2	2 Avg = 2
Beacon Nursing and Rehabiliation	2	2	2	2	2	3 Avg = 2.16
Highland Care Center	1	1	1	1	1	2 Avg = 1.16
Kindred Transitional Care And Rehab- Smith Ranch	4	4	4	4	4	4 Avg = 4

OPERATIONS TRANSFER AND SURRENDER AGREEMENT

THIS OPERATIONS TRANSFER AND SURRENDER AGREEMENT, together with all exhibits and schedules (this "OTA"), dated as of June 30, 2017 (the "Execution Date"), is by and among Kindred Healthcare Operating, Inc., a Delaware corporation ("Kindred"); Kindred Nursing Centers East, L.L.C., a Delaware limited liability company, Kindred's affiliated seller entity which operates the Facility ("Transferor"); Birchwood Operations LLC, a Vermont limited liability company ("Transferee"); and E&R Operations LLC, a Vermont limited liability company ("Transferee Guarantor"). Each of Kindred, Transferor, and Transferee may be referred to herein as a "Party" and collectively as the "Parties." Capitalized terms used herein but not defined shall have the same meaning ascribed to such terms in the Purchase Agreement. A glossary of capitalized terms is set forth in Exhibit A attached hereto.

WHEREAS, Transferor is the operator or manager of the skilled nursing facility located at 43 Starr Farm Rd., Burlington, VT 05408-1321 (the "Facility");

WHEREAS, Kindred has agreed to sell (or transfer the relevant leasehold interest in) the real property and assets comprising the Facility to BM Eagle Holdings, LLC ("Purchaser") pursuant to that certain Asset Purchase Agreement dated as of June 30, 2017 (the "Purchase Agreement"), by and between Kindred and Purchaser;

WHEREAS, Transferor owns certain Assets used in connection with the operation of the Facility:

WHEREAS, Transferor desires to divest itself of the operations or management of the Facility that it currently operates and all of its interest in the tangible and intangible property and related other interests relating to the operation and/or management of the Facility, all upon the terms and conditions contained in this OTA;

WHEREAS, concurrent with the Closing under the Purchase Agreement, Transferee has agreed to lease the Facility from Purchaser;

WHEREAS, the Parties hereto desire for Transferor, subject to the satisfaction of the conditions contained herein, to cease operating the Facility and surrender all rights in and to the Facility, except as provided in this OTA, and for Transferee to commence operation of the Facility; and

WHEREAS, the Parties wish to provide for an orderly and lawful transition of the operations of the Facility from Transferor to Transferee.

NOW, THEREFORE, in consideration of the premises, the mutual obligations of the Parties contained in this OTA, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I SURRENDER

1.1 <u>Surrender</u>. Transferor agrees that Transferor's rights and obligations in and to the Facility and all of its rights to occupy or otherwise operate the Facility shall terminate as of the Effective Time, except those rights or obligations which survive or are retained by such Transferor pursuant to this OTA. Transferor agrees to convey, assign and deliver to Transferee the Assets and all of Transferor's right, title and interest in and to the business operations of the Facility, effective as of the Effective Time.

ARTICLE II ASSETS, LIABILITIES AND OTHER MATTERS

2.1 <u>Assets</u>. Upon the terms and subject to the conditions set forth in this OTA, on the Closing Date, and except for the Excluded Assets, to the fullest extent of its interest, Transferor shall (i) transfer to Transferee physical custody (at the Facility) with regard to items in subparagraph "(c)"; and (ii) sell, transfer, convey and/or assign to Transferee, free and clear of all Encumbrances of any nature whatsoever except for Permitted Encumbrances, all of Transferor's right, title and interest in and to the items listed in subparagraphs "a" through "l" (the "Assets").

(a) Reserved;

- (b) all computers, computer equipment and hardware, office equipment, trucks, vehicles and other transportation equipment, parts, supplies and other tangible personal property owned by and in Transferor's possession as of the date of this OTA or acquired by Transferor prior to the Closing Date which are used exclusively in connection with the operation of its Facility;
- (c) software licenses related exclusively to the operation of the Facility, if applicable and to the extent assignable (and if licensor consent to such assignment is required, to the extent such consent is granted), subject to any license transfer fees which would be the responsibility of Transferee;
- (d) all inventory and supplies (excluding those provided by a National Contract) but not less than a quantity of inventory and supplies that is required by Law including, but not limited to, office, foodstuffs, medical, disposables, prescription medications and pharmaceutical inventories and supplies and other inventories, supplies and articles of personal property of every kind and nature attached to or used in connection with the Facility, but only to the extent such inventory and supplies are owned by Transferor (collectively "Inventory");
- (e) all contracts, agreements, leases (excluding real estate leases), purchase orders relating exclusively to the Facility (collectively, the "Contracts"), whether oral or written, to the extent transferable and to the extent expressly assumed in writing by Transferee in its sole discretion; excluding, however, rights, claims or responsibilities thereunder existing and relating to the period of time prior to the Effective Time. Transferee will inform Transferor in writing of the Contracts which it intends to assume by delivering to Transferor a listing of such Contracts within the later of (i) three (3) Business Days after the Execution Date or (ii) ten (10) days of the date of receipt of copies of such contracts. A preliminary list of those Contracts which Transferee has identified as Assumed Contracts as of the Execution Date is attached hereto as Schedule 2.1(e), which Transferee may modify until fifteen (15) days prior to Closing, after which time such Schedule shall be deemed final. All Contracts as to which Transferor receives from Transferee timely notice of assumption are referred to as the "Assumed Contracts." To the extent mutually agreed by Kindred and Transferee, Transferee will also assume therapy Contracts relating to the Facility between any Seller Party and RehabCare or any of its Affiliates, subject to any modifications or amendments Kindred, Transferee and RehabCare negotiate as part of such mutual agreement.
- (f) all menus, operating manuals for equipment at the Facility (but specifically excluding operating policy and procedure manuals), marketing, sales and promotional materials; notwithstanding the foregoing, Transferor shall leave all operating policy and procedure manuals in electronic form (the "Policy and Procedure Manuals") for Transferee to use for a period of one hundred fifty (150) days after the Closing. After one-hundred fifty (150) days (the "Policy Return Date"), Transferee shall return all Policy and Procedure Manuals to Transferor; provided, however, to the extent that Transferee uses the Policy and Procedure Manuals between the Closing Date and the Return Date, Transferee shall be required to place its name thereon and remove the name of Transferor and its

Affiliates; *provided*, *further* that the Parties acknowledge Transferor makes no representation or warranty as to the compliance of Policy and Procedure Manuals with applicable Law, and Transferor shall not be liable for any use thereof by Transferee;

- (g) to the extent of its interest therein, all rights to telephone and facsimile numbers used by the Facility, any "yellow page" and other advertising rights of such Facility, and all of the rights of Transferor in the name used for the Facility (excluding the marks set forth in <u>Exhibit 2.7</u>), for example, "Rehabilitation and Skilled Nursing Center";
- (h) all files, charts, and other information located at the Facility in Transferor's possession or control relating to all (i) current Residents of the Facility as of the Closing Date (including, but not limited to, all resident records, billing and collection records, medical records, therapy records, pharmacy records, clinical records, and Resident Trust Funds records), (ii) Residents who previously occupied the Facility or used the Facility prior to the Effective Time and are not Residents of the Facility as of the Effective Time (including, but not limited to, all patient records, medical records, therapy records, pharmacy records, clinical records, and Resident Trust Funds records) for the period between and including the Effective Time and the date that is three (3) years prior to the Effective Time, (iii) employment records for the Transferee Employees (including all medical and health records and all non-medical records including payroll and schedule records, evaluations, etc.), (iv) administrative compliance records including, but not limited to, all state surveys and plans of correction, and (v) correspondence and any other written data which was utilized in connection with the operation of the Facility or the Business (collectively, "Current Records");
- (i) licenses, certificates, permits, waivers, consents, authorizations, variances, approvals, accreditations, guaranties, certificates of occupancy, utility lease agreements, covenants, commitments, and warranties relating to the Facility and the Assets, if any, issued to or on behalf of Transferor relating to the Assets or the Facility, provided same are transferable and assumed by Transferee ("**Permits**");
 - (j) goodwill;
- (k) such Transferor's right, title and interest as trustee or otherwise to residents/patient funds held in trust (collectively, "Resident Trust Funds") to the extent permitted by Law shall be transferred to Transferee on the Closing Date. Transferee shall accept such assignment on behalf of such a resident/patient and shall indemnify and hold Transferor harmless in connection with any such resident/patient to the extent of the Resident Trust Property received by Transferee;
- (l) to the extent permitted by applicable Law and in accordance with the terms and conditions set forth herein, Transferor's rights and interests in and to its provider number and provider and reimbursement agreement under the Medicare program and Medicaid; and
- (m) all other assets, properties, rights, business and tangible personal property of every kind and nature owned by Transferor on the Closing Date, known or unknown, fixed or unfixed, choate or inchoate, accrued, absolute, contingent or otherwise, whether or not specifically referred to in this OTA relating exclusively to the Facility and its operations to the extent transferable and not expressly excluded pursuant to Section 2.7.

2.2 Nursing Home License; Medicare and Medicaid Provider Agreements.

(a) Transferee will file, within seven (7) days of the Execution Date, an application for a license to operate the Facility (the "New License") with the Vermont Division of Licensing and Protection (the "Department"), and Transferee shall file applications for the Ancillary Permits and

Approvals as and when permitted or required under the laws of the applicable issuing authority. Transferee will provide Transferor with a copy of its filed application for the New License within one (1) Business Day after its filing of the application. Transferee shall diligently proceed to secure the New License and the Ancillary Permits and Approvals and shall (i) from time to time, upon request of Transferor, advise Transferor of the status of Transferee's efforts to secure the New License and the Ancillary Permits and Approvals, (ii) promptly advise Transferor once Transferee has received confirmation of the date on which the New License will be issued, and (iii) promptly upon receipt of a request therefor from Transferor, shall provide Transferor with copies of the document(s) evidencing the New License. For purposes hereof, "Ancillary Permits and Approvals" shall mean all ancillary permits or licenses required for the operation of the Facility from and after the Closing Date including, but not limited to, the Medicare tie in notice and Medicaid provider agreement, business licenses, food service permits, elevator permits, vending machine permits, beauty shop licenses and CLIA waivers. Hereinafter, the New License and the Ancillary Permits and Approvals will be collectively referred to as the "Regulatory Approvals." The Parties will use reasonable efforts to cooperate by providing such information necessary for Transferee to file the application for the Regulatory Approvals contemplated under this Section 2.2.

(b) To the extent permitted by Law:

- (i) Effective as of the Effective Time, Transferor's rights and interests in and to its Medicare provider number and provider and reimbursement agreement (the "Medicare Agreement") shall be assigned to Transferee.
- (ii) Effective as of the Effective Time, Transferee shall have the right to bill Medicaid using Transferor's Medicaid provider number and Transferor's provider and reimbursement agreement (the "Medicaid Agreement").
- Transferee shall provide all notices and make all necessary filings as required under applicable Law in order for Transferee to become the certified Medicare and Medicaid provider at the Facility. So long as Transferee is utilizing its best efforts to become the certified Medicare and Medicaid provider at the Facility, Transferee shall be permitted to bill under the Medicare Agreement and Medicaid Agreement, utilizing Transferor's Submitter ID and NPI numbers and/or Medicaid provider number, as applicable, during the period (the "Transition Period") that commences on the Effective Time and that ends on the earlier of (A) in the case of Medicare, the issuance of the Medicare tie-in notice or in the case of Medicaid, the issuance of the new Medicaid number and related provider agreement to Transferee, or (B) the date which is twelve (12) months following the Effective Time. If, notwithstanding Transferee's continuing best efforts, the Medicare tie-in notice shall not have been issued or a new Medicaid provider agreement shall not have been issued to Transferee within such 12-month period, as applicable, Transferor, upon Transferee's written request, shall agree to such reasonable extensions of the Transition Period as may be necessary for Transferee to complete the applicable certification process. In no event shall Transferee bill under the Medicare Agreement or Medicaid Agreement following expiration of the Transition Period. Transferee shall indemnify and hold Transferor harmless from and against any and all liabilities arising out of Transferee's use of the Medicare Agreement and/or Medicaid Agreement following the Effective Time.
- (c) Following the Execution Date, Transferor shall use its commercially reasonable efforts to provide Transferee with all documents and information necessary for Transferee to seek a novation (the "Novation") of contract number M0003665 (the "VA Contract") between Transferor and the Department of Veterans Affairs (the "VA"). Transferor shall use its commercially reasonable efforts to cooperate with Transferee in seeking the Novation of the VA Contract including, but not limited to, signing any novation agreement and other forms reasonably requested by Transferee or the VA. Transferee shall be entitled to bill under the VA Contract pursuant to that certain Subcontract Pending

Novation substantially in the form attached hereto as <u>Exhibit 2.2(c)</u> (the "VA Subcontract"). For the avoidance of doubt, Transferee will continue to receive payment from the VA and any such payments will be considered Accounts Receivable hereunder, subject to the provisions of <u>Section 6.10</u>. Transferee acknowledges that (i) the Novation is not expected to occur until after the Closing Date and, therefore, the issuance of such Novation shall not be a condition to close, and (ii) Transferee shall reimburse Transferor for all reasonable out-of-pocket expenses (including, without limitation, any audit costs) incurred by Transferor in securing the necessary deliverables for the Novation.

2.3 <u>Transfer of Resident Trust Funds</u>. To the extent permitted by applicable Law at the Closing, Transferor shall deliver to Transferee (a) original copies of the trust fund records, (b) a written statement that sets forth the Resident Trust Funds, and (c) an assignment of the Resident Trust Funds to Transferee. Within ten (10) Business Days following the Closing Date, Transferor shall prepare and deliver to Transferee a true, correct and complete accounting, properly reconciled and balanced, of the Resident Trust Funds as of the Effective Time. Transferee hereby agrees that it will accept such Resident Trust Funds and hold the same in trust for the residents, in accordance with applicable statutory and regulatory requirements.

2.4 Assumption of Liabilities.

- (a) Upon the terms and subject to the conditions set forth in this OTA, at the Effective Time, Transferee agrees to assume the following liabilities relating to the Assets, subject to the provisions of Section 2.4(b): (i) all obligations and liabilities under the Assumed Contracts, (ii) any Taxes with respect to the operation of the Business at the Facility, (iii) all liabilities under the terms of the Permits, and (iv) all obligations and liabilities (in each case, whether or not accrued, whether fixed, contingent or otherwise, and whether known or unknown) pertaining to the operation of the Facility, but in the case of each of clauses (i) through (iv), only to the extent such liabilities relate to the period after the Effective Time (collectively, "Assumed Liabilities").
- Except for the Assumed Liabilities, Transferor shall retain all of its liabilities and (b) obligations of any kind or nature, at any time existing or asserted, whether or not accrued, whether fixed, contingent or otherwise, whether known or unknown, arising out of and by reason of the ownership or operation of the Assets and the Facility prior to the Effective Time. Except to the extent expressly and unambiguously expressed herein to the contrary, Transferee is not the successor to liability of Transferor and is not herein assuming any liability or detriment from, arising from, out of, or relating to, Transferor's ownership of the Assets, the Facility or any activity of Transferor prior to the Effective Time or conduct of Transferor after the Effective Time. Transferee does not and shall not assume (except to the extent included in Assumed Liabilities) any payable of Transferor, governmental claim or charge, liability of any governmental claim or charge, liability for any general liability, malpractice, professional liability, resident rights violations, or violation of employee rights or contracts, whether such claims arise in law, equity, tort, contract, statute, common law, or from any other source or precedent. Without limiting the generality of the foregoing, Transferor shall retain and Transferee shall not assume any (i) Medicaid and/or Medicare liabilities for the period prior to the Effective Time, all of which Transferor agrees to satisfy in full as and when due upon expiration of any applicable period for the contesting or appeal of such liabilities, (ii) accrued expenses which were incurred prior to the Effective Time, (iii) Encumbrances affecting the Assets other than Permitted Encumbrances, (iv) liability or obligation of Transferor arising out of or based upon Transferor's ownership and operation of the Facility prior to the Effective Time, (v) liability or obligation relating to any RAC, ZPIC, or MAC audits as well as any and all investigations from a Governmental Entity or any entity acting with the authority of the foregoing or by a whistleblower or other private citizen claiming a violation of a healthcare related statute or a violation of the Medicare, Medicaid or other third party payor agreement, in each case for the period prior to the Effective Time, (vi) liability or obligation relating to or arising from any Pre-Closing Imposition, (vii) capital repairs or physical improvements required to remove or resolve a Pre-Closing Imposition, and/or (viii) liability or

obligation of Transferor arising out of or based upon Transferor's ownership and operation of the Excluded Assets. Transferor shall retain all of its applicable foregoing liabilities and obligations ("Retained Liabilities").

2.5 <u>Employees and Employee Benefits</u>. The Parties hereby agree that:

- (a) Between forty-five (45) and fifty (50) days prior to the Closing, Transferor shall update the list of employees on <u>Schedule 4.6(a)</u> to reflect new hires and terminations of employment that occurred after the Execution Date regarding the applicable Transferred Facilities and Transferor shall also provide a list of employees on <u>Schedule 2.5(a)</u> of employees who are not subject to offers of employment from Transferee or its Affiliates in <u>Section 2.5(b)</u> below.
- (b) Not less than thirty (30) days prior to the Closing, Transferee shall (or shall cause one of its Affiliates) to offer in writing employment to substantially all of those employees listed on such revised Schedule 4.6(a) who meet Transferee's employment eligibility requirements, effective as of the Effective Time (and subject to such employee's continued employment with Transferor as of immediately prior to the Effective Time), on the terms and conditions set forth in this Section 2.5. Employees who accept Transferee's (or its Affiliate's) offers of employment and commence employment with Transferee are referred to herein as "Transferee Employees." The employment of each Transferee Employee shall be effective as of the Effective Time. Nothing contained in this OTA shall constitute a guaranty of employment or continued employment of any kind for any current or former employee of Transferor, whether or not such employee is hired by Transferee.
- (c) As of 11:59:59 p.m. on the applicable Closing Date, Transferor shall terminate the employment of all employees at the Facility including, without limitation, Persons temporarily absent from active employment by reason of disability, illness, injury, workers' compensation, approved leave of absence or layoff. Transferee's or its Affiliate's offer of employment to Transferor employees pursuant to Section 2.5(b) above shall commence at the Effective Time, such that those Transferor employees who accept employment with Transferee or its Affiliate shall not experience a period of unemployment in connection with the transactions contemplated herein. Notwithstanding the foregoing, Transferor shall be solely responsible for any liabilities related to or arising out of employment of any such employees of Transferor and the termination of employment of such employees by Transferor.
- (d) Not less than thirty (30) days prior to the Closing, Transferee shall (i) identify to Transferor all Transferor employees identified on Schedule 4.6(a) to whom Transferee and its Affiliates will not offer employment, and (ii) identify to Transferor all employees of Affiliates of Transferor (such as, without limitation, dieticians, clinicians, division vice-presidents, sales people, and such other similar positions) providing services to such Transferred Facilities to whom Transferee or its Affiliates would propose to make offers (such individuals under subsection (ii) being "Affiliated-Service Transferee Employees"), subject to consent in the sole discretion of the applicable Transferor Affiliate. To the extent that Transferor's applicable Affiliate provides such consent, then the requirements regarding employment and hiring of Affiliated-Service Transferee Employees by Transferee and its Affiliates will be the same for Transferee and its Affiliates as those for Transferee Employees under this Section 2.5.
- (e) The Parties shall work together in good faith to coordinate reasonably regarding employee changes that occur between the date of the scheduling updates in this Section and the Effective Time so that each Party can update its schedules and records accordingly.
- (f) Except as otherwise required by Law, Transferor shall pay the employees at the Facility in accordance with its standard payroll practice, all earned wages due and payable as of the Closing Date including, but not limited to, paid time off, personal leave, and vacation benefits as of the Effective Time, and any severance, retention bonus or other change in control payment payable to any

Transferred Employee or Affiliated Service Transferee Employee, as applicable, that become due or owed as a result of the consummation of the transactions contemplated by this OTA.

- All Transferee Employees hired by Transferee who accept and commence employment with Transferee following the Effective Time shall be employed by Transferee on an "at will" basis. Transferee shall initially employ Transferee Employees on the following terms and conditions in such manner as not to trigger WARN Act liability: (i) comparable base salary or rates of pay as in effect immediately prior to the Closing Date for employees of similar tenure performing comparable services at Transferee's other skilled nursing facilities, and (ii) employee benefits that are comparable in the aggregate to the benefits that are provided by Transferee to its employees under the Plans at its other skilled nursing facility operations. In furtherance and not in limitation of the foregoing, Transferee shall treat prior service with Transferor as service with Transferee for purposes of determining eligibility to receive and participate in all benefits programs maintained by Transferee. It is understood that Transferee shall not be responsible to pay any disability or workers' compensation benefits to or for any Transferor's employee who is receiving such benefits or who experienced a disability or injury covered under Transferor's or Seller's benefit plans or workers compensation insurance program on or before the Closing, and that Transferor or Seller, as applicable, shall continue to be responsible for payment of such benefits until such obligation terminates under the applicable benefit plans or Laws. Transferee, at reasonable times in advance of Closing upon prior written notice to and coordination with Transferor, shall be entitled to meet with the employees of the Facilities and distribute employment applications and benefit enrollment packages. This OTA shall not create and shall not be deemed to create or grant to any Transferred Employee any third party beneficiary rights or claims or any cause of action of any kind or nature.
- (h) Pursuant to Treasury Regulations Section 1.409A-1(h)(4), Transferor and Transferee agree that on the Closing Date, each Transferee Employee shall be treated as having a "separation from service" for purposes of Section 409A of the Code and Treasury Regulations Section 1.409A-1(h).
- (i) Subject to <u>Section 2.5(m)</u>, Transferor or Seller shall retain the liability for the claims respecting all employees of Transferor (including the Transferee Employees) that are incurred under any Plan prior to the Effective Time.
- (j) Transferor, Seller and, to the extent applicable, Seller Affiliates, expressly assume and retain any liability arising under COBRA with respect to any "M&A qualified beneficiaries" (as that term is defined in the COBRA regulations). To satisfy this liability, such party shall continue to maintain group health plan coverage for at least eighteen (18) months after the Closing and shall not terminate its COBRA obligations by terminating its group health plan prior to the expiration of such minimum eighteen (18) month period after Closing.
- (k) Transferee shall be responsible for any and all liabilities arising out of or with respect to any Transferee Employee arising with respect to employment by Transferee after the Effective Time or attributable to events or circumstances occurring after the Effective Time.
- (1) The Parties acknowledge and agree that all provisions contained in this Section 2.5 with respect to employees are included for the sole benefit of the respective Parties and shall not create any right (i) in any other Person, including any employees, former employees, any participant in any Plan or Transferee Plan or any beneficiary thereof, or (ii) to continued employment with any Transferor or Transferee, or particular benefits or coverage in any Plan or Transferee Plan. For the avoidance of doubt, (A) the provisions of this Section 2.5 shall not constitute an amendment to any Plan or Transferee Plan, and (B) in no event shall any employee, former employee, any participant in any Plan

or Transferee Plan or any beneficiary thereof or any other Person described herein be a third party beneficiary for purposes of this OTA.

- Immediately following the Closing, Transferee shall provide Transferee Employees who accept employment with Transferee, as well as eligible dependents of such employees (collectively, with the Transferee Employees, "Affected Participants") the opportunity to participate in the applicable employee benefit plans, programs or policies maintained or established by Transferee that are comparable to the plans and benefits Transferee provides at its other skilled nursing facility operations (each, a "Transferee Plan"), which may include medical, dental, vision, and/or any other applicable group medical plan, program, insurance coverage or arrangement (collectively "Group Health Plan"). If elected, the benefits offered to such employees must extinguish Transferor's COBRA insurance coverage obligations. With respect to each Transferee Plan in which Affected Participants become eligible to participate, subject to the consent of any applicable insurer, Transferee shall: (i) waive any eligibility waiting periods, any evidence of insurability requirements and the application of any pre-existing condition limitations under such Plan, except to the extent that such waiting period, evidence of insurability requirement, or pre-existing condition limitations would not have been satisfied or waived under the comparable Plan in which the Affected Participant participated immediately prior to the Effective Time; and (ii) provide each Affected Participant credit for copays and deductibles under any Transferee Group Health Plan to place each Affected Participant in the same place that such Affected Participant was in under the Transferor's Group Health Plans immediately prior to the Closing.
- Consents to Assignment. Notwithstanding anything to the contrary contained herein, this 2.6 OTA shall not constitute an agreement to assign or transfer any contract or lease, or any claim, right or benefit arising thereunder or resulting therefrom, if an attempted assignment or transfer thereof, without the consent of a third party thereto or of the issuing Governmental Entity, as the case may be, would constitute a breach thereof. The Parties shall cooperate to obtain any consents of any parties necessary to permit the assignment of the Assumed Contracts. Transferor and Transferee acknowledge that certain of the Assumed Contracts may not, by their terms, be assignable and, accordingly, none of such nonassignable Assumed Contracts shall be deemed assigned to or assumed by Transferee unless and until the same shall become so assignable. If and when any necessary consent shall be obtained or any such Assumed Contract shall otherwise become assignable, Transferor shall take all necessary action to assign all of its rights and obligations thereunder to Transferee and Transferee shall, without the payment of any pre-closing liabilities, assume such rights and obligations. Until such time as the Assumed Contracts are assigned to Transferee, Transferor shall not enter into any amendments of such non-assignable Assumed Contracts without the prior written consent of Transferee. Until such time as the non-assignable Assumed Contracts are assumed by Transferee, (a) Transferee shall perform and discharge fully all of the obligations of Transferor under any of such non-assignable Assumed Contracts to the extent the same would have constituted assumed liabilities if the Assumed Contracts had been assumed by Transferee as of the Closing Date and to the extent the same relate to the period of time after the Closing Date, and Transferee shall indemnify, hold harmless, protect and defend Transferor and its respective officers, employees, managers and members, from and against any and all damages, demands, costs, expenses and liabilities arising out of Transferee's failure to make payments or perform any other obligations which relate to the period of time after the Closing Date occurring under the Assumed Contracts or nonassignable Assumed Contracts after the Closing Date, and (b) Transferor shall, without further consideration therefore, pay, assign and remit to Transferee promptly all monies received or which may be received or obtained in respect of the non-assignable Assumed Contracts related to periods after the Closing Date and Transferor shall take such reasonable actions as shall be necessary to confer on Transferee any other benefits that may be available under such non-assignable Assumed Contracts.
- 2.7 <u>Excluded Assets</u>. Transferee shall not purchase, and Transferor shall retain, any right, title and interest in the assets listed on <u>Exhibit 2.7</u> (collectively, "*Excluded Assets*").

ARTICLE III THE CLOSING

- Time and Place of Closing. Subject to the satisfaction or waiver of all of the conditions precedent set forth in this OTA, the consummation of the transactions contemplated under this OTA (the "Closing") shall take place on the same date as the closing with respect to the Facility under the Purchase Agreement, or on such other date as shall be mutually agreed upon by the Parties hereto and Purchaser. The date on which the Closing occurs is referred to herein as the "Closing Date." Notwithstanding the actual time at which the Closing occurs, the time ("Effective Time") as of which the Closing shall be deemed to be effective and the risk of loss shall pass from Transferor to Transferee shall be 12:00:01 a.m. (local time where the Facility is located) on the day after the Closing Date.
- 3.2 <u>Closing Matters.</u> Upon the terms and subject to the conditions set forth in this OTA, at the Closing:
- (a) Transferor shall deliver to Transferee a bill of sale in the form attached hereto as Exhibit 3.2(a) (the "Bill of Sale") and such endorsements, assignment instruments, and other instruments of transfer and conveyance as shall be reasonable or necessary to convey, transfer, assign and deliver the Assets to Transferee pursuant to the terms of this OTA and to convey, transfer, assign and deliver any and all interest it has in the furniture, fixtures and equipment of the Facility to the fee purchaser under the Purchase Agreement (if applicable);
- (b) Transferor shall deliver an assignment of its interests to the items listed in <u>Article II</u> including, without limitation, the Assumed Contracts, pursuant to an assignment and assumption agreement in the form attached hereto as <u>Exhibit 3.2(b)</u> (the "Assignment and Assumption Agreement");
- (c) Transferor and Transferee shall execute and deliver to the other a Certificate in the form attached hereto as Exhibit 3.2(c) ("Bring Down Certificate");
- (d) Transferor and Transferee shall execute a closing statement with respect to the prorations contemplated by <u>Section 3.3</u> hereof, and the Party owing pursuant to such statement shall pay the amount due in immediately available funds at Closing;
- (e) Not later than ten (10) days after the Closing, Transferor shall execute and deliver to Transferee a detailed schedule and an assignment of all Resident Trust Funds;
- (f) Transferor shall execute and deliver to Transferee an assignment of any and all security deposits or advance deposits from Residents, and Transferee shall deliver to Transferor a written receipt for such funds indicating that Transferee is accepting such funds in trust for the Residents;
- (g) Transferor shall deliver to Transferee, to the extent that they are not posted at the Facility, certificates, licenses, permits, authorizations and/or approvals issued for or with respect to such Facility by any Governmental Entity;
- (h) Transferor shall deliver to Transferee, or leave at the Facility, the originals (or copies) of all Assumed Contracts in effect on the Closing Date;
- (i) Transferor shall deliver to Transferee a current and complete list of the names of each Resident in the Facility;

- (j) Not later than ten (10) days after the Closing, Transferor shall deliver a detailed Accounts Receivable aging as of the Effective Time noting all balances owed to Transferor for dates of service prior to the Effective Time as described in <u>Section 6.10</u> below;
- (k) Transferor shall provide to Transferee an updated <u>Schedule 4.6(a)</u> that lists Transferor's employees as of the Closing Date;
- (l) Transferor shall deliver an Assignment of Admission Agreements, a Termination of any existing real property lease where a Kindred affiliate is the Fee owner or Ground Lessee, the IMA, where applicable, and keys to the Facility; and
- 3.3 <u>Closing and Post-Closing Adjustments: Costs and Prorations</u>. In addition to any other items agreed upon by the Parties, the following items are to be apportioned between Transferor and Transferee on a *pro-rata* basis as to ownership as of the Closing Date, in accordance with the general principle that Transferor shall be entitled to the revenue attributable to, and responsible for such expenses and obligations attributable to, the Facility for the period up to and including the Closing Date, and Transferee shall be entitled to the revenue attributable to, and responsible for such expenses and obligations attributable to, the conduct of the Facility after the Closing Date:
- (a) Water, gas, electric, telephone and other utility charges, and sewer and waste water charges, shall be prorated, to the extent possible prior to the Closing, as of the applicable Closing Date. For metered service, Transferor shall pay or cause to be paid the utility bills for services rendered prior to the readings, and Transferee shall pay the utility bills for the services rendered after the readings. If any metered utility is read on any day other than the applicable Closing Date, Transferor and Transferee shall prorate such utility charges consistent with the most recent bills, and then reconcile following the Closing as provided in Section 3.3(i). In furtherance of the foregoing, Transferee shall work to transition the utilities serving the Facility into the name of Transferee effective as of the Effective Time, and Transferor shall reasonably cooperate with Transferee.
- (b) Subject to and consistent with <u>Section 6.10</u>, all revenue (including rent and Residents' occupancy fees) attributable to any period ending on or prior to the Closing Date shall belong to Transferor, and all revenue (including rent and Residents' occupancy fees) attributable to any period after the Closing Date shall belong to Transferee.
- (c) For expenses of the Facility, Transferee shall remit to Transferor any invoices which reflect a service or delivery date on or before the applicable Closing Date, and Transferee shall assume responsibility for the payment of any invoices which reflect a service or delivery date after the Closing Date. Notwithstanding the foregoing, Transferee acknowledges and agrees that it shall have no right, title or interest in and to any retroactive workers compensation insurance program payments whether or not the same are paid prior to or after the Closing Date if and to the extent they relate to any period prior to the Closing Date.
- (d) Any and all deposits of Transferor with respect to the Facility including, without limitation, any and all equipment leases, security and/or utility deposits paid to and/or cash or other collateral held by any equipment lessor or by any utility, insurance company or surety, shall remain the sole and exclusive property of Transferor, and Transferee shall have no right or interest therein or thereto, and to the extent that Transferor does not receive a return of any such deposit on the Closing Date and such security deposit has been assumed by Transferee, Transferee shall reimburse Transferor on the Closing Date the full amount of any such security deposit assumed by Transferee.
- (e) Any bed Tax or similar provider Taxes or fees shall be *pro-rated* between Transferor and Transferee based on the period of its operation of the Facility occurring before and after

the Closing Date, as the case may be, including, but not limited to, any such assessments made by the State in which the Facility is located and/or paid by Transferor prior to or on the Closing Date that would apply to operation of the Facility after the Closing Date.

- (f) In the event that there is a governmental assessment against the property upon which the Facility is situated, Transferor shall be responsible for that which relates to the period prior to the Effective Time and Transferee shall be responsible for that which relates to the period after the Closing Date.
- (g) Transferor shall pay the reasonable cost of Transferee's compliance with those physical plant Life Safety Code ("LSC") Deficiencies and/or fire safety standards violations that were identified on a survey of the Facility as a deficiency and that was not cured by Transferor prior to the Closing ("Deficiencies").
- (h) Transferor shall be responsible for any and all fines, and the reasonable cost to bring the Facility into substantial compliance to the extent the Facility is cited for compliance violations attributable to the operation of the Facility prior to Closing by any Governmental Authority or Government Reimbursement Program which relates to equipment, furniture, fixtures or the condition of the building or grounds, or operational issues (collectively, "Compliance Violations"). For the avoidance of doubt, Deficiencies and Compliance Violations will not include (i) any violations where a plan of correction has been accepted by a Governmental Authority, except for (1) liabilities or obligations relating to or arising from any Pre-Closing Imposition, or (2) capital repairs or physical improvements required to remove or resolve a Pre-Closing Imposition; or (ii) any citation for which Transferor currently holds a waiver issued by a Governmental Authority.
- (i) All such prorations shall be made on the basis of actual days elapsed in the relevant accounting, billing or revenue period and shall be based on the most recent information available to Transferor and Transferor and Transferor and Transferor shall reasonably cooperate to produce prior to the Closing a schedule of prorations to be made under this Section 3.3 at the Closing that is as complete and accurate as reasonably possible. All prorations that can be accurately or reasonably estimated as of the Closing shall be made at the Closing. All other prorations, and adjustments to initial estimated prorations, shall be made by the Parties with due diligence and cooperation within one hundred twenty (120) days following the Closing, or such later time as may be required to obtain necessary information for proration, by payment in immediately available funds by wire transfer to one or more bank accounts designated in writing of the Party yielding a net credit from such prorations from the other Party.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF TRANSFEROR

In order to induce Transferee to enter into this OTA, Transferor hereby represents and warrants to Transferee as of the Execution Date and as of the Closing Date (or in the case of representations and warranties that by their terms speak as of a specified date, as of such specified date), as to itself and the Facility only, as follows:

4.1 Corporate.

- (a) <u>Organization</u>. Transferor is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization.
- (b) <u>Power and Authority; Authorization; Enforceability</u>. Transferor has all necessary corporate, partnership or similar power and authority to own, operate and lease the Assets, and to carry on its business as and where such is now being conducted, including the Business. Transferor has all

necessary corporate, partnership or similar power and authority to enter into the documents and instruments to be executed and delivered by Transferor pursuant hereto and to carry out the transactions contemplated hereby. The execution and delivery of this OTA and the performance of this OTA by Transferor has been duly and validly authorized. This OTA constitutes the legal, valid and binding obligation of Transferor, enforceable against Transferor in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar Laws and equitable principles relating to or limiting creditors' rights generally.

- (c) <u>Qualification</u>. With respect to the Business, Transferor is duly qualified or licensed to do business, and is in good standing, in all jurisdictions (domestic and foreign) in which the character or the location of the assets owned or leased by it or the nature of the business conducted by it requires such licensing or qualification.
- (d) No Conflicts or Violations. Subject to obtaining the consents, neither the execution and delivery of this OTA, the consummation of the transactions contemplated hereby and thereby, nor the fulfillment of the terms hereof by Transferor shall (i) violate or result in a breach of any of the material terms and provisions of, constitute a default under, conflict with, or result in any acceleration of rights, benefits or obligations of any party under any Assumed Contract to which Transferor is a party or by which it is bound, (ii) violate any Order of any Governmental Authority applicable to Transferor, (iii) result in the creation of any material Encumbrance upon any Asset pursuant to the terms of any such Assumed Contract, (iv) constitute a violation by Transferor of any applicable Law, (v) result in the breach of any of the material terms or conditions of, or constitute a default under, or otherwise cause any impairment of, any permit, license or other governmental authorization held by Transferor, or (vi) conflict with or violate any organizational document of Transferor, except in the case of sub-clauses (i), (ii), (iii), (iv) and (v), to the extent that any such violation, breach or Encumbrance would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Facility.
- 4.2 <u>Notices</u>. Except as listed in <u>Schedule 4.2</u>, neither the Facility nor Transferor has received, during the last two (2) years, written notice: (a) that the Facility will be subject to a rate reduction for Medicaid or Medicare services provided therein as a result of a Medicare or Medicaid audit; or (b) from any Governmental Entity that results in a Material Adverse Effect to the Facility or identifies that the Facility is in violation of Law which has not been cured, except where such violation would not reasonably be expected to result in a Material Adverse Effect. Transferor shall disclose any rate reduction or bed count reduction of which it becomes aware and any rate reduction proposal or bed count reduction proposal of which it becomes aware that will affect the Facility.
- 4.3 <u>Litigation</u>. Except as disclosed on <u>Schedule 4.3</u>, there is no Action pending or, to Transferor's Knowledge, threatened against Transferor with respect to the Assets or the Business. Transferor is not subject to any Order relating to the Assets or the Business.

4.4 Taxes.

- (a) All Returns required to be filed by or on behalf of Transferor on or before the Closing Date with respect to the Business or the Assets have been duly and timely filed (or subject to proper extensions) with the appropriate taxing authority in all jurisdictions in which such Returns are required to be filed, and all such Returns are true, complete and correct in all material respects.
- (b) All Taxes of Transferor shown on any such Return with respect to the Business and the Assets that are due and payable on the Execution Date have been fully and timely paid.

(c) There are no Encumbrances for Taxes upon the Assets other than statutory liens for Taxes not yet due or payable.

4.5 <u>Employee Benefit Plans</u>.

- (a) Schedule 4.5(a) lists all Employee Benefit Plans including, without limitation, any welfare plan within the meaning of Section 3(1) of ERISA, or any pension plan within the meaning of Section 3(2) of ERISA, that Transferor or Affiliate sponsors, maintains, contributes or is obligated to sponsor, maintain, or contribute to the benefit of any current or former employees or other service provider of the Business (or any dependent or beneficiary thereof), or under which Seller, Kindred, or Transferor has any material liability with respect any current or former employee or other service provider of the Business (each, a "Plan"). Transferor has delivered or otherwise made available in the VDR to Transferee true, accurate and complete copies of each Plan (or, if the Plan has not been reduced to writing, a written summary of all material terms).
- (b) Each Plan that is intended to be qualified under Section 401(a) of the Code (each a "Qualified Plan") has received a favorable determination, opinion, or advisory letter from the IRS indicating that such Qualified Plan (or the master, prototype, or volume submitter form on which it is established) is so qualified under the Code in form, and no fact or event has occurred since the issuance of the most recent such letter that creates a material risk of revocation of any such letter. Transferor has made available to Transferee a copy of such determination, opinion, or advisory letter.
- (c) Neither Transferor nor any ERISA Affiliate thereof currently sponsors, maintains, or contributes to (or has an obligation to sponsor, maintain or contribute to) or sponsored, maintained or contributed to (or had an obligation to sponsor, maintain or contribute to) any "employee pension benefit plan" (as defined in Section 3(2) of ERISA) for which Transferor or any ERISA Affiliate has any liability covering employees of the Business that is subject to Title IV of ERISA or Code Section 412, including any "multi-employer plan" as defined in Section 3(37) or 4001(a)(3) of ERISA, or any "multiple employer plan" subject to Section 4063 or 4064 of ERISA. "ERISA Affiliate" means any Person that is considered a single employer with such Person under Section 414(b), (c), (m) or (o) of the Code, any Transferor, Transferor Affiliate or Joint Venture Interest.
- (d) With respect to each Plan, Transferor, or its Affiliates, as applicable, have funded, administered and maintain each Plan in material compliance with all applicable Laws, including ERISA and the Code, and there is no litigation or proceeding pending (other than routine claims for benefits) or, to the Knowledge of Transferor, threatened or anticipated with respect to any Plan in connection with the Business.
- (e) Except as required under COBRA, no Plan provides or promises benefits, including death or medical or other health-related benefits, with respect to current or former Service Providers of the Business beyond retirement or other termination of service, and Transferor or its Affiliates have no obligation to provide or contribute toward the cost of any such benefits.
- (f) Neither the execution and delivery of this OTA and any related documents nor the consummation of the transactions contemplated hereby will, either alone or in combination with any other event, (i) increase any benefits payable under any Plan, including acceleration of the payment or vesting of any benefit under any Plan, or (ii) entitle any employee to severance payable by Transferee or any Affiliate thereof; *provided*, *however*, that benefits due and payable from a Plan may be paid or made available due to a termination of employment from Transferor in the ordinary course of administration of any such Plan.

4.6 Employees.

- (a) Schedule 4.6(a) contains a true and correct list of (i) all of the current employees of the Business as of June 12, 2017, including those employees on a leave of absence of any kind, (ii) each such employee's name, title, and location of employment, (iii) each such employee's employment status (i.e., whether employees is actively employed or not actively employed due to illness, short-term disability, sick leave, authorized leave of absence, layoff for lack of work or service in the Armed Forces of the United States or for any other reason), (iv) each such employee's hourly wage rate, salary level or annual rate of compensation, including bonuses and incentive pay, (v) the hours worked by each such employee during the preceding twelve months, the exempt or non-exempt status of each employee (whether or not paid at an hourly or salary rate), (vi) each employee's date of hire or commencement of most recent employment, (vii) a description of any fringe benefits (other than the standard fringe benefits offered by Transferor to all qualifying employees), and (viii) existing contractual arrangement with employees, if any (it being understood that all Parties do not consider any "at-will" arrangements with employees to be Contracts).
- (b) Except as disclosed on <u>Schedule 4.6(b)</u>, all salary, wages, commissions, bonuses and other cash compensation due and payable to employees of Transferor on or prior to the date hereof, as of the Closing Date, shall be paid in full on or promptly following the Closing Date in accordance with Transferor's standard payroll practices.
- (c) All workers directly engaged by Transferor (excluding any engagement through a staffing agency) and classified by Transferor as independent contractors since January 1, 2014, have in good faith satisfied the requirements of applicable Law to be so classified, and Transferor has in good faith fully and accurately reported each such person's compensation on IRS Forms 1099 during such period when required to do so.
- Transferor has complied, in all material respects, with all applicable Laws pertaining to labor or employment practices or relations (including, but not limited to, the terms and conditions of employment, management-labor relations, employee classification, records retention, equal opportunity employment, non-discrimination, disability accommodation, human rights, statutory and regulatory employer notice requirements (including requirements pursuant to the Fair Credit Reporting Act, the United States Immigration and Nationality Act, as amended, federal, state, provincial and local minimum wage laws, regulations and ordinances, federal and state family, medical and military leave laws and regulations, occupational safety and health laws and regulations, and any other similar applicable Law mandating employer notice of employer and/or employee rights and responsibilities under such Law), statutory and contractual leaves of absence, wage and hour issues, immigration, occupational safety and health, workers' compensation, pay equity and human rights and the employment or termination of employment of their employees, including all such Laws relating to equal employment opportunities, payment of wages (including, but not limited to, payment of hourly wages, overtime, salaries, commissions, bonuses, profit sharing, unemployment compensation, benefits, and vacation, sick or other earned time off benefits due and payable to such employees under any policy, practice, Contract, program or applicable Law) or illegal discrimination).
- (e) Except as set forth in <u>Schedule 4.6(e)</u>, there are no outstanding, pending or, to Transferor's Knowledge, threatened, actions, causes of action, claims, complaints, grievances, demands, orders, prosecutions, or suits against Transferor (including its and their respective directors, officers, agents, or employees) claiming that Transferor has violated any applicable employment Laws before any Governmental Authority or labor relations board, including the National Labor Relations Board, the Department of Labor, and the Equal Employment Opportunity Commission regarding any employees of the Business. No written notice has been received by Transferor of the intent of any Governmental

Authority responsible for the enforcement of labor or employment Laws to conduct an investigation of Transferor regarding any employees of the Business and no such investigation is in progress.

- (f) Except as set out on <u>Schedule 4.6(f)</u>, Transferor is not a party to any collective bargaining agreement relating to the Business.
- (g) Except as set forth in <u>Schedule 4.6(g)</u>, since January 1, 2014, Transferor has not experienced any labor disputes, any union organization attempts or any work stoppages, walk outs, strikes, or lock outs due to labor disagreements. There are no unfair labor practice charges or complaints pending or threatened against Transferor. There is no labor strike, dispute, request, petition or pending election for representation, slowdown or stoppage pending, or to Transferor's Knowledge, threatened or anticipated against or affecting Transferor.
- (h) Transferor maintains an Employment Eligibility Form on Form I-9 for each employee currently employed in the United States in accordance with applicable Law.
- (i) Except as set forth in <u>Schedule 4.6(i)</u> or with respect to employees subject to a collective bargaining agreement, all employees of Transferor are employed at-will, may be terminated at any time with or without notice and for any reason or no reason at all, without material cost or penalty to Transferor.
- (j) Except with respect to transactions contemplated by this OTA, Transferor has not implemented any employee layoffs that could implicate the WARN Act or any similar applicable foreign, state or local Law, and no such events are currently planned, anticipated or announced. To Transferor's Knowledge, no officer or executive of Transferor (i) has any present intention to terminate his or her employment within the first twelve (12) months following the Closing Date, or (ii) is a party to any confidentiality, non-competition, proprietary rights or other such agreement that would materially restrict the performance of such employee's employment duties, or the ability of Transferor to conduct the Business.

4.7 Encumbrances.

- (a) Transferor has good and marketable title to, or in the case of personal property held under a lease or other Assumed Contract (subject to the terms of the lease or other Assumed Contract), an enforceable leasehold interest in, or right to use, the Assets, and none of the Assets are subject to any Encumbrance other than Permitted Encumbrances.
- (\$100,000.00), all contractors, subcontractors and other Persons furnishing work, labor, materials or supplies for the development and construction of the Facility and/or Assets have been paid, or prior to Closing shall be paid, whether the work is in progress or completed, for all work performed, material, supplies and the like up to and including the Closing Date, and there are no claims against any Transferor or Facility, or any of the Assets in connection therewith which may give rise to a mechanic's lien against any Facility, the Assets or any portion thereto, except as set forth on Schedule 4.7(b).
- (c) Except as disclosed on <u>Schedule 4.7(c)</u>, there are no Encumbrances against Transferor's A/R, provider agreements, bank accounts or licenses, and to the extent there are such liens extant, the obligations they secure will be paid in full at Closing and the liens on such assets will be released at Closing unless otherwise indicated as or such constitute an Assumed Liability.

4.8 Certain Healthcare Matters; Compliance Generally.

(a) Government Reimbursement Programs.

- (i) Except as set forth on Schedule 4.8(a)(i), the Facility is (A) qualified for participation in, and has current and valid provider contracts with, the applicable Government Reimbursement Programs and/or their fiscal intermediaries or paying agents in which the Facility participates, all of which Government Reimbursement Programs are listed on Schedule 4.8(a)(i), and is in compliance with the conditions of participation or requirements applicable with respect to such participation, and (B) eligible for payment under the applicable Government Reimbursement Programs in which the Facility participates for services rendered to qualified beneficiaries.
- (ii) All Cost Reports required to be filed for each of the Facility have been prepared and filed in good faith in accordance with applicable Laws when due, and are true, correct, and complete in all material respects (and true and complete copies of Cost Reports for the past three (3) fiscal years have been set out in the VDR).
- (iii) All amounts shown as due from the Facility in the Cost Reports either were remitted with such Cost Reports or will be remitted when required by applicable Law and are appropriately reflected in the Financial Statements, and all amounts shown in the Notices of Program Reimbursement as due have been, or prior to the Closing will be, paid when required by applicable Law.
- (iv) Except as set forth on Schedule 4.8(a)(iv), Transferor has not, during the last two (2) years, received written notice of any dispute or claim by any Governmental Authority, fiscal intermediary or other Person regarding the Facility and the Government Reimbursement Programs or the participation by the Facility in such Programs that have not been cured, and to Transferor's Knowledge, there are no (A) threatened recoupment claims for services provided by the Business, or (B) threatened suspensions, terminations, or restrictions to any contracts with Government Reimbursement Programs and/or their fiscal intermediaries or paying agents.
- (v) Except as set forth in <u>Schedule 4.8(a)(v)</u>, there are no (A) current, pending or outstanding Government Reimbursement Program audits or appeals, (B) Cost Reports that are subject to audits, (C) Cost Reports that remain "open" or unsettled, and (D) current or pending Government Reimbursement Program or Private Program recoupment efforts (other than those conducted in the ordinary course), in each case with respect to the Facility.
- Licenses. The Facility's skilled nursing facility license and CLIA waivers are set (b) forth on Schedule 4.8(b)(i), and all other material Licenses applicable to the Business are located at the Facility. All such Licenses are all of the material Licenses necessary for the ownership and operation of the Facility as currently conducted. Such Licenses are in full force and effect, have not been pledged as collateral security, no proceeding is pending or, to Transferor's Knowledge, threatened, seeking the revocation or limitation of any such License. The Facility is duly licensed as a skilled nursing facility or assisted living facility, as applicable, as required under the Laws in the State where the Facility is located, for at least that number of beds as currently listed on the Licenses. Schedule 4.8(b)(ii) sets forth a true, correct, and complete list of the number and types of licensed beds at the Facility and whether such beds are Medicaid and/or Medicare certified. Except as set forth on Schedule 4.8(b)(iii), there are no proceedings or actions pending or, to Transferor's Knowledge, contemplated to reduce the number of licensed or certified beds of any Facility. Except as set forth on Schedule 4.8(b)(iv), Transferor has not received written notice of any violations of the LSC, fire, building and other applicable codes, ordinances, current zoning requirements, rules, and regulations that have not been cured or for which Transferor has received a waiver under applicable Law. Schedule 4.8(b)(v) sets forth a complete and accurate list of

LSC waivers, decertification proceedings, licensure revocations, and termination and suspension proceedings for the past two (2) years.

- Compliance Generally. The Facility has been operated in compliance in all material respects with all applicable Laws, including all Healthcare Requirements, governing the conduct or operation of the Business, and Licenses. Except as set forth on Schedule 4.8(c)(i), Transferor has not received any written notice of any violation of any such Law or License that has not been cured and, to Transferor's Knowledge, no notice of such violation has been threatened, except where any such violation would not have a Material Adverse Effect on the Facility. There are no outstanding or, to Transferor's Knowledge, threatened or potential Order, subpoena, or investigation from any Governmental Authority, whistleblower suits, or suits brought pursuant to federal or state False Claims Acts or Laws relating to any Governmental Reimbursement Program of or relating to any alleged or actual, violation of any Laws. Except as set forth on Schedule 4.8(c)(ii), there have been no written notices of violations of Referral Laws relating to the operation of the Facility, nor to Transferor's Knowledge are there any conditions at the Facility which would reasonably be expected to cause a violation of Referral Laws or analogous state statute. To Transferor's Knowledge, Transferor has not (i) made any contributions, payments or gifts to or for the private use of any governmental official, employee or agent where either the payment or the purpose of such contribution, payment or gift is illegal under the laws of the United States or the jurisdiction in which made, (ii) established or maintained any unrecorded fund or asset for any such purpose or made any false or artificial entries on its books, (iii) given or received any payments or other forms of remuneration in connection with the referral of patients that would violate the Referral Laws or any analogous state statute, or (iv) made any payments to any person with the intention or understanding that any part of such payment was to be used for inducing a referral or any purpose other than that described in the documents supporting the payment. Transferor has instituted, and the Facility is operated in compliance in all material respects with, a compliance plan which follows all applicable Healthcare Requirements.
- Convictions; Exclusions. Except as set forth on Schedule 4.8(d), neither (d) Transferor nor any current director, officer, or managing employee, is or has been party to a corporate integrity agreement, corporate compliance agreement, or other settlement agreement with the Office of the Inspector General of the United States Department of Health and Human Services, the Centers for Medicare & Medicaid Services, the United States Department of Justice, any Medicaid Fraud Control Unit, or any state Attorney General, as a result of an alleged violation of any Healthcare Requirements (and the Business, Facility, and Assets are in no way subject to or liable with respect to any such corporate integrity agreement, corporate compliance agreement, or other settlement agreement). Neither Transferor nor any current director, officer, or employee has been excluded from participating in the Medicare program or any other Government Reimbursement Program. No current officer, director, or managing employee (as that term is defined in 42 U.S.C. § 1320a-5(b)) of Transferor has been (i) excluded from participating in the Medicare program or any other applicable Government Reimbursement Program; (ii) subject to sanction pursuant to 42 U.S.C. § 1320a-7a or 1320a-8; or (iii) to Transferor's Knowledge, convicted of, a criminal offense under or in connection with (A) the Referral Laws, (B) any Law relating to fraud, theft, embezzlement, breach of fiduciary responsibility, or other financial misconduct in connection with the delivery of a health care item or service or with respect to any act or omission in a program operated by or financed in whole or in part by Governmental Authority, (C) any Law relating to the unlawful manufacture, distribution, prescription, or dispensing of a controlled substance, (D) any Law relating to the interference with or obstruction of any investigation into the criminal offenses described herein, or (E) any offense which would permit the exclusion of any Facility from a Government Reimbursement Program.
- (e) <u>Audits; Settlements</u>. Transferor has provided to Transferee true and complete copies of all survey reports, notices, and waivers of deficiencies, plans of correction, and any other investigation reports issued with respect to the Facility together with material correspondence with any

Governmental Authority issued within three (3) years of the Execution Date concerning the Facility by any Governmental Authority, ZPIC audit, RAC auditor or other contract auditor on behalf of a Governmental Authority, an identification of any material settlement agreements and, to Transferor's Knowledge, any material unresolved matters raised in writing by any such Governmental Authority, RAC auditor or other contract auditor on behalf of a Governmental Authority. Except as set forth on Schedule 4.8(e)-1, (i) Transferor has not had any cited deficiencies on its most recent survey (standard or complaint) that have resulted in a written notice of civil money penalties or a denial of payment for new admissions as of the Closing Date that have not been cured, and (ii) Transferor has not had any deficiencies at "level G" or above, or an IJ at Level I or above on its most recent survey (standard or complaint) that have not been cured. All deficiencies and violations cited in any survey or resurvey have been corrected or corrective action plans have been submitted and approved therefor and will be fully implemented/corrected prior to Closing. Except as set forth in Schedule 4.8(e)-2, the Facility is not currently designated as a Special Focus Facility (as such term is defined by the Centers for Medicare and Medicaid Services Special Focus Facility Program), and has not received any written notice of inclusion or intended inclusion as a Special Focus Facility.

- 4.9 <u>Resident Agreements</u>. A copy of the current standard form of Admission Agreement used by the Facility has been provided to Transferee or otherwise made available to Transferee in the VDR. To Transferor's Knowledge, there are no other agreements with residents of the Facility which materially deviate from the standard form.
- 4.10 <u>Absence of Changes</u>. Except as otherwise disclosed in <u>Schedule 4.10</u> or the other Schedules, or as contemplated by this OTA, from January 1, 2017, to the Execution Date, (a) the Business has been conducted in all material respects in the ordinary course consistent with past practice, and (b) to Transferor's Knowledge, there has been no change, event, or loss affecting the Business that has had a Material Adverse Effect on the Facility.
- 4.11 <u>Inventory</u>. On the Closing Date, Transferor shall maintain its normal inventory of supplies, which will be in sufficient quantities of supplies required by Law in all material respects and consistent with past practices for operation of the Facility.

4.12 Contracts.

As of the Execution Date, true, correct, and complete copies of all of the (a) Assumed Contracts set forth on Schedule 2.1(e) have been made available to Transferee or otherwise made available in the VDR. As of Closing, true, correct, and complete copies of all of the Assumed Contracts set forth on Schedule 2.1(e) as revised and supplemented prior to Closing will have been made available to Transferee or otherwise made available in the VDR. A list of proposed Assumed Contracts is attached hereto as Schedule 4.12(a). Except as set forth in Schedule 4.12(a), (i) each of the Assumed Contracts is valid, binding and enforceable in accordance with its terms subject to bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the rights of creditors generally and by general principles of equity (regardless of whether such enforceability is considered in the proceeding in equity or at law), and (ii) there is not any existing material default or material event of default, or any event which, with or without notice or lapse of time or both, would constitute a material default under any Assumed Contract by Transferor. In addition, with respect to each Assumed Contract that is a lease of equipment or other personal property, except as set forth in Schedule 4.12(a), (i) such lease creates a valid leasehold interest in all property purported to be leased thereunder, (ii) all rent and other required payments have been timely paid by Transferor through the date hereof, and (iii) Transferor is in lawful possession of all of such property. Solely to the extent such changes are required in order to track the addition or deletion of Assumed Contracts from Schedule 2.1(e), this Schedule 4.12(a) may be updated by Transferor in consultation with Transferee until fifteen (15) days prior to Closing to reflect such required changes, after which time this Schedule 4.12(a) shall be deemed final.

- (b) The execution and delivery of this OTA by Transferor and the consummation of the transactions contemplated hereby by Transferor do not require any consent under, constitute (with or without notice or lapse of time or both) a default under, result in any breach of, or give any Person any rights of termination, acceleration or cancellation of, any Assumed Contract; except for such consents, approvals, authorizations, defaults, breaches, terminations, accelerations, cancellations or Encumbrances, or failures to make any such filing, obtain any such consent, approval or authorization, or provide any such notice, which would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 4.13 <u>Resident Trust Funds</u>. The Resident Trust Funds transferred hereunder are the only such funds required to be held by applicable Law.
- 4.14 Environmental Laws. Transferor has not received any written notice of alleged, actual or potential responsibility for, or any inquiry or investigation regarding the presence or release of any Hazardous Substance at the Facility in violation of any Environmental Law, which Hazardous Substances were allegedly manufactured, used, generated, processed, treated, stored, disposed or otherwise, handled at, or transported or released from such Facility or regarding compliance with Environmental Laws. Transferor has not received any written notice of any other claim, demand or action by an individual or entity alleging any actual or threatened injury or damage to any Person or entity, property, natural resource or the environment arising from or relating to the presence or release of any Hazardous Substances at, on, under, in, to or from its Facility in connection with any operations or activities of Transferor thereat.
- 4.15 <u>Improvements.</u> Except as disclosed on any Title Reports, as defined in the Purchase Agreement, Transferor has not received any written notice for any assessments for public improvements against the Facility which remain unpaid including, without limitation, those for construction of sewer, water, gas and electric lines and mains, streets, roads, sidewalks and curbs.
- 4.16 <u>Insurance</u>. Transferor has provided to Transferee or otherwise made available in the VDR, a true and correct list of all general liability, professional liability, fire, casualty, fidelity, workers' compensation and other insurance policies currently held by or on behalf of Transferor relating to the Facility, and a description of any self-insurance arrangements by or affecting the Facility, including any reserves established thereunder. Each of said policies is in full force and effect and shall be maintained by Transferor in full force and effect until the Closing, and all premiums due thereunder have been paid and shall be paid by Transferor until the Closing. Transferor has maintained or caused to be maintained insurance policies that have insured the Facility and the Assets continuously since the date Transferor first operated the Facility.

4.17 Financial Statements: Undisclosed Liabilities.

- (a) Transferor has delivered to Transferee or otherwise made available in the VDR, prior to the Execution Date, the Financial Statements. Except as set forth on Schedule 4.17(a), the Financial Statements, in all material respects, are complete and accurate and present fairly the financial position of the Facility as of the dates and periods indicated, in accordance with GAAP subject to normal year-end adjustments and absence of notes and, to the extent consistent with GAAP, Transferor's past practices in preparing financial statements, subject, in the case of any quarterly Financial Statements included therein, to normal year-end audit adjustments.
- (b) Except as reflected or reserved for or disclosed in the Financial Statements, Transferor has no material liabilities relating to the Business of a type or nature to be reflected, in accordance with GAAP and, to the extent consistent with GAAP, Transferor's past practices in preparing financial statements, on the face of a balance sheet except for (i) liabilities incurred in the ordinary course

of business since December 31, 2016, consistent with past practice, (ii) obligations arising or resulting from the terms of any Assumed Contract, and (iii) Excluded Liabilities.

- 4.18 <u>Broker</u>. Except as set forth on <u>Schedule 4.18</u>, Transferor has not engaged, nor is liable to pay any fees, costs or commissions to, any broker, finder, agent or financial advisor (each, a "*Broker*" and collectively, "*Brokers*") in connection with the transactions contemplated hereby.
- 4.19 <u>Intellectual Property</u>. To Transferor's Knowledge, except as would not reasonably be expected to have a Material Adverse Effect on the Facility, (a) Transferor owns or possesses all licenses or other rights to use all Intellectual Property necessary to conduct the Business as presently conducted, (b) Transferor has not received any written notice from any third party that the Business as currently conducted misappropriates or infringes upon any Intellectual Property rights of others, and (c) Transferor has not received any written notice that any third party is infringing any Intellectual Property owned by Transferor and used exclusively in connection with the Business.
- THE ASSETS ARE BEING SOLD, NO WARRANTY OF CONDITION. 4.20 TRANSFERRED, ASSIGNED AND DELIVERED BY TRANSFEROR AND RECEIVED BY TRANSFEREE AS IS WHERE IS, AND WITH ALL FAULTS, AND WITHOUT ANY REPRESENTATIONS OR WARRANTIES WHATSOEVER, EXPRESS OR IMPLIED, WRITTEN OR ORAL, WHETHER STATUTORY, ARISING BY OPERATION OF LAW, ARISING BY CUSTOMS OR USAGES OF TRADE, OR OTHERWISE, EXCEPT SOLELY AS EXPRESSLY SET FORTH IN THIS ARTICLE IV TO THIS OTA, THE PURCHASE AGREEMENT, AND THE OTHER TRANSACTION DOCUMENTS, AND SUBJECT TO ANY AND ALL LIMITATIONS AND QUALIFICATIONS HEREIN; IT BEING THE INTENTION OF TRANSFEROR AND TRANSFEREE TO EXPRESSLY REVOKE, RELEASE, WAIVE, DISCLAIM, NEGATE AND EXCLUDE ALL EXPRESS AND IMPLIED REPRESENTATIONS AND WARRANTIES (EXCEPT SOLELY AS EXPRESSLY SET FORTH IN THIS ARTICLE IV TO THIS OTA AND SUBJECT TO ANY AND ALL LIMITATIONS AND QUALIFICATIONS HEREIN) INCLUDING, WITHOUT LIMITATION, AS TO (a) THE CONDITION OF THE ASSETS OR ANY ASPECT THEREOF INCLUDING, WITHOUT LIMITATION, ANY AND ALL EXPRESS OR IMPLIED REPRESENTATIONS AND WARRANTIES OF OR RELATED TO MERCHANTABILITY, FITNESS FOR A PARTICULAR USE OR PURPOSE OR NON-INFRINGEMENT; (b) THE NATURE OR QUALITY OF CONSTRUCTION, STRUCTURAL DESIGN, OR ENGINEERING OF THE ASSETS OR ANY OTHER ASSET OR PROPERTY, IF ANY; (c) THE QUALITY OF THE LABOR OR MATERIALS INCLUDED IN THE ASSETS; (d) ANY FEATURES OR CONDITIONS AT OR WHICH AFFECT THE ASSETS WITH RESPECT TO ANY PARTICULAR PURPOSE, USE, POTENTIAL, OR OTHERWISE; (e) THE SIZE, SHAPE, CONFIGURATION, CAPACITY, QUANTITY, QUALITY, CASH FLOW, EXPENSES, VALUE, MAKE, MODEL OR CONDITION OF THE ASSETS; (f) ALL EXPRESS OR IMPLIED REPRESENTATIONS OR WARRANTIES CREATED BY ANY AFFIRMATION OF FACT OR PROMISE OR BY ANY DESCRIPTION OF THE ASSETS; (g) ANY STRUCTURAL OR CONDITION OR HAZARD OR THE ABSENCE THEREOF HERETOFORE, NOW OR HEREAFTER AFFECTING IN ANY MANNER ANY OF THE ASSETS; AND (h) ALL OTHER EXPRESS OR IMPLIED WARRANTIES AND REPRESENTATIONS BY TRANSFEROR WHATSOEVER, EXCEPT AS EXPRESSLY SET FORTH IN THIS ARTICLE IV AND SUBJECT TO ANY AND ALL LIMITATIONS AND QUALIFICATIONS HEREIN. FURTHERMORE, TRANSFEROR MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, AS TO THE FUTURE PROFITABILITY, FUTURE CASH FLOW OR VIABILITY OF THE BUSINESS RELATED TO THE ASSETS, ALL OF WHICH TRANSFEREE MUST DETERMINE FROM ITS INVESTIGATION OF THE RECORDS OF TRANSFEROR AND THE FACILITY AND TRANSFEREE'S OWN BUSINESS ACUMEN.

4.21 <u>Disclosure Updates</u>. At any time, and from time to time on or prior to the applicable Closing Date, Transferor may supplement or amend the schedules (a "*Disclosure Update*"), provided such Disclosure Update (a) does not seek to cure a breach of this OTA existing as of the Execution Date and (b) does not have a Material Adverse Effect on the Facility; and provided further that no such Disclosure Update shall affect or limit the rights of any Purchaser Indemnified Party to seek indemnification under <u>Article IX</u> with respect to the facts and circumstances underlying such Disclosure Update.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF TRANSFEREE

In order to induce Transferor to enter into this OTA, Transferee hereby represents and warrants to Transferor and to the Seller of the Facility as of the Execution Date and as of the Closing Date (or in the case of representations and warranties that by their terms speak as of a specified date, as of such specified date), as follows:

5.1 Corporate.

- (a) <u>Organization</u>. Transferee is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization.
- (b) <u>Power and Authority: Authorization; Enforceability</u>. Transferee has all necessary corporate, partnership or similar power and authority to enter into the documents and instruments to be executed and delivered by Transferee pursuant hereto and to carry out the transactions contemplated hereby. The execution and delivery of this OTA and the performance of this OTA by Transferee has been duly and validly authorized. This OTA constitutes the legal, valid and binding obligation of Transferee, enforceable against Transferee in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar Laws and equitable principles relating to or limiting creditors' rights generally.
- (c) <u>Qualification</u>. Transferee is duly qualified or licensed to do business, and is in good standing, in all jurisdictions (domestic and foreign) in which the character or the location of the assets owned or leased by it or the nature of the business conducted by it requires such licensing or qualification.
- (d) <u>No Conflicts or Violations</u>. Neither the execution and delivery of this OTA or the other Transaction Documents, the consummation of the transactions contemplated hereby and thereby, nor the fulfillment of the terms hereof by Transferee shall (i) violate or result in a breach of any of the material terms and provisions of, constitute a default under, conflict with, or result in any acceleration of rights, benefits or obligations of any party under any contracts to which Transferee is a party or by which it is bound, (ii) violate any Order of any Governmental Authority applicable to Transferee, or (iii) constitute a violation by Transferee of any applicable Law, or (iv) conflict with or violate any organizational document of Transferee.
- 5.2 <u>Litigation</u>. There are no proceedings, orders, or determinations by or with any arbitrator, court, or other governmental body, authority or agency, or to Transferee's Knowledge, threatened against or by Transferee or any of its Affiliates that challenge (or could challenge) or seek (or could seek) to prevent, enjoin, or otherwise delay the consummation of the transactions contemplated under this OTA or the execution and delivery of any agreement in connection therewith.

5.3 <u>Broker</u>. Except as set forth in <u>Schedule 5.3</u>, Transferee has not engaged, nor is liable to pay any fees, costs or commissions to any Broker(s) in connection with the transactions contemplated hereby.

5.4 Transferee's Reliance.

- Transferee acknowledges that it has been assured by Transferor that Transferee will be permitted full and complete access to the Facility, the Records, equipment, Returns, Contracts, insurance policies (or summaries thereof), and other properties and assets of Transferor concerning the Facility, that it and its representatives have desired or requested to see or review, and that it has been assured by Transferor that Transferee and its representatives will be permitted a full opportunity to meet with the officers, management and employees of Transferor to discuss the Facility. acknowledges that prior to the Execution Date, it will have conducted such independent investigation of the Assets and the Facility and Transferor to its own full satisfaction. In connection with Transferee's investigation, Transferee may have received from Transferor certain projections, forward-looking statements and other forecasts and certain business plan information. Transferee acknowledges that there are uncertainties inherent in attempting to make such estimates, projections and other forecasts and plans, that Transferee is familiar with such uncertainties, that Transferee is taking full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections and other forecasts and plans so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections, forecasts or plans), and that (except in the case of fraud) Transferee shall not have or make any claim against any Person with respect thereto. Accordingly, Transferee acknowledges that Transferor or any other Person have not and do not make any direct or indirect representation or warranty with respect to such forward-looking estimates, projections, forecasts or plans (including the reasonableness of the assumptions underlying such estimates, projections, forecasts or plans).
- Transferee acknowledges that Transferor or any other Person have not made any (b) representation or warranty, expressed or implied, as to the accuracy or completeness of any information regarding Transferor, the Assets and the Facility furnished or made available to Transferee and its representatives, except as expressly set forth in Article IV and Article IV of the Purchase Agreement, and Transferor or any other Person (including any officer, director, manager, member or partner of any of Transferor) shall not have or been subject to any liability to Transferee (except in the case of fraud), or any other Person, resulting from Transferee's use of any information, documents or material made available to Transferee in any confidential information memoranda, "data rooms," management presentations, due diligence or in any other form in expectation of the transactions contemplated hereby. Transferee acknowledges that except as expressly set forth in Article IV and Article IV of the Purchase Agreement, the Facility and the Assets have been acquired without any representation or warranty as to merchantability or fitness for any particular purpose of their respective assets, in an "as is" condition and on a "where is" basis. For the avoidance of doubt, nothing in this Section 5.4 is intended to limit or modify the representations and warranties contained in this Article V. Transferee acknowledges that, except for the representations and warranties contained in Article IV and Article IV of the Purchase Agreement, neither Transferor nor any other Person has made any other express or implied representation or warranty by or on behalf of Transferor.

ARTICLE VI COVENANTS AND AGREEMENTS

6.1 <u>Conduct of Business</u>. From and after the date hereof and pending the Closing, unless Transferee shall otherwise consent in writing, from and after the date hereof until the earlier of the termination of this OTA or the Closing, Transferor shall (a) maintain and operate its applicable Assets and Facility only in the ordinary and usual course of business diligently and in good faith, consistent with past practice; (b) maintain, repair and replace where appropriate, consistent with past practice, with no

less than like kind, the real and personal property, equipment, furniture and fixtures, leasehold improvements in substantially the same condition that exists on the date hereof, reasonable wear and tear excepted and subject to the requirements to repair and replace as set forth in the Purchase Agreement; (c) refrain from delaying such repair and/or replacement as a result of the pending transfer, except where such delay is consistent with past practice; (d) replace inventory, supplies and equipment consistent with past practice; (e) otherwise operate the Facility so as not to cause a breach of any covenant or warranty contained in this OTA; (f) exercise its commercially reasonable efforts to preserve Transferor's existing relationships with suppliers, distributors, customers and others having business relations with Transferor such that the Facility will not be impaired; (g) maintain all existing policies of insurance (or comparable policies) of or relating to the Assets and the Facility in full force and effect; (h) use its commercially reasonable efforts to keep available the services of the present officers and employees of Transferor involved in the day to day operation of the Facility; (i) use its commercially reasonable efforts maintain the quality of care to the residents; (j) invoice and collect revenue consistent with past practice; (k) take reasonable steps to safeguard, maintain and preserve all Facility employee and medical records transferred under this OTA in accordance with the provisions of, and for the periods prescribed by all applicable Laws, which shall in no event be less than the steps taken by Transferor in the operation of the Facility prior to the transactions contemplated under this OTA; (1) avoid immediate jeopardy violations, maintain provider agreements without suspension, qualification or limitation or revocation, and avoid what is commonly known as a "ban on admission" or imposition of civil money penalties or the providing of substandard care; (m) undertake or implement all actions, payments, and plans of correction required in connection with a Pre-Closing Imposition; (n) complete any capital repairs or physical improvements required to remove or resolve a Pre-Closing Imposition; (o) notify Transferee if Transferor becomes aware of any violation or non-compliance with any Law, except where any such violation or non-compliance would not reasonably be expected to result in a Material Adverse Effect; and (p) actively market the Facility in a manner consistent with past practice.

- 6.2 <u>Forbearances</u>. Without limiting the effect of any other provision of this OTA, between the date hereof and the earlier of the termination of this OTA or the Closing Date, Transferor shall not do any of the following with respect to the Assets or Facility without the prior written consent of Transferee:
- (a) sell, lease, transfer, convey or otherwise dispose of (other than in the ordinary course of business consistent with past practice), or cause or permit any Encumbrance (other than Permitted Encumbrances) to exist on, any of the Assets which will not be released at or prior to the Closing;
- (b) cancel any Assumed Contract or materially default in the performance of any Assumed Contract, or obligation, or waive any material default or potential material default by any other Party, or waive, release, compromise, settle or assign any rights or claims under any Assumed Contract;
- (c) notwithstanding anything else contained in this <u>Article VI</u>, enter into any contract or other transaction that would be material to the Facility, other than in the ordinary and usual course of business consistent with past practice and shall not, whether consistent with past practice or otherwise, enter into or amend any material Contract which is not at will on the part of Transferor or terminable by Transferor on thirty days or less written notice;
- (d) violate in any material respect, terminate or permit the lapse of, or failure to preserve any material licenses, permits and other authorizations including, but not limited to, the certificate of need, if any, or any provider agreements including, without limitation, its provider agreements with Medicare and Medicaid which are necessary or desirable for the operation of the Facility as it exists on the date hereof;

- (e) release, compromise or settle any material claim, action or legal proceeding that would result in a Material Adverse Effect or may be construed as an obligation of Transferee, other than in the ordinary and usual course of business consistent with past practice;
- (f) enter into any transaction with any owner, officer, director, manager or Affiliate of Transferor or any of their Affiliates, or any relative or Affiliate of any such owner, officer, director, manager or Affiliate other than consistent with past practice or in contemplation of the transactions to be carried out pursuant to this OTA or the Purchase Agreement;
- (g) materially change employment terms for any executive or group of employees or institute, amend or terminate any employee plan other than consistent with past practice or in contemplation of the transactions to be carried out pursuant to this OTA or the Purchase Agreement;
- (h) enter into any agreement, or adopt any resolution, to do any of the things described in subsections (a) through (f) above or otherwise commit any act which would cause Transferor to breach any covenant, representation or warranty contained in this OTA;
- (i) remove, discharge or transfer residents from the Facility to a nursing facility owned, operated or managed by Transferor or any of its Affiliates, nor shall there be any voluntary transfers by Transferor of residents from the Facility to any other nursing facility, where such transfer is not in the ordinary course of business and not (i) for reasons relating to the health and well-being of the resident transferred, (ii) for the election to transfer by the resident or his or her family or attorney-in-fact, or (iii) otherwise required by Law; or
- (j) (i) remove or relocate to any Affiliate, any administrator, director of nursing or other key employee, or (ii) hire new employees except in the ordinary course of business consistent with past practice.
- ending two (2) years after the Closing (the "Restricted Period"), neither Transferor nor its Affiliates (each a "Restricted Party") shall (a) directly or indirectly, own, operate or manage any skilled nursing facility within a ten (10) mile radius of the Facility, except for those facilities listed on Schedule 6.3 (the "Non-Competition Covenant"); or (b) hire or knowingly and intentionally solicit the Director of Nursing or Facility Administrator of the Facility, or Purchaser (the "Non-Solicitation Covenant"); provided, however, Transferor or a Restricted Party may hire any Person after the earlier to occur of (x) twelve (12) months after such Person's termination, or (y) the expiration of the Restricted Period. Further, notwithstanding the foregoing, general employment solicitations made pursuant to newspaper, television, radio or other general advertisement which are not specifically targeted at any particular person or group of persons shall not be deemed a violation of this Section. In connection with a violation of the Non-Competition Covenant, Transferor further acknowledges that the scope and duration of the provisions of this Section 6.3 are reasonable. The terms and provisions of this Section 6.3 shall survive the Closing.

6.4 Access.

(a) As of the date of execution of this OTA, Transferor shall provide Transferee and its employees, accountants, consultants, legal counsel, agents and other authorized representatives, during regular business hours and upon reasonable notice, reasonable access to the Facility, and all other properties, contracts, commitments, and Records of Transferor that relate to the Facility as Transferee may reasonably request for the purpose of transferring the Assets and Facility, and facilitating the smooth transition of operations, and Transferor shall promptly furnish Transferee such information as Transferee may from time to time reasonably require with respect to the Assets and/or the Facility. Transferor shall

cause the officers and employees of such Transferor to take commercially reasonable steps to assist Transferee (at Transferee's expense) in preparing to transfer the Assets and shall cause the counsel, accountants, consultants, and other non-employee representatives of Transferor to be reasonably available to Transferee for such purposes. Transferor shall, upon written request by Transferee and reasonable notice, (i) permit Transferee to conduct on-site visits of Transferor's properties and the Assets that comprise the Facility; and (ii) assist Transferee in contacting and arranging meetings with such suppliers of Transferor as consented to by Transferor (such consent not to be unreasonably withheld).

(b) To the extent permitted under Transferor's credit facility and any other similar financing document, Transferor will reasonably cooperate with Transferee and its financing sources in an effort to enter into subordination and intercreditor agreements, both of which shall be in a form mutually agreeable to all such parties.

6.5 Announcement and Disclosure.

- (a) No Party shall issue an initial public announcement, report, statement or press release (collectively, a "Public Announcement") regarding this OTA or the transactions contemplated hereby without the prior written consent of the other Party, except as otherwise required by Law. Notwithstanding the foregoing, (i) an announcement by Transferor to its employees or any union representing same shall not be a breach of the foregoing covenant, and (ii) Transferor shall not be prohibited from issuing a Public Announcement, at any time, regarding the transaction contemplated by the Purchase Agreement, including any such announcement that references Transferee, the Facility and/or the purchase price associated with the same.
- Except as may be necessary to enforce this OTA, or to comply with applicable Laws including securities Laws, for three (3) years after the last Closing, Transferor shall (i) treat and hold as confidential any proprietary and confidential information of Transferor exclusively related to the Assets or the Assumed Liabilities related to the Facility (collectively, "Confidential Information"), and (ii) refrain from using any of the Confidential Information except in connection with this OTA. The term "Confidential Information" shall not include information that is or becomes generally available to the public by actions of Persons other than Transferor or that pertains to any of the Excluded Assets or the Excluded Liabilities. If Transferor is required to disclose any Confidential Information in order to comply with, or avoid violating, any applicable Law, Transferor will use commercially reasonable efforts to provide Transferee with prompt notice thereof to the extent legally permissible. With the exception of securities filings reasonably required of a public company like Transferor's indirect parent, to the extent legally permissible and at Transferee's sole expense, Transferor shall provide Transferee, in advance of any such disclosure, with copies of any Confidential Information that Transferor intends to disclose (and, if applicable, the text of the disclosure language itself) and shall reasonably cooperate with Transferee, at Transferee's sole expense, if permitted by applicable Law, to the extent Transferee may reasonably seek to limit such disclosure in a manner consistent with applicable Law.
- (c) Except as may be necessary to enforce this OTA or any other Transaction Document, for three (3) years after the last Closing, Transferee shall (i) treat and hold as confidential any proprietary and confidential information of Transferor or any of its Affiliates that does not exclusively relate to the Assets or the Assumed Liabilities related to the Facility, including any proprietary and confidential information relating to any of the Excluded Assets or the Excluded Liabilities (collectively, "Transferor Confidential Information"), and (ii) refrain from using any of Transferor Confidential Information except in connection with this OTA. The term "Transferor Confidential Information" shall not include information that is or becomes generally available to the public by actions of Persons other than Transferee or any of its Affiliates. If Transferee is required to disclose any Transferor Confidential Information in order to avoid violating any applicable Law, Transferee will use commercially reasonable efforts to provide Transferor with prompt notice thereof to the extent legally permissible. To the extent

legally permissible and at Transferor's sole expense, Transferee shall provide Transferor, in advance of any such disclosure, with copies of any Transferor Confidential Information that Transferee intends to disclose (and, if applicable, the text of the disclosure language itself) and shall reasonably cooperate with Transferor, at Transferor's sole expense, if permitted by applicable Law, to the extent Transferor may reasonably seek to limit such disclosure in a manner consistent with applicable Law.

- Appropriate Action; Consents; Filings. From and after the Execution Date, each of the Parties shall use its commercially reasonable efforts to obtain from any Governmental Entities or third parties any consents, licenses, permits, waivers, approvals, authorizations or orders required to be obtained, or made, by such Party in connection with the authorization, execution and delivery of this OTA and the consummation of the transactions contemplated hereby and shall provide such notices, and Transferee shall post such escrows, as required by the applicable Governmental Entities and Laws, and each Party shall comply with any written agreements with third parties to consummate the transaction. The Parties shall cooperate with each other in connection with the making of all such filings, including the timing of such filings and providing copies of all such documents to the non-filing Parties and their advisors prior to filing and, if requested, to accept all reasonable additions, deletions or changes to such filings suggested in connection therewith.
- Access to Records. From and after the Closing Date, Transferee shall allow Transferor and its Affiliates, agents and representatives to have reasonable access to (upon reasonable notice and during normal business hours), and to make copies of the Records (at Transferor's expense), to the extent reasonably necessary to enable Transferor to, among other things, investigate and defend malpractice, employee or other claims, to support medical review requests from Medicare or Medicaid, to support Medicare and Medicaid claims appeals, to file or defend Cost Reports and Tax returns, to complete/revise, as needed, any patient assessments which may be required for Transferor to seek reimbursement for services rendered prior to the Closing Date and to enable Transfer to complete, in accordance with Transferor's policies and procedures, any and all post-Closing Date accounting, reconciliation and closing procedures including, but not limited to, a month end close out of all accounts including, but not limited to, accounts payable and Medicare and Medicaid billing. Transferor agrees not to use or disclose any of the information obtained from Transferee except solely for the purposes described herein, and further agrees to maintain this information as confidential. Likewise, from and after the Closing Date, Transferor shall allow Transferee and its agents reasonable access to the Records including, without limitation, the Prior Records, to the extent Transferee reasonably requires such access in connection with, without limitation, accounting, billing, Tax filings or securities filings, Medicare and/or Medicaid filings and appeals. Transferor shall use its commercially reasonable efforts to provide such items which require expedited handling to Transferee within ten (10) Business Days of Transferee's request. Transferee agrees not to use or disclose any of the information obtained from Transferor except solely for the purposes described herein, and further agrees to maintain this information as confidential. Transferee shall assure that any successor operator of the Facility is legally obligated to provide Transferor access to the Records in the manner required by this <u>Section 6.7</u>.
- 6.8 Further Assurances. From time to time after the Closing, Transferor shall, at the reasonable request of Transferee and at Transferee's expense but without further consideration, execute and deliver any further deeds, bills of sale, endorsements, assignments, and other instruments of conveyance and transfer, and take such other actions as Transferee may reasonably request and consistent with this OTA in order to (a) more effectively transfer, convey, assign and deliver to Transferee, and to place Transferee in actual possession and operating control of, and to vest, perfect or confirm, of record or otherwise, in Transferee all right, title and interest in, to and under the Assets or the Facility, (b) assist in the collection or reduction to possession of any and all of the Assets or the Facility or to enable Transferee to exercise and enjoy all rights and benefits with respect thereto, (c) with respect to any payor agreement that is non-transferrable, reasonably cooperate with Transferee to assist Transferee in securing a new agreement, or (d) otherwise carry out the intents and purposes of this OTA. In the case of rights

26

(including, without limitation, under any Contract) which cannot be transferred effectively without the consent of third parties, Transferor shall use its commercially reasonable efforts (within commercially reasonable limits) to obtain such consent and to assure to Transferee the benefits thereof during the terms thereof.

6.9 No Negotiation. Until such time as this OTA may be terminated pursuant to Article IX, Transferor shall not directly or indirectly solicit, initiate, encourage or entertain any inquiries or proposals from, or discuss or negotiate with any Person other than Transferee or its representatives relating to an acquisition or other disposition of any material Assets of the Facility or any other asset which is required by the OTA to be transferred to Transferee at Closing. Notwithstanding the foregoing, Transferor shall not be in any way limited from initiating or participating in discussions concerning any transactions involving the Excluded Assets.

6.10 Accounts Receivable.

- (a) Accounts Receivable. Transferor shall retain whatever right, title and interest it may have in and to all outstanding Accounts Receivable with respect to the Facility which relate to periods ending on or before the Effective Time, including any Accounts Receivable arising from rate adjustments which relate to a period ending on or before the Effective Time even if such adjustments occur after the Effective Time, and including any Medicaid lag payments (collectively, "Transferor's A/Rs"). Transferor acknowledges that Transferee owns all Accounts Receivable arising from services provided by or at the Facility after the Effective Time ("Transferee's A/Rs").
- (b) <u>Receipts by Transferee</u>. In furtherance and not in limitation of the requirements set forth in <u>Section 6.3(a)</u>, payments received by Transferee after the Effective Time from third party payors including, but not limited to, Medicare, Medicaid, VA, managed care and health insurance, shall be handled as follows:
- (i) If such payments either specifically indicate on the accompanying remittance advice, or if Transferor and Transferee agree that such payments relate to the period ending before the Effective Time, they shall be forwarded by Transferor to Transferor, along with the applicable remittance advice, within twenty (20) days after receipt thereof; and
- (ii) If such payments indicate on the accompanying remittance advice, or if Transferor and Transferee agree that such payments relate to the period after the Effective Time, they shall be retained by Transferee.
- (c) <u>Receipts by Transferor</u>. Payments received by Transferor after the Effective Time from third party payors including, but not limited to, Medicare, Medicaid, VA, managed care and health insurance, shall be handled as follows:
- (i) If such payments either specifically indicate on the accompanying remittance advice, or if Transferor and Transferee agree that such payments relate to the period after the Effective Time, they shall be forwarded by Transferor to Transferee, along with the applicable remittance advice, within twenty (20) days after receipt thereof; and
- (ii) If such payments indicate on the accompanying remittance advice, or if Transferor and Transferee agree that they relate to the period ending on or before the Effective Time, they shall be retained by Transferor.
- (d) Other Receipts. If the remittance advice indicates or the Parties agree that any payment relates to periods both prior to or on and after the Effective Time, the Party receiving the

payment shall forward the amount relating to the other Party's operation of the Business, along with the applicable remittance advice, within twenty (20) days after receipt thereof. If the remittance advice does not indicate the period to which a payment relates or whether it is for Transferor or Transferee, or if there is no accompanying remittance advice, or the payment is not otherwise identifiable using commercially reasonable efforts, and if the Parties do not otherwise agree as to how to apply such payment, then 100% of such payments received within the first sixty (60) days after the Effective Time shall be deemed to have been collected in respect of Transferor's A/Rs due from the payee in respect of services provided on or prior to the Effective Time. All such payments received in excess of the amount of Transferor's A/Rs due from said payee and all such payments received sixty (60) days after the Effective Time shall be deemed to have been collected in respect of Transferee's A/Rs from said payee. All such payments received by Transferee but which are deemed to be due Transferor under this Section 6.10 shall be forwarded by Transferee to Transferor within twenty (20) days after receipt thereof, and all such payments received by Transferor but which are deemed to be due Transferee under this Section 6.10 shall be forwarded by Transferor to Transferee within twenty (20) days after receipt thereof. All such payments received by Transferor which are deemed to have been collected in respect to Transferor's A/Rs shall be retained by Transferor and all such payments received by Transferee which are deemed to have been collected in respect to Transferee's A/Rs shall be retained by Transferee. Transferee shall pay to Transferor any and all reimbursements including retroactive rate adjustments, appeal settlements and/or Cost Report settlements for all Cost Report periods with fiscal years ended prior to the Effective Time that it receives after the Effective Time. During the five (5) year period following the Effective Time, Transferee shall also make a good faith effort to reconcile its Cost Report reimbursements and/or settlements with documentation Transferor provides to Transferee regarding Transferor Bad Debt and shall pay to Transferor any and all reimbursements and/or settlements related to Transferor Bad Debt pursuant to Section 6.11.

(e) <u>Medicaid Applications</u>. In connection with Transferor's attempts to collect Medicaid funds for services rendered to those Residents with pending Medicaid applications (collectively, the "*Pending Medicaid Applicants*"), each set out on <u>Schedule 6.10(e)</u>, (i) Transferee shall provide Transferor with a written monthly progress report on the Medicaid application status of each Pending Medicaid Applicant until such time as all Pending Medicaid Applicants have been approved or denied by Medicaid, and (ii) if Transferee receives any notice or correspondence regarding such applications, Transferee shall provide such notice or correspondence to Transferor within five (5) Business Days following receipt. Transferee shall cooperate with and provide Transferor with such documents and information as Transferor shall reasonably request to enable Transferor to contest any denial or negative determinations by Medicaid with respect to the Pending Medicaid Applicants.

(f) Accounting for Accounts Receivable.

- (i) Attached hereto as <u>Schedule 6.10(f)(i)</u> is a schedule of Transferor's A/Rs listing by Resident the amount due as of five (5) days prior to Closing. As soon as reasonably possible but not later than fifteen (15) Business Days after the Closing Date, Transferor shall provide Transferee with a schedule of Transferor's A/Rs listing by Resident the amounts due as of the Effective Time.
- (ii) For a period of twelve (12) months following the applicable Effective Time or until Transferor receives payment of all Accounts Receivable attributable to the operation of the Facility on or before the Effective Time, whichever is sooner, Transferee shall provide Transferor (no less frequently than monthly) with (A) an accounting setting forth all amounts received by Transferee during the preceding month with respect to Transferee's A/Rs and Transferor's A/Rs using the same format of schedule as that provided by Transferor pursuant to Section 6.10(b), and (B) copies of all remittance advices relating to such amounts received and any other reasonable supporting documentation as may be required for Transferor to determine that Transferee's A/Rs and Transferor's A/Rs have been paid. Transferee shall deliver such accounting to Transferor at the following physical and email addresses: 680

South Fourth Street, Louisville, Kentucky 40202, Attention: Raye Ann Cole (raye.cole@kindred.com) and Linda Fisher (linda.fisher@kindred.com).

(iii) For a period of twelve (12) months following the Effective Time or until
Transferee receives payment of all Accounts Receivable attributed to the operation of the Facility prior to
the Closing Date, whichever is sooner, Transferor shall provide Transferee (no less frequently than
monthly) with (A) an accounting setting forth all amounts received by Transferor with respect to
Transferee's A/Rs and Transferor's A/Rs using the same type of schedule as that provided by Transferor
pursuant to Section 6.10(b), and (B) copies of all remittance advices relating to such amounts received
and any other reasonable supporting documentation as may be required for Transferee to determine
Transferee's A/Rs and Transferor's A/Rs that have been paid. Transferor shall deliver such accounting to
Transferee at the following address:, Attention:

- (iv) On two (2) occasions during the period of one (1) year following the Effective Time, Transferor and Transferee shall, upon reasonable notice and during normal business hours, have the right to inspect all cash receipts of the other Party in order to confirm the other Party's compliance with the obligations imposed on it under <u>Sections 6.10 and 6.11</u>. Notwithstanding the foregoing, if such information can be transmitted through electronic mail, then Transferor and Transferee may satisfy their obligations under this <u>Section 6.10</u> in that manner.
- (v) To enable Transferor to close its books with respect to the period ending on the Closing Date, Transferee will permit appropriate personnel of Transferor reasonable access to the Facility, in a manner that does not materially interfere with the operation of the Facility, for a period of no more than forty-five (45) Business Days after the Closing Date. During that period, Transferee will permit certain individuals employed by Kindred or its Affiliate immediately before the Closing Date to provide reasonably necessary assistance to Kindred in its closing of the books. Those individuals are the persons previously employed by Kindred or Transferor in following positions: Executive Director; Business Office Manager; Accounts Receivable Assistant; Accounts Payable Coordinator; Payroll Benefits Coordinator and MDS Coordinator.
- (vi) Any amounts to be paid by one Party to the other Party under this <u>Article VI</u> shall be made by electronic transfer using the wire instructions set forth on <u>Exhibit 6.10(f)</u>.
- (g) <u>Transferor Collection Activities</u>. After the Closing Date, Kindred and Transferor shall have the right, and any agent or representative retained by the foregoing shall have the right on behalf of Kindred and Transferor, to engage in any commercially reasonable collection activities with respect to any unpaid Transferor's A/R's, including private pay amounts.
- (h) <u>Delivery of Mail</u>. To the extent that Transferee or any of its Affiliates receives any mail or packages addressed to Kindred, Transferor or any of their Affiliates not relating to the Assets or the Assumed Liabilities, Transferee shall promptly deliver such mail or packages to Transferor. After the Closing Date, Transferee may deliver to Transferor any checks or drafts made payable to Transferor or its Affiliates that constitutes an Asset, and Transferor shall promptly deposit or cause to be deposited such checks or drafts and, upon receipt of funds, reimburse Transferee within ten (10) Business Days for the amounts of all such checks or drafts, or, if so requested by Transferee, endorse such checks or drafts to Transferee for collection. To the extent Transferor or its Affiliates receives any mail or packages addressed to Transferor or its Affiliates but relating to the Assets or the Assumed Liabilities relating to the Facility, Transferor shall promptly deliver such mail or packages to Transferee. After the Closing Date, to the extent that Transferee receives any cash or checks or drafts made payable to Transferee that constitutes an Excluded Asset, Transferee shall promptly use such cash to, or deposit such checks or drafts and upon receipt of funds from such checks or drafts, reimburse Transferor within ten (10) Business Days for such amount received, or, if so requested by Transferor, endorse such checks or drafts

to Transferor for collection. The Parties may not assert any set off, hold back, escrow or other restriction against any payment described in this <u>Section 6.10(h)</u>.

6.11 <u>Cost Reports</u>.

- (a) Transferor shall prepare and file with its fiscal intermediary the final Medicare Cost Reports covering its operation of the Business through the Effective Time as soon as reasonably practicable after the Effective Time, but in no event later than the date on which such final Cost Report is required to be filed by applicable Law under the terms of the Medicare program, and will provide the fiscal intermediary or CMS with any information needed to support claims for reimbursement made by Transferor either in said final Cost Report or in any Cost Reports filed for prior Cost Reporting periods. Simultaneously with such filing, Transferor shall provide Transferee with a copy of the final Medicare Cost Reports and such supporting documentation reasonably requested by Transferee in writing.
- (b) After the Closing Date, Transferor shall promptly and diligently provide Transferee with reasonable and appropriate documentation regarding the Medicare bad debts incurred by Transferor prior to the Effective Time associated with the Facility ("*Transferor Bad Debt*") for purposes of facilitating Transferee's preparation of related Cost Reports. Transferor agrees to reasonably cooperate by providing reasonably requested pre-Effective Time data to Transferee in connection with Transferee's preparation of Cost Reports with respect to the period after the Effective Time.
- (c) Transferee shall timely prepare and file with CMS and the appropriate state agency for the Facility, its initial Cost Report for the fiscal year commencing with the fiscal year in which the Closing Date occurs, and will include Transferor Bad Debt in its initial Cost Report.
- (d) Transferee shall notify Transferor within ten (10) Business Days of receipt of any notice of adverse audit adjustments, overpayment, recoupment, fine, penalty, late charge or assessment accruing in relation to Transferor Bad Debt. Transferee agrees to appeal at the request of, on behalf of, and at the sole expense of Transferor, any Medicare claims audit, Cost Report audit, overpayment, recoupment, fines, penalties, late charges and assessment accruing in relation to Transferor Bad Debt. Transferor and Transferee shall each reasonably cooperate with the other respective Party, with respect to any such matters including, but not limited to, timely providing any requested documentation within the other Party's possession or control relating to such matters. Transferee is not responsible for (i) the actual results of any such appeal, or (ii) Transferor's failure to provide information and/or documents necessary to process any such appeal.
- (e) Transferor and Transferee shall comply with all patient identity and information protection Laws in providing information under this <u>Section 6.11</u>.
- (f) In the event that, following the applicable Effective Time, Transferee or any of its Affiliates suffers any offsets against reimbursement under any third party payor or reimbursement programs owed to such Party relating to amounts owing under any such program by Transferor or any of its Affiliates for services rendered prior to the Effective Time, Transferor shall immediately upon written demand from Transferee pay to such Party the amounts so billed or offset, even if Transferor appeals the adverse claim. To the extent that Transferor is successful in any appeal of any adverse audit adjustments, overpayment, recoupment, fine, penalty, late charge or assessment by any third party payor accruing for any period prior to the Effective Time, and Transferee or its Affiliates receive any monies from a third party payor or reimbursement program as a result of Transferor's successful appeal, then Transferee and/or its Affiliates agree that it will promptly refund to Transferor any amounts previously paid by Transferor to such Party for any reimbursement offsets in accordance with the preceding sentence.

- 6.12 <u>Assistance in Proceedings</u>. Transferee shall cooperate with any Transferor and its counsel in the contest or defense of, and make available its personnel and provide any testimony and access to its Records in connection with, any proceeding involving or relating to (a) any contemplated transaction herein, or (b) any action, activity, circumstance, condition, conduct, event, fact, failure to act, incident, occurrence, plan, practice, situation, status or transaction on or before the Closing Date involving Transferor, the Facility or its Business.
- 6.13 Overhead and Shared Services; National Contracts. Transferee acknowledges that the Facility currently benefits from the National Contracts and receives Overhead and Shared Services from Transferor and its Affiliates. Transferee further acknowledges that, as it relates to the operation of the Facility, all such benefits from the National Contracts and provision of Overhead and Shared Services shall cease, and any agreement by the Facility with Transferor or any of its Affiliates in respect to benefiting from the National Contracts or the provision of Overhead and Shared Services shall terminate, as of the Closing Date for the Facility. No Overhead and Shared Services shall be provided by Transferor or any of its Affiliates to the Facility after the Effective Time.
- 6.14 <u>Business Relationships</u>. After the Closing Date, each Party will reasonably cooperate with the other Parties in its efforts to continue and maintain for the benefit of those business relationships of Transferor existing prior to the Closing Date and relating to the business to be operated by Transferee after the Closing, including relationships with lessors, employees, regulatory authorities, licensors, patients, suppliers and others. The foregoing notwithstanding, Transferor does not make any representation or warranty as to the prospects or outlook for such business relationships as carried on by Transferee after the Closing Date.

6.15 Information Systems, Records in Electronic Form, Software and Data.

- (a) Transferor shall use reasonable efforts to permit transfer of current data in fully operational form for use in Transferee's computer applications. Transferor further agrees that in order to assist Transferee in ensuring the continued operation of the Facility after the Closing Date in compliance with applicable Law and in a manner which does not jeopardize the health and welfare of the Residents of the Facility, Transferor shall, for a period of no longer than sixty (60) days after the Closing Date, provide Transferee access to Transferor's electronic medical records system to enable Transferee, at its expense, to print the medical treatment records and physician orders for each Resident as of the Closing Date that was a Resident as of the period between and including the Effective Time and the date that is eighteen (18) months prior to the Effective Time, and cooperate with Transferee regarding the delivery of all such Records to Transferee, in electronic form (including the provision to Transferee of the last eighteen (18) months of MDS (Minimum Data Set) history in the format submitted to CMS by Transferor), in order to enable Transferee to obtain the necessary copies of such medical records and physician orders.
- (b) At least thirty (30) days prior to the Closing Date, Transferor shall provide Transferee and its representatives with access to the Facility so that Transferee can install lines necessary for computer hardware, together with servers, computer hardware and software. In addition, Transferor shall cooperate with Transferee and provide Transferee with such assistance as Transferee may reasonably request in order to provide for an orderly, efficient and safe transition of the operations from Transferor to Transferee and the continued operation of the Facility after the Closing Date in compliance with applicable Law and in a manner which does not jeopardize the health and welfare of the Residents of the Facility. During such time as Transferor provides Transferee with access to the Facility under this Section 6.15(b), Transferee shall not connect with or hook into Transferor's computer lines or computers. Such access shall be prescheduled with Transferor and shall not interrupt normal business operation. Transferee shall indemnify Transferor for any damage resulting from such access or installation of the lines. If, for any reason, this OTA terminates prior to the Closing Date, Transferor shall remove any lines

installed, at Transferee's cost and expense, and Transferee shall remit payment to Transferor within five (5) days of receiving an invoice for such costs and expenses of removal.

ARTICLE VII CONDITIONS PRECEDENT TO THE OBLIGATIONS OF PARTIES

- 7.1 <u>Conditions to Obligations of Transferee</u>. The obligations of Transferee hereunder are subject to the fulfillment of all of the following conditions precedent unless such fulfillment is waived in writing by Transferee, subject to the limitations contained herein, as the case may be:
- (a) Representations and Warranties. The representations and warranties of Transferor set forth in Article IV shall be true and correct in all material respects (or, with respect to any representation qualified as to materiality, true and correct) on and as of the Closing Date as if made on and as of the Closing Date, except to the extent any such representation or warranty expressly is made as of an earlier date or with respect to a particular period, in which case such representation or warranty shall have been true and correct in all material respects (or, with respect to any representation qualified as to materiality, true and correct) as of such date or with respect to such period.
- (b) <u>No Litigation</u>. Without limiting the generality of any representation, no injunction, temporary restraining order, judgment or other order of any court or governmental agency or instrumentality shall have issued or have been entered which would be violated by the consummation of the transactions contemplated hereby; and no suit, action or other proceeding brought by the United States, the State of Vermont or any political subdivision, which any Facility is located or any agency or instrumentality thereof shall be pending in which it is sought to restrain or prohibit this OTA or the consummation of the transactions contemplated hereby.
- (c) <u>Compliance with Covenants</u>. Transferor shall have performed and complied, in all material respects, with all terms, agreements, covenants and conditions of this OTA to be performed or complied with by it at or prior to the Closing.
- (d) <u>Authorization</u>. Transferor shall have approved and authorized the transactions contemplated by this OTA.
- (e) <u>No Portfolio Material Adverse Effect</u>. Since the date of execution of this OTA, there shall have been no Portfolio Material Adverse Effect.
- (f) <u>Purchase Agreement</u>. All the conditions for closing of the Facility under the Purchase Agreement have been satisfied or waived other than the closing of the transactions contemplated under this OTA.
- (g) <u>Termination of Management Agreements</u>. Any management agreements between Transferor and Kindred and/or its Affiliates shall have been terminated and any existing Leases related to the Facility between Transferor and Kindred and/or its Affiliates shall have been terminated.
- (h) <u>New License</u>. Transferee shall have received the New License as of the Closing Date (or shall have obtained reasonable assurances from the Department that the New License has been or will be issued by the Department effective as of the Effective Time or promptly thereafter).

(i) Reserved.

- (j) <u>Closing Certificate</u>. Transferor shall have delivered to Transferee a certificate of a duly authorized officer of Transferor dated as of the Closing Date stating that the conditions specified in Sections 7.1(a) and 7.1(c) have been satisfied.
- (k) <u>Good Standing Certificate</u>. Transferor shall have delivered to Transferee a certificate of the Secretary of State of Delaware as of a recent date as to the legal existence and good standing of Transferor.
- (l) <u>Assignment and Assumption Agreement</u>. Transferor shall have executed and delivered any Assignment and Assumption Agreements related to the Facility.
- (m) <u>Resident Trust Deposits</u>. Transferor shall have delivered any and all assignment and assumptions of resident trust deposits.

(n) Reserved.

- 7.2 <u>Conditions to Obligations of Transferor</u>. The obligations of Transferor hereunder are subject to the fulfillment of all of the following conditions precedent unless such fulfillment is waived in writing by Transferor, subject to the limitations contained herein, as the case may be:
- (a) Representations and Warranties. The representations and warranties of Transferee set forth in Article V shall be true and correct in all material respects (or, with respect to any representation qualified as to materiality, true and correct) on and as of the Closing Date as if made on and as of the Closing Date, except to the extent any such representation or warranty expressly is made as of an earlier date or with respect to a particular period, in which case such representation or warranty shall have been true and correct in all material respects (or, with respect to any representation qualified as to materiality, true and correct) as of such date or with respect to such period.
- (b) <u>No Litigation</u>. Without limiting the generality of any representation, no injunction, temporary restraining order, judgment or other order of any court or governmental agency or instrumentality shall have issued or have been entered which would be violated by the consummation of the transactions contemplated hereby; and no suit, action or other proceeding brought by the United States, the State of Vermont, any political subdivision, which any Facility is located or any agency or instrumentality thereof shall be pending in which it is sought to restrain or prohibit this OTA or the consummation of the transactions contemplated hereby.
- (c) <u>Compliance with Covenants</u>. Transferee shall have performed and complied, in all material respects, with all terms, agreements, covenants and conditions of this OTA to be performed or complied with by it at or prior to the Closing.
- (d) <u>Authorization</u>. Transferee shall have approved and authorized the transactions contemplated by this OTA.
- (e) <u>No Portfolio Material Adverse Effect</u>. Since the date of execution of this OTA, there shall have occurred, no event, circumstance or other change in Transferee or its assets that, alone or in the aggregate, has had or, reasonably could be expected to have, a Portfolio Material Adverse Effect with regard to Transferee.
- (f) <u>Purchase Agreement</u>. All the conditions for closing of the Facility under the Purchase Agreement have been satisfied or waived other than the closing of the transactions contemplated under this OTA.

(g) <u>New License</u>. Transferee shall have received the New License as of the Closing Date (or shall have obtained reasonable assurances from the Department that the New License has been or will be issued by the Department effective as of the Effective Time or promptly thereafter).

(h) Reserved.

- (i) <u>Closing Certificate</u>. Transferee shall have delivered to Transferor a certificate of a duly authorized officer of Transferee dated as of the Closing Date stating that the conditions specified in Sections 7.2(a) and 7.2(c) have been satisfied.
- (j) <u>Good Standing Certificate</u>. Transferee shall have delivered to Transferor a certificate of the Secretary of State of Vermont as of a recent date as to the legal existence and good standing of Transferee.
- (k) <u>Assignment and Assumption Agreement</u>. Transferee shall have executed and delivered any Assignment and Assumption Agreements related to the Facility.

Reserved.

(m) Other Operations Transfer Agreements. The Affiliates of Transferee shall have consummated the transactions contemplated by the Other Operations Transfer Agreements, except to the extent any facility governed by any of the Other Operations Transfer Agreements is subject to a "Delayed Closing" (as governed by Sections 3.1, 6.9, 10.3 and 10.4 of the Purchase Agreement and defined below).

ARTICLE VIII DELAYED CLOSING

8.1 <u>Delayed Closing</u>. Notwithstanding anything in this OTA to the contrary, in the event the sale of the Facility under the Purchase Agreement does not occur or is delayed as a result of the provisions of Sections 3.1, 6.9, 10.3 or 10.4 of the Purchase Agreement (a "*Delayed Closing*"), the transfer of operations from Transferor associated with the Facility to Transferee shall occur, if at all, on the date of the eventual sale of such Facility pursuant to the Purchase Agreement. All of the provisions of this OTA shall apply to such transfer of operations, other than the Closing Date for such transfer of operations being adjusted accordingly in accordance with the Purchase Agreement.

ARTICLE IX TERMINATION

- 9.1 <u>Termination</u>. This OTA may be terminated and the transactions contemplated hereby abandoned at any time prior to the Closing:
- (a) by either Transferor or Transferee if (i) the Closing has not occurred by the Outside Date (unless (A) such date has been extended under the Purchase Agreement in accordance with its terms, or (B) the Facility is subject to a Delayed Closing); or (ii) the Purchase Agreement has been terminated in accordance with its terms either in full or with respect to the Facility;
 - (b) by the mutual written consent of Transferor Representative and Transferee;
- (c) by Transferee, by reason of the breach, inaccuracy or non-fulfillment of any representation, covenant, obligation or agreement by Transferor under this OTA that (i) has a Material Adverse Effect, and (ii)(A) is incapable of being cured prior to the Outside Date, or (B) has not been cured by Transferor within one hundred eighty (180) days after written notice thereof from Transferee; or

(d) by Transferor, by reason of the breach, inaccuracy or non-fulfillment of any representation, covenant, obligation or agreement by Transferee under this OTA that (i) is incapable of being cured prior to the Outside Date, or (ii) has not been cured by Transferee within forty-five (45) days after written notice thereof from Transferor Representative, except in the case of a breach of the covenants set forth in Section 2.2(a), in which case Transferee shall have five (5) days after written notice thereof from Transferor Representative to file for the applicable New Licenses and/or to notify Transferor that the necessary filings have been made in accordance with Section 2.2(a).

9.2 Procedure and Effect of Termination.

- (a) In the event of termination of this OTA pursuant to this Article IX, the terminating Party shall give written notice thereof to the other Parties and this OTA shall terminate, and the transactions contemplated hereby shall be abandoned, without further action by any of the Parties.
- (b) If this OTA is terminated as provided herein, no Party shall have any liability or further obligation hereunder to any other Party to this OTA, except (i) as provided in <u>Section 6.5</u> or <u>Article X</u>, and/or (ii) as otherwise provided for in the Purchase Agreement, and (iii) nothing herein will relieve any Party from liability for any breach of this OTA.

ARTICLE X INDEMNIFICATION

Survival of Representations, Warranties and Covenants. All representations, warranties, pre-closing covenants and obligations of Transferor, including with respect to any Facility, on the one hand, and Transferee, on the other hand, contained in this OTA or in any document to be executed and delivered pursuant to this OTA at the Closing shall survive the Closing for such Facility for eighteen (18) months and automatically terminate thereafter without any action on the part of any Party hereto; provided, however, that (a) the representations and warranties set forth in Sections 4.1 (Corporate), 4.7 (Encumbrances), 4.18 (Broker), 5.1 (Corporate), and 5.3 (Broker) shall survive indefinitely after the Closing for such Facility, (b) the representations and warranties set forth in Sections 4.4 (Taxes), shall survive until thirty (30) days after the expiration of the statute of limitations period (including all extensions thereof) applicable to the underlying subject matter being represented, and (c) the representations and warranties set forth in Section 4.5 (Employee Benefit Plans) and 4.8 (Healthcare) shall survive until the three-year anniversary of the Closing Date for such Facility. The representations and warranties contained in Sections 4.1, 4.4, 4.7, 4.18, 5.1, and 5.3 are sometimes collectively referred to herein as the "Fundamental Representations." Except as otherwise set out in this OTA, post-Closing covenants and obligations of the Parties shall survive the Closing Date for such Facility for three (3) years and automatically terminate without any action on the part of any Party hereto; provided, however, that (a) non-monetary obligations for access and/or retention of records, confidentiality, general cooperation, delivery of property received belonging to the other Party, and further assurances, shall survive for the Closing Date for the period of the statute of limitations or the specific period set forth herein, (b) Transferee's obligations with respect to Assumed Liabilities will survive the Closing Date for the period of the underlying obligation plus the relevant statute of limitations (including all extensions thereof) applicable for such Assumed Liability, and (c) Transferor's obligations with respect to Retained Liabilities will survive the Closing Date for the period of the underlying obligation plus the relevant statute of limitations (including all extensions thereof) applicable for such Retained Liability. Notwithstanding the foregoing, any covenant, obligation, representation or warranty in respect of which indemnity may be sought hereunder shall survive the time at which it would otherwise terminate pursuant to this Section 10.1 (such time, the "Expiration Date") if a Notice of Indemnification shall have been given to the applicable Indemnifying Party on or before the applicable Expiration Date; provided, however, that such survival shall automatically expire if Indemnified Party does not bring a judicial action against Indemnifying Party within one hundred eighty (180) days following the Expiration Date, and further, in the absence of the filing of such an action, the Escrow shall be released one hundred eighty (180) days after the three-year anniversary of the Initial Closing Date.

- 10.2 <u>Indemnification by Transferor</u>. Subject to <u>Section 10.1</u> and any cure periods set forth in this OTA, Transferor and Kindred or their respective successors and assigns, as applicable, shall jointly and severally indemnify and hold harmless Transferee, Purchaser and their respective Affiliates (collectively, "*Transferee Indemnified Parties*") from and against any Loss incurred or suffered by such Transferee Indemnified Party arising out of or resulting from:
- (a) a breach of any representation or warranty made by Transferor in this OTA or any other Transaction Document;
- (b) a failure by Transferor to perform or comply with the covenants on the part of Transferor set forth in this OTA or any other Transaction Document; and
- (c) any Retained Liabilities, and any obligations arising with respect to an Excluded Asset from and after the Closing Date.
- 10.3 <u>Indemnification by Transferee</u>. Subject to <u>Section 10.1</u> and any cure periods set forth in this OTA, Transferor or its successors and assigns, as applicable, shall indemnify and hold harmless Transferee and its Affiliates ("*Transferor Indemnified Parties*"), from and against any Loss incurred or suffered by such Transferor Indemnified Party arising out of or resulting from:
- (a) a breach of any representation or warranty made by Transferee in this OTA or any other Transaction Document;
- (b) a failure by Transferee to perform or comply with any covenant of Transferee in this OTA or any other Transaction Document; and
- (c) any Assumed Liability, and any obligations arising with respect to an Asset from and after the Closing Date.

10.4 Indemnification Limitations. Notwithstanding Section 10.2, if the Closing occurs:

- (a) An Indemnifying Party shall not have any obligation to indemnify an Indemnified Party with respect to a Facility whatsoever from and against any Loss pursuant to Section 10.2(a) or Section 10.3(a) unless and until the aggregate claims for such Losses with respect to such Facility (and with respect to any Transferee Indemnified Party, combined with claims for Losses by the applicable Purchaser for breaches of Seller's representations and warranties under the Purchase Agreement) exceed Fifty Thousand Dollars (\$50,000.00) (the "Indemnification Threshold"), at which time Indemnified Parties shall be entitled to recover all such Losses in excess of the Indemnification Threshold, subject to the allocated portion of the Indemnification Cap attributed to the Facility under the Purchase Agreement.
- (b) Indemnified Parties shall not be entitled to recover Losses with respect to the Facility pursuant to Section 10.2(a) or Section 10.3(a) for an aggregate amount (and with respect to Transferee Indemnified Parties, combined with the aggregate amount of Losses recovered by Purchaser for breaches of Seller's representations and warranties under the Purchase Agreement) in excess of the amount of the Indemnification Cap allocated to the Facility pursuant to the Purchase Agreement; provided, however, that claims for fraud or any breach of any of the Fundamental Representations shall not be subject to the foregoing limits and shall not be included in the determination of whether the Indemnification Cap has been reached. For all purposes of this Article X, when determining the amount

of the Losses arising out of or resulting from a breach of a representation or warranty of Transferor or Transferee, any Material Adverse Effect or other materiality qualifier contained in any such representation or warranty will be disregarded.

- Any Losses for which any Indemnified Party would be entitled to indemnification under this Article X shall be reduced by the amount of insurance proceeds actually received or recovered under any insurance policies for the benefit of such Indemnified Party (including any title policies) and any cash payments, setoffs or recoupment of any payments actually recovered by such Indemnified Party in respect of such Losses. Each Indemnified Party shall use commercially reasonable efforts to mitigate losses for which such Indemnified Party is subject to indemnification under this Article X. If, after Indemnifying Party has made an indemnification payment to an Indemnified Party with respect to Losses in satisfaction of its obligations under this Article X, Indemnified Party actually recovers from any third parties amounts in respect of such Losses, Indemnified Party shall as promptly as practicable forward to Indemnifying Party such amounts, but not in excess of the indemnification payment received by Indemnified Party. For the avoidance of doubt, Transferor shall have no obligation to indemnify (whether under this Article or otherwise) both (A) Purchaser, or an assignee of Purchaser, and (B) a Transferee with respect to any single Loss, and shall not be required to pay duplicative damages, and the Indemnification Cap allocated to the Facility under the Purchase Agreement shall be the maximum liability for indemnification claims with respect to the Facility under both this OTA and the Purchase Agreement. In no event shall the Indemnified Parties receive duplicative Losses under such agreements.
- (d) Any indemnification payments made pursuant to this OTA shall be treated as an adjustment to the allocated Purchase Price for the Facility as set forth and in accordance with the Purchase Agreement (as determined for U.S. federal income tax purposes). In the event of a claim under this Article X, a Party shall have a duty to mitigate its Losses.
- Assumption of Defense. An Indemnified Party shall promptly give notice (each, a "Notice of Indemnification") to each Indemnifying Party after obtaining knowledge of any matter as to which recovery may be sought against such Indemnifying Party because of the indemnity set forth above and, if such indemnity shall arise from the claim of a third party and Indemnifying Party provides written notice to Indemnified Party stating that Indemnifying Party is responsible for the entire claim within ten (10) days after Indemnifying Party's receipt of the applicable Notice of Indemnification, shall permit such Indemnifying Party to assume the defense of any such claim or any proceeding resulting from such claim; provided, however, that failure to give any such Notice of Indemnification promptly shall not affect the indemnification provided under this Article X, except and only to the extent such Indemnifying Party shall have been actually prejudiced as a result of such failure or if such Notice of Indemnification is not given to Indemnifying Party prior the applicable Expiration Date. If an Indemnifying Party assumes the defense of such third party claim, such Indemnifying Party shall have full and complete control over the conduct of such proceeding on behalf of Indemnified Party and shall, subject to the provisions of this Section 10.5, have the right to decide all matters of procedure, strategy, substance and settlement relating to such proceeding; provided, further, however, that any counsel chosen by such Indemnifying Party to conduct such defense shall be reasonably satisfactory to Indemnified Party; and provided, further, however, that Indemnifying Party shall not without the written consent of Indemnified Party consent to the entry of any judgment or enter into any settlement with respect to the matter which (a) does not include a provision whereby the plaintiff or the claimant in the matter releases Indemnified Party from all liability with respect thereto; and (b) in the case of Transferee as Indemnifying Party, does not include any provision that would impose any obligation (including an obligation to refrain from taking action) upon Seller. Indemnified Party may participate in such proceeding and retain separate co-counsel at its sole cost and expense (and, for the avoidance of doubt, such cost and expense shall not constitute a Loss for purposes of the Indemnification Obligations).

- 10.6 Non-Assumption of Defense. If no Indemnifying Party is permitted or elects to assume the defense of any such claim by a third party or proceeding resulting therefrom, Indemnified Party shall diligently defend against such claim or litigation in such manner as it may deem appropriate and, in such event, Indemnifying Party or Parties shall reimburse Indemnified Party for all reasonable and actually incurred out-of-pocket costs and expenses, legal or otherwise, incurred by Indemnified Party and its Affiliates in connection with the defense against such claim or proceeding, within thirty (30) days after the receipt of detailed invoices.
- 10.7 <u>Indemnified Party's Cooperation as to Proceedings</u>. Indemnified Party will cooperate in all reasonable respects with any Indemnifying Party in the conduct of any proceeding as to which such Indemnifying Party assumes the defense, except to the extent Indemnified Party could reasonably be expected to be prejudiced thereby. Indemnifying Party or Parties shall promptly reimburse Indemnified Party for all reasonable out-of-pocket costs and expenses, legal or otherwise, incurred by Indemnified Party or its Affiliates in connection therewith, within thirty (30) days after the receipt of detailed invoices therefor.

10.8 Indemnification for Resident Trust Property.

- (a) Kindred and Transferor will jointly and severally indemnify, protect, defend and hold Transferee harmless for, from and against all liabilities, claims and demands, including reasonable attorneys' fees and costs, in the event the corpus of the Resident Trust Property transferred to Transferee does not represent the correct balance of Resident Trust Property delivered to Transferor as custodian, and for claims which arise from actions or omissions of Transferor with respect to the Resident Trust Property held or handled by Transferor at any time.
- (b) Transferee will indemnify, protect, defend and hold Transferor harmless for, from and against all liabilities, claims and demands, including reasonable attorneys' fees and costs, in the event a claim is made against Transferor by a resident or his or her family for his/her Resident Trust Property where such Resident's funds or other property were properly transferred to Transferee pursuant to the terms hereof.
- 10.9 <u>Damages Disclaimed</u>. EXCEPT AS SUCH MAY BE PART OF ANY CLAIM OF ANY THIRD PARTY THAT IS NOT A TRANSFEREE INDEMNIFIED PARTY OR A PURCHASER INDEMNIFIED PARTY (AS DEFINED IN THE PURCHASE AGREEMENT), UNDER NO CIRCUMSTANCES (WHETHER UNDER THIS ARTICLE OR OTHERWISE) SHALL ANY PARTY BE RESPONSIBLE OR LIABLE IN ANY WAY HEREUNDER FOR LOSS OF PROFITS, INCIDENTAL, CONSEQUENTIAL, SPECIAL OR PUNITIVE DAMAGES, DIMINUTION IN VALUE, OR ANY EXEMPLARY DAMAGES, REGARDLESS OF WHETHER THE ACTION IS FOUNDED IN CONTRACT, TORT, STATUTORY OR OTHERWISE.
- 10.10 <u>Individual Liability Disclaimed</u>. For the avoidance of doubt, except in the event of fraud, no individual officer, director, member, managing member, shareholder, equity holder, partner, employee, agent, or representative of either Party shall have any liability for any claims of the other Party related to this OTA, or any agreements, certificates or instruments delivered in connection herewith, in any way.
- 10.11 <u>Exclusive Remedy Post-Closing</u>. With the exception of fraud and injunctive relief for specific performance or an action required under this OTA post-Closing for any Transferred Facility, the exclusive remedy of any Party after a Closing shall be indemnity under this <u>Article X</u>.

ARTICLE XI ASSIGNMENT

Assignment. Neither this OTA, nor any rights, interests or obligations hereunder, may be assigned or transferred, in whole or in part, by operation of law or otherwise by Transferor or Transferee without the prior written consent of the other Party which shall not be unreasonably withheld, conditioned or delayed, and any such assignment that is not consented to shall be null and void. Notwithstanding the foregoing, upon prior written notice to Transferor Representative, Transferee may assign all, but not less than all, of its rights, duties and obligations under this OTA to a wholly-owned subsidiary of Transferee or Purchaser or to a Substitute OTA Transferee, provided (a) it is understood that a Substitute OTA Transferee which is substantially similar to Transferee as of the Closing shall be deemed acceptable to Transferor, and (b) that no such assignment shall relieve Transferee Or Transferee Guarantor from their obligations under this OTA.

ARTICLE XII MISCELLANEOUS

- Disclosure Schedules. The information contained in the Disclosure Schedules shall be 12.1 deemed to qualify to the specific Section (or subsection, as appropriate) of this OTA to which it corresponds, and shall be cumulative so that if the existence of the fact or item or its contents disclosed in any particular schedule is relevant to any other schedule, then such fact or item shall be deemed to be disclosed with respect to the other schedule to the extent such relevance is reasonably apparent whether or not a specific cross-reference appears. The headings contained in the Disclosure Schedules are included for convenience only, and are not intended to limit the effect of the disclosures contained in such schedule or to expand the scope of the information required to be disclosed in such schedule. Descriptions of documents in the Schedules are summaries only and are qualified in their entirety by the specific terms of such documents. Matters reflected in the Disclosure Schedules are not necessarily limited to matters required by this OTA to be reflected herein; additional matters are set forth for informational purposes and the fact that any item of information is disclosed in the Disclosure Schedules shall not be construed to mean that such information is required to be disclosed by this OTA. Any information and the dollar thresholds set forth herein shall not be used as a basis for interpreting the term "material" or other similar terms in this OTA or constitute an admission that such items are required to be disclosed under this OTA.
- Payment of Expenses. Except as otherwise provided in this OTA, each of the Parties shall bear its own expenses in connection with the negotiation and the consummation of the transactions contemplated by this OTA. Subject to the foregoing, no expenses of Transferor relating in any way to the purchase and sale of the Assets hereunder and the transactions contemplated hereby, including legal, accounting or other professional expenses of Transferor shall be charged to or paid by Transferee or included in any of the Assumed Liabilities. No expenses of Transferee relating in any way to the purchase and sale of the Assets hereunder and the transactions contemplated hereby, including legal, accounting or other professional expenses of Transferee shall be charged to or paid by any Transferor or included in any of the Excluded Liabilities. The foregoing shall not limit, however, any Party's right to include such expenses in any claim for damages against any other Party who breaches any legally binding provision of this OTA to the extent provided in this OTA.
- 12.3 Entire Agreement; Assignment; Etc. This OTA (including the Disclosure Schedules and all other schedules and exhibits hereto which are incorporated into and are a part of this OTA), together with the Purchase Agreement, and with any certificates and other instruments delivered hereunder, state the entire agreement of the Parties, merge all prior negotiations, agreements and understandings, if any, whether written or oral, and state in full all representations, warranties, covenants and agreements that have induced this OTA. Each Party agrees that in dealing with third parties no contrary representations

will be made. This OTA shall not be assignable by operation of Law or otherwise. The Parties acknowledge that Purchaser is a third party beneficiary of this OTA to the extent provided below.

- 12.4 <u>Captions</u>. The Article, Section and paragraph captions in this OTA are for convenience of reference only, do not constitute part of this OTA and shall not be deemed to limit or otherwise affect any of the provisions hereof.
- 12.5 <u>Severability</u>. The invalidity or unenforceability of any provision of this OTA shall not affect the validity or enforceability of any other provision of this OTA.

12.6 Enforcement.

- (a) The Parties agree that irreparable damage would occur in the event that any of the provisions of this OTA were not performed in accordance with their specific terms or were otherwise breached and that any breach of this OTA could not be adequately compensated in all cases by monetary damages alone. The Parties acknowledge and agree that, prior to the valid termination of this OTA pursuant to Section 9.1, the Parties shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this OTA and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which they are entitled at Law or in equity, but in all events subject to the limitations set forth in this OTA.
- (b) Nothing set forth in this <u>Section 12.6</u> shall require Transferor to institute any proceeding for (or limit Transferor's right to institute any proceeding for) specific performance under this <u>Section 12.6</u> prior or as a condition to exercising any termination right under <u>Section 9.1</u>, nor shall the commencement of any legal proceeding pursuant to this <u>Section 12.6</u> or anything set forth in this <u>Section 12.6</u> restrict or limit Transferor's right to terminate this OTA in accordance with the terms of <u>Section 9.1</u> or pursue any other remedies under this OTA.
- (c) To the extent any Party brings any Action to enforce specifically the performance of the terms and provisions of this OTA (other than an Action to specifically enforce any provision that expressly survives termination of this OTA pursuant to Section 9.1 when expressly available to such Party pursuant to the terms of this OTA, each Termination Date shall automatically be extended by (i) the amount of time during which such Action is pending, plus twenty (20) Business Days, or (ii) such other time period established by the court presiding over such Action.
- 12.7 <u>Modification or Amendment</u>. The Parties may modify or amend this OTA at any time, only by a written instrument duly executed and delivered by Transferee and Transferor Representative. Notwithstanding the foregoing, prior to the Closing, the Parties may not amend, modify or terminate this OTA without the prior written consent of Purchaser.
- 12.8 <u>Construction of Agreement</u>. If an ambiguity or question of intent or interpretation arises under this OTA, this OTA shall be construed as if drafted jointly by the Parties, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this OTA.
- 12.9 <u>Notices</u>. All notices and other communications given or made pursuant hereto shall be in writing and shall be deemed to have been duly given or made as of the date delivered if delivered personally or by a nationally recognized overnight courier service to the Parties at the following addresses (or at such other address for a Party as shall be specified by like notice, except that notices of changes of address shall be effective upon receipt):

If to Transferor, addressed to:

Kindred Healthcare, Inc. 680 South Fourth Street Louisville, Kentucky 40202

Attn: Joseph L. Landenwich, General Counsel

Attn: Douglas Curnutte, Senior Vice President, Corporate Development

With a copy to (which shall not constitute notice):

Polsinelli PC 401 Commerce Street, Suite 900 Nashville, TN 37219 Attn: Bobby Guy, Esq.

If to Transferee, addressed to:

Birchwood Operations LLC c/o Ari Erlichman
101 Lawrence Avenue
Lawrence NY 11559
arierlichman@gmail.com

With a copy to (which shall not constitute notice):

Shireen T. Hart Primmer Piper Eggleston & Cramer PC shart@primmer.com

or to such other address or to such other Person as either Party shall have last designated by such notice to the other Party.

- 12.10 <u>Remedies Cumulative</u>. Except as otherwise provided herein, the remedies provided for or permitted by this OTA shall be cumulative and the exercise by any Party of any remedy provided for herein shall not preclude the assertion or exercise by such Party of any other right or remedy provided for herein.
- 12.11 Governing Law; Consent to Jurisdiction. This OTA shall be governed by and construed in accordance with the domestic Laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware.

12.12 Forum; Waiver of Jury Trial.

- (a) With respect to any Action between any of the Parties arising out of or relating to this OTA, or any of the transactions contemplated by this OTA, (i) each of the Parties irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of either the state or federal courts located in the State of Delaware, and (ii) each of the Parties irrevocably consents to service of process by first-class certified mail, return receipt requested, postage prepaid.
- (b) Each of the Parties hereby irrevocably waives any and all right to trial by jury of any claim or cause of action in any legal proceeding arising out of or related to this OTA or the

transactions or events contemplated hereby or any course of conduct, course of dealing, statements (whether verbal or written) or actions of any Party. The Parties each agree that any and all such claims and causes of action shall be tried by the court without a jury. Each of the Parties hereto further waives any right to seek to consolidate any such legal proceeding in which a jury trial has been waived with any other legal proceeding in which a jury trial cannot or has not been waived.

- 12.13 <u>Time of Essence</u>. With regard to all dates and time periods set forth or referred to in this OTA, time is of the essence unless such delay is caused by factors outside the control of the Party in which case a reasonable delay shall be granted to the requesting Party.
- 12.14 <u>Counterparts</u>. This OTA may be executed in the original or by facsimile or electronic pdf in any number of counterparts, each of which shall be deemed to be an original and all of which together shall constitute one and the same instrument. The exchange of copies of this OTA and of signature pages by facsimile transmission or e-mail shall constitute effective execution and delivery of this OTA as to the Parties and may be used in lieu of the original OTA for all purposes. Signatures of the Parties transmitted by facsimile or e-mail shall be deemed to be their original signatures for all purposes.
- behalf and on behalf of its members, partners, officers, employees and Affiliates, that Transferor is a client of Polsinelli PC and Cleary Gottlieb Steen & Hamilton (collectively, the "Firms") in the preparation, negotiation and execution of this OTA and the other Transaction Documents. After the Closing, it is possible that the Firms will represent Transferor and/or its Affiliates in the future in connection with issues that may arise under this OTA and the other Transaction Documents or any claims that may be made thereunder. Each of the Firms (or any successor) may serve as counsel to Transferor and/or its Affiliates or any member, partner, manager, officer, employee, representative or Affiliate of such Persons in connection with any claim arising out of or relating to this OTA or the other Transaction Documents. Each of the Parties hereto consents thereto, and waives any conflict of interest arising therefrom, and each such Party shall cause any Affiliate thereof to consent to waive any conflict of interest arising from such representation. Each of the Parties hereto acknowledges that such consent and waiver is voluntary, that it has been carefully considered, and that the Parties have consulted with counsel or have been advised they should do so in connection therewith.
- 12.16 <u>Third-Party Beneficiary</u>. Purchaser shall be a third-party beneficiary of the representations and warranties of Transferor under <u>Article IV</u> and the indemnification provisions of <u>Article X</u>; provided, however, that Transferor shall have no obligation to indemnify (whether under this Article or otherwise) both (a) Purchaser, or an assignee of Purchaser other than Transferee, and (b) any Transferee with respect to any single Loss.

12.17 Transferor Representative.

(a) Transferor hereby irrevocably constitutes and appoints Kindred as its representative ("Transferor Representative") and its true and lawful attorney-in-fact, with full power and authority in each of their names and on behalf of each of them to act on behalf of Transferor in the absolute discretion of Transferor Representative for purposes of this OTA, the Purchase Agreement and the transactions to be carried out pursuant hereto and thereto, and the execution of this OTA, by Transferor will constitute ratification and approval of such designation on the terms set forth herein. All decisions, actions, consents and instructions by Transferor Representative with respect to this OTA will be binding upon Transferor, and Transferor will not have the right to object to, dissent from, protest or otherwise contest the same. Transferee will be entitled to rely on any decision, action, consent or instruction of Transferor Representative as being the decision, action, consent or instruction of Transferor. By way of example and not limitation, Transferor Representative will be authorized and empowered, as agent of and on behalf of Transferor to (i) execute and deliver and take all actions under the OTA on

behalf of Transferor; (ii) give and receive notices and communications as provided herein; (iii) object to any claims of an Indemnified Party; (iv) agree to, negotiate, enter into settlements and compromises of, and comply with orders of courts and awards of arbitrators with respect to, such claims or Losses; (v) waive after the Closing Date any breach or default of Transferee of any obligation to be performed by it under this OTA; (vi) receive service of process on behalf of Transferor in connection with any claims against Transferor arising under or in connection with this OTA; and (vii) take all other actions that are either (A) necessary or appropriate in the judgment of Transferor Representative for the accomplishment of the foregoing, or (B) specifically mandated by the terms of this OTA. Notices or communications to or from Transferor Representative will constitute notice to or from Transferor.

- (b) The grant of authority provided for in this <u>Section 12.17</u> is coupled with an interest and is being granted, in part, as an inducement to Transferee to enter into this OTA, and will be irrevocable and survive the dissolution, liquidation or bankruptcy of any Transferor, and will be binding on any successor thereto.
- 12.18 Attorney-Client Privilege. Neither of Transferor or Transferee is waiving, and each will not be deemed to have waived or diminished, any of its attorney work product protections, attorney-client privileges or similar protections and privileges as a result of disclosing its Confidential Information (including Confidential Information related to pending or threatened litigation) to the others, regardless of whether such Party has asserted, or is or may be entitled to assert, such privileges and protections. The Parties (a) share a common legal and commercial interest in all of the Confidential Information that is subject to such privileges and protections; (b) are or may become joint defendants in proceedings to which such Party's Confidential Information covered by such protections and privileges relates; (c) intend that such privileges and protections remain intact should any Party become subject to any actual or threatened proceeding to which the Confidential Information covered by such protections and privileges relates; and (d) intend that after the Closing the Proprietary Party whose Confidential Information is at issue shall have the right to assert such protections and privileges. No Party shall admit, claim or contend, in proceedings involving any Party or otherwise, that any Party waived any of its attorney workproduct protections, attorney-client privileges or similar protections and privileges with respect to any information, documents or other material not disclosed to a Party due to any Party disclosing its Confidential Information (including Confidential Information related to pending or threatened litigation) to another Party.
- 12.19 <u>Guaranty</u>; <u>Obligations of Transferee Guarantor</u>. Transferee Guarantor unconditionally guarantees the full and prompt payment and performance of all of Transferee's obligations to Transferor, Kindred, and Transferor Indemnified Parties in accordance with this OTA or any other agreement between Transferor and Transferee arising in connection with the Transaction. The liability of Transferee Guarantor under this <u>Section 12.19</u> will in no way be affected or impaired by any failure or delay by Transferor in enforcing payment of any amount required under this OTA, in enforcing the performance of any obligations under this OTA, in enforcing payment under this <u>Section 12.19</u>, or in exercising any right or power in respect thereto, or to any compromise, waiver, settlement, change, subordination, modification, or disposition of any payments due under this OTA or any performance required under this OTA or any of the other Transaction Documents, and Transferee Guarantor hereby waives all defenses of suretyship. The amendment or modification of this OTA will not affect Transferee Guarantor's liability under this <u>Section 12.19</u>, unless such Transferee Guarantor's liability is amended or modified in a writing signed by the Parties.
- 12.20 <u>Guaranty</u>; <u>Obligations of Kindred</u>. Kindred unconditionally guarantees the full and prompt payment and performance of all of Transferor's obligations to Transferee, Transferee Guarantor, and Transferee Indemnified Parties in accordance with this OTA or any other agreement between Transferor and Transferee arising in connection with the Transaction. The liability of Kindred under this <u>Section 12.20</u> will in no way be affected or impaired by any failure or delay by Transferee in enforcing

payment of any amount required under this OTA, in enforcing the performance of any obligations under this OTA, in enforcing payment under this Section 12.20, or in exercising any right or power in respect thereto, or to any compromise, waiver, settlement, change, subordination, modification, or disposition of any payments due under this OTA or any performance required under this OTA or any of the other Transaction Documents, and Kindred hereby waives all defenses of suretyship. The amendment or modification of this OTA will not affect Kindred's liability under this Section 12.20, unless such Kindred's liability is amended or modified in a writing signed by the Parties.

[The Next Page is the Signature Page]

IN WITNESS WHEREOF, each of the undersigned in the capacity indicated below has executed this OTA as of the day and year first above written.

TRANSFEROR:

Kindred Nursing Centers East, L.L.C.

Douglas L. Curnutte

Senior Vice President, Corporate Development

KINDRED:

Kindred Healthcare Operating, Inc.

By: _______ Douglas L. Curnutte

Senior Vice President, Corporate Development

TRANSFEREE:

Birchwood Operations LLC

Name: Ari Erlichman

Title: Owner

TRANSFEREE GUARANTOR:

E&R Operations LLC

Name: Ari Erlichman Title: Owner

Exhibit A

Definitions

- <u>Definitions</u>. In addition to the terms otherwise defined herein, the following terms shall have the following meaning:
- "Accounts Receivable" means all accounts receivable and incentive payments of the Business, including without limitation, the IGT Credit Amount and other incentive payments related to the QIPP, QASP or similar incentive programs in additional states.
 - "Action" has the meaning set forth in the Purchase Agreement.
 - "Affected Participants" has the meaning set forth in Section 2.5(m).
 - "Affiliate" has the meaning set forth in the Purchase Agreement.
 - "Affiliated-Service Transferee Employees" has the meaning set forth in Section 2.5(d).
 - "Ancillary Permits and Approvals" has the meaning set forth in Section 2.2(a).
 - "A/R Collection Period" has the meaning set forth in Section 6.10(b).
 - "Assets" has the meaning set forth in Section 2.1.
 - "Assignment and Assumption Agreement" has the meaning set forth in Section 3.2(b).
 - "Assumed Contracts" has the meaning set forth in Section 2.1(e).
 - "Assumed Liabilities" has the meaning set forth in Section 2.4(a).
 - "Bill of Sale" has the meaning set forth in Section 3.2(a).
 - "Bring Down Certificate" has the meaning set forth in Section 3.2(c).
 - "Broker" has the meaning set forth in Section 4.18.
- "Business" means the business conducted by Transferor exclusively at or exclusively related to the Facility.
 - "Business Days" has the meaning set forth in the Purchase Agreement.
 - "Closing" has the meaning set forth in Section 3.1.
 - "Closing Date" has the meaning set forth in Section 3.1.
 - "CMS" means the Centers for Medicare and Medicare Services.
 - "COBRA" means the Consolidated Omnibus Budget Reconciliation Act or similar state law.
 - "Code" has the meaning set forth in the Purchase Agreement.
 - "Compliance Violations" has the meaning set forth in Section 3.3(h).

- "Confidential Information" has the meaning set forth in Section 6.5(b).
- "Contracts" has the meaning set forth in Section 2.1(e).
- "Cost Reports" means all Cost Reports exclusively related to the Facility filed by Transferor prior to the Execution Date pursuant to the requirements of any applicable Government Reimbursement Programs for cost-based payments or reimbursement due to or claimed by Transferor from any applicable Government Reimbursement Programs or their fiscal intermediaries or payor agents.
 - "Current Records" has the meaning set forth in Section 2.1(h).
 - "Deficiencies" has the meaning set forth in Section 3.3(g).
 - "Delayed Closing" has the meaning set forth in Section 8.1.
 - "Department" has the meaning set forth in Section 2.2(a).
 - "Disclosure Updates" has the meaning set forth in Section 4.21.
 - "Effective Time" has the meaning set forth in Section 3.1.
- "Employee Benefit Plan" means any plan, program, agreement or policy for the benefit of any current or former employee, director, independent contractor, or owner (or any dependent or beneficiary thereof) that is (a) a welfare plan within the meaning of Section 3(1) of ERISA, (b) a pension plan within the meaning of Section 3(2) of ERISA, (c) a stock bonus, stock purchase, stock option, restricted stock, stock appreciation right or similar equity-based plan, or (d) any other compensation, deferred-compensation, retirement, welfare-benefit, bonus, incentive, retention, severance pay, sick leave, vacation pay, salary continuation, disability, dental, vision, medical, life insurance or fringe-benefit plan, program, agreement or policy.
 - "Encumbrance" has the meaning set forth in the Purchase Agreement.
 - "Environmental Law" has the meaning set forth in the Purchase Agreement.
 - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
 - "ERISA Affiliate" has the meaning set forth in Section 4.5(c).
 - "Escrow Facility" has the meaning set forth in the Purchase Agreement.
- "Excluded Assets" shall mean the Assets which are not being transferred to Transferee from Transferor as described in Section 2.7.
 - "Execution Date" has the meaning set forth in the Preamble.
 - "Expiration Date" has the meaning set forth in Section 10.1.
 - "Facility" has the meaning set forth in the Recitals.
- "Financial Statements" means the unaudited balance sheets and profit and loss statements relating to the operations of the Business for the 2016 fiscal year.
 - "Firms" has the meaning set forth in Section 12.15.

- "Fundamental Representations" has the meaning set forth in Section 10.1.
- "GAAP" means U.S. generally accepted accounting principles, as in effect on the Execution Date, consistently applied.
- "Government Reimbursement Program" means the Medicare program, any relevant state Medicaid program and any other similar or successor federal, state or local health care payment programs with or sponsored by any Governmental Authority (excluding the TRICARE Program).
- "Governmental Authority" or "Governmental Entity" means any federal, state, or local government or any court of competent jurisdiction, administrative agency or commission or other domestic governmental or quasi-governmental authority or instrumentality.
- "Group Health Plan" means a group health plan offering major medical, dental and other medical coverage subject to COBRA.
- "Hazardous Substances" means any chemicals, materials, compounds or substances defined, regulated, listed or otherwise classified under any applicable Law as a "hazardous substance," "extremely hazardous substance," "hazardous material," "hazardous waste," "universal waste," "mixed waste," "bio-hazardous waste," "medical waste," "radioactive waste," "pharmaceutical waste," "commingled waste," "mold," "toxic substance," "toxin," "pollutant" or "contaminant," including petroleum (including petroleum products, constituents, additives, or derivatives thereof), asbestos, asbestos-containing materials, and polychlorinated biphenyls.
- "Healthcare Requirements" means the requirements of or with respect to Government Reimbursement Programs, Referral Laws, Patient Privacy Requirements, the False Claims Act, 31, U.S.C. Section 3729 et seq. as amended, and 42 USC Section 1320a-7k(d), 42 U.S.C. 1320a-7a(a).
 - "HIPAA" means the Health Insurance Portability and Accountability Act of 1996.
- "IGT Credit Amount" shall mean the amount of any cash or other amounts due to Transferor (regardless of whether such amounts are paid after the applicable Closing) which is set forth on the IGT balance sheets of the Facility as of immediately prior to the applicable Closing, and any payment or part thereof received pursuant to a UPL Program related to or attributable to the period prior to the Effective Time.
 - "IMA" has the meaning set forth in Section 4.1(b).
 - "Indemnification Cap" has the meaning set forth in the Purchase Agreement.
- "Indemnification Obligation" means the indemnification obligations of an Indemnifying Party under this OTA.
- "Indemnified Party" and "Indemnified Parties" means any Person entitled to indemnification under $\underline{\text{Article } X}$ of this OTA and Article IX of the Purchase Agreement.
- "Indemnifying Party" and "Indemnifying Parties" means any Person required to provide indemnification under Article X of this OTA and Article IX of the Purchase Agreement.
 - "Indemnity Threshold" has the meaning set forth Section 10.4(a).
- "Intellectual Property" has the meaning set forth in the Purchase Agreement (excluding such intellectual property listed as an Excluded Asset).

"Inventory" has the meaning set forth in Section 2.1(d).

"Kindred" has the meaning set forth in the Preamble.

"Knowledge" means, when used with respect to Transferor, the actual awareness after due inquiry of a particular fact or matter of any of the following: Michael Beal, President Nursing Center Division; Jerry Kemper, Chief Financial Officer Nursing Center Division; Jill Bosa, Chief Operating Officer Nursing Center Division, Melonie McManus, Chief Clinical Officer Nursing Center Division, Patricia McGillan, Chief Counsel Nursing Center Division, and Glenn Cote, DVP, Facilities Management Nursing Center Division.

"Law" means any statute, law, rule or regulation or ordinance of any Governmental Authority.

"Loss" or "Losses" has the meaning set forth in the Purchase Agreement.

"LSC" has the meaning set forth in Section 3.3(g).

"Material Adverse Effect" means any event, change, development or occurrence that has had or would reasonably be expected to have a material and adverse effect on the operations, condition (financial or otherwise) or results of operations of the Business, taken as a whole, but excluding any such event, change, development or occurrence attributable to or resulting from (i) any change in applicable Law or the interpretation thereof, (ii) any change in GAAP or the interpretation thereof, (iii) any events, changes, developments or occurrences generally affecting the industries in which the Business operates, (iv) general economic, political or market conditions, (v) any disasters, calamities, emergencies, acts of war, sabotage or terrorism (or an escalation or worsening of any of the foregoing), (vi) the entry into or announcement of this OTA and the transactions contemplated hereby, (vii) any action taken or omitted to be taken by Transferor or its Affiliates pursuant to this OTA or at the written request or with the prior written consent of Transferee, (viii) any loss of, or change in, the relationship of the Business with its customers, employees or suppliers (but not any breach of Contract by Transferor or its Affiliate) that is a direct result of the execution, delivery or performance (in accordance with its terms) of this OTA, the consummation of the transactions contemplated by this OTA or the announcement of any of the foregoing, (ix) the failure of the Business to achieve internal or external financial forecasts or projections, provided that the events, changes, developments or occurrences underlying such failure shall not be excluded as a result of this clause (ix), or (x) any breach by Transferee of this OTA.

"Medicaid Agreement" shall have the meaning set forth in Section 2.2(b)(ii).

"Medicare Agreement" shall have the meaning set forth in Section 2.2(b)(i).

"National Contracts" means all Contracts between Kindred, Transferor, or any of their respective Affiliates, on the one hand, and any third party, on the other hand, that have been entered into on a national or regional basis including, without limitation, any Contract pursuant to which any services are provided by or to any hospital, inpatient rehabilitation facility, nursing facility or other facility of Kindred or any of its Affiliates that is not the Facility.

"New License" shall have the meaning set forth in Section 2.2(a).

"Non-Competition Covenant" has the meaning set forth in Section 6.3.

"Non-Senior Executive Employee Liquidated Damages" has the meaning set forth in Section 6.3.

"Non-Solicitation Covenant" has the meaning set forth in Section 6.3.

- "Notice of Indemnification" has the meaning set forth in Section 10.5.
- "Novation" has the meaning set forth in Section 2.2(c).
- "Order" has the meaning set forth in the Purchase Agreement.
- "OSHPD" means the California Office of Statewide Health Planning and Development.
- "OSHPD Work" has the meaning set forth in Section 4.8(c).
- "OTA" has the meaning set forth in the Preamble.
- "Other Operations Transfer Agreements" means those certain Operations Transfer Agreements dated the same date hereof, by and between the Affiliates of Transferor and Transferee for the purposes of transferring the operations of facilities located in Vermont.
- "Outside Date" means September 30, 2018, in all instances, provided that such date shall be October 31, 2018, with respect to and Escrow Facility eligible for Closing hereunder except for satisfaction of the closing conditions set forth in Section 7.1, so long as the applicable parties are using its good faith efforts to satisfy such closing conditions.
- "Overhead and Shared Services" means ancillary corporate or shared services provided to or in support of any Facility that are general corporate, overhead or other services or provided to both (a) the Facility, and (b) any other business or facility of Seller and its Affiliates that is not a Facility including, without limitation, access to hardware and software related to financial and clinical operations, use of intellectual property, travel and entertainment services, temporary labor services, purchasing and supply services, personal telecommunications services, computer hardware and software services, energy/utilities services, treasury services, public relations, legal and risk management services (including workers' compensation), payroll services, sales and marketing support services, information technology and telecommunications services, accounting services, tax services, internal audit services, executive management services, investor relations services, human resources and employee relations management services, employee benefits services, credit, collections and accounts payable services, logistics services, property management services, environmental support services, training, federal and state reimbursement services, state licensing and Medicare and Medicaid certification and maintenance support, in each case including services relating to the provision of access to information, operating and reporting systems and databases and all hardware and software or other intellectual property used in connection therewith.
 - "Party" and "Parties" have the meaning set forth in the Preamble.
- "Patient Privacy Requirements" means the applicable requirements of the Administrative Simplification Provisions of the Health Insurance Portability and Accountability Act of 1996 as amended by the American Recovery and Reinvestment Act of 2009 and the implementing regulations thereunder governing the privacy of individually identifiable health information and the security of such information maintained in electronic form or of any similar state Laws.
 - "Pending Medicaid Applicants" has the meaning set forth in Section 6.10(e).
 - "Permits" has the meaning set forth in Section 2.1(i).
 - "Permitted Encumbrance" has the meaning set forth in the Purchase Agreement.

"Person" means an individual, partnership, venture, unincorporated association, organization, syndicate, corporation, limited liability company, or other entity, trust, trustee, executor, administrator or other legal or personal representative or any government or any agency or political subdivision thereof.

"Plan" has the meaning set forth in Section 4.5(a).

"Policy and Procedure Manual" has the meaning set forth in Section 2.1(f).

"Policy Return Date" has the meaning set forth in Section 2.1(f).

"Portfolio" has the meaning set forth in the Purchase Agreement.

"Portfolio Material Adverse Effect" means any event, change, development or occurrence that has had or would reasonably be expected to have a material and adverse effect on the operations, condition (financial or otherwise) or results of operations of the Portfolio (as such term is defined in the Purchase Agreement) in which the Facility is located, taken as a whole, but excluding any such event, change, development or occurrence attributable to or resulting from (i) any change in applicable Law or the interpretation thereof, (ii) any change in GAAP or the interpretation thereof, (iii) any events, changes, developments or occurrences generally affecting the industries in which the Business operates, (iv) general economic, political or market conditions, (v) any disasters, calamities, emergencies, acts of war, sabotage or terrorism (or an escalation or worsening of any of the foregoing), (vi) the entry into or announcement of this OTA and the transactions contemplated hereby, (vii) any action taken or omitted to be taken by Transferor or Transferor its Affiliates pursuant to this OTA or at the written request or with the prior written consent of Transferor, (viii) any loss of, or change in, the relationship of the Business with its customers, employees or suppliers (but not any breach of Contract by Transferor or its Affiliate) that is a direct result of the execution, delivery or performance (in accordance with its terms) of this OTA, the consummation of the transactions contemplated by this OTA or the announcement of any of the foregoing, (ix) the failure of the Business to achieve internal or external financial forecasts or projections, provided that the events, changes, developments or occurrences underlying such failure shall not be excluded as a result of this clause (ix), or (x) any breach by Transferee of this OTA.

"Pre-Closing Imposition(s)" one or more deficiencies identified on the most recent pre-Closing survey of the Facility by a Governmental Entity that remains unresolved and results in or imposes (1) a deficiency rating of "I", "J" or higher, (2) civil money penalties, or (3) denial of payment for new admission (DPNA).

"Prior Records" has the meaning set forth in Exhibit 2.7.

"Public Announcement" has the meaning set forth in Section 6.5.

"Purchase Agreement" has the meaning set forth in the Recitals.

"Purchaser" has the meaning set forth in the Recitals.

"Qualified Plan" has the meaning set forth in Section 4.5(b).

"Records" has the meaning set forth in Exhibit 2.7.

"Referral Laws" means Section 1128B(b) of the Social Security Act, as amended, 42 USC Section 1320a 7(b) (Criminal Penalties Involving Medicare or State Health Care Programs), commonly referred to as the "Federal Anti-Kickback Statute," Section 1877 of the Social Security Act, as amended, 42 USC Section 1395nn and related regulations (Prohibition Against Certain Referrals), commonly referred to as "Stark Law," 42 USC Section 1320a-7a(a)(5).

- "Regulatory Approvals" shall have the meaning set forth in Section 2.2(a).
- "RehabCare" means RehabCare Group, Inc. or its Affiliate.
- "Resident" means a resident of the Transferred Facilities.
- "Resident Trust Funds" has the meaning set forth in Section 2.1(k).
- "Restricted Party" has the meaning set forth in Section 6.3.
- "Restricted Period" has the meaning set forth in Section 6.3.
- "Retained Liabilities" has the meaning set forth in Section 2.4(b).
- "Return" and "Returns" have the meaning set forth in the Purchase Agreement.
- "Seller" has the meaning set forth in the Purchase Agreement.
- "Seller Party" has the meaning set forth in the Purchase Agreement.
- "Senior Executive" has the meaning set forth in Section 6.3.
- "Senior Executive Liquidated Damages" has the meaning set forth in Section 6.3.
- "Substitute OTA Transferee" has the meaning set forth in the Purchase Agreement.
- "Tax" and "Taxes" have the meaning set forth in the Purchase Agreement.
- "Termination Date" has the meaning set forth in the Purchase Agreement.
- "Transaction Documents" has the meaning set forth in the Purchase Agreement.
- "Transferee" has the meaning set forth in the Preamble.
- "Transferee Employees" has the meaning set forth in Section 2.5(b).
- "Transferee Guarantor" means E&R Operations LLC.
- "Transferee Indemnified Parties" has the meaning set forth in Section 10.2.
- "Transferee's A/R" has the meaning set forth in Section 6.10(a).
- "Transferee Plan" has the meaning set forth in Section 2.5(m).
- "Transferor" has the meaning set forth in the Preamble.
- "Transferor Bad Debt" has the meaning set forth in Section 6.11(b).
- "Transferor Confidential Information" has the meaning set forth in Section 6.5(c).
- "Transferor Indemnified Parties" has the meaning set forth in Section 10.3.
- "Transferor Representative" has the meaning set forth in Section 12.17(a).

- "Transferor's A/R" has the meaning set forth in Section 6.10(a).
- "Transferred Employees" has the meaning set forth in Section 2.5(b).
- "Transferred Facilities" has the meaning set forth in the Purchase Agreement.
- "Transition Period" has the meaning set forth in Section 2.2(b)(ii).
- "UPL Contract" has the meaning set forth in the Purchase Agreement.
- "UPL Program" has the meaning set forth in the Purchase Agreement.
- "VA" has the meaning set forth in Section 2.2(c).
- "VA Contract" has the meaning set forth in Section 2.2(c).
- "VA Subcontract" has the meaning set forth in Section 2.2(c).
- "VDR" has the meaning set forth in the Purchase Agreement.
- "WARN Act" means the Worker Adjustment and Retraining Notification Act of 1988, as amended.

Exhibit 2.2 (c)

SUBCONTRACT PENDING NOVATION

THIS SUBCONTRACT PENDING NOVATION (this "Agreement") is made as of this day of, 2017, by and between, a ("Contractor"), and, a ("Successor"). Reference is made herein to that certain Operations Transfer and Surrender Agreement dated as of, 2017, to which Successor and Contractor are parties (the "OTA"), the terms and conditions of which are incorporated herein by reference.
<u>RECITALS</u> :
WHEREAS, Contractor is transferring to Successor, and Successor is assuming from Contractor, the operations of that certain skilled nursing facility located at and commonly known as, effective as of, 2017 (the "Closing Date"), according to the terms set forth in the OTA;
WHEREAS, Contractor is currently a party to a contract with the United States of America (the "Government"), Contract Number (the "VA Contract"), and Successor is assuming the obligations and liabilities of Contractor thereunder in accordance with Section 2.2(c) of the OTA;
WHEREAS, Successor and Contractor will file an application for novation of the VA Contract (the "Novation Application") pursuant to which the parties will request that the Government recognize Successor as successor to Contractor under the VA Contract; and
WHEREAS, the parties hereto desire to set forth their rights and obligations with respect to the continued performance of the VA Contract pending approval of the Novation Application.
NOW, THEREFORE, the parties, intending to be legally bound, hereby agree as follows:
1. <u>Statement of Work.</u> On the terms and subject to the conditions hereinafter set forth, Contractor subcontracts to Successor all aspects of performance of the VA Contract during the term of this Agreement. Successor agrees to comply with all terms and conditions of the VA Contract including, but not limited to, the Federal Acquisition Regulation clauses contained therein. Contractor agrees to cooperate with Successor and to take such actions as may be reasonably requested by Successor to facilitate Successor's performance of the VA Contract during the Term (as defined herein).
2. <u>Term.</u> The term of this Agreement shall commence as of the Closing Date, and shall continue until the date on which the Novation Application is approved or the date upon which a final denial of such application is received by Contractor or Successor, unless sooner terminated as provided herein (the " <i>Term</i> "). Contractor shall have the right to terminate this Agreement at any time if such termination is required by the Government or is otherwise necessary to avoid Contractor's termination for default under the VA Contract.
3. <u>Successor's Duties</u> . During the Term, Successor shall:
(a) keep accurate accounting records showing all services performed and expenditures made or incurred in connection with its performance of the VA Contract;
(b) keep such records and underlying receipts as may be required under the VA Contract;

- (c) perform the VA Contract in a legal manner, abiding by all applicable terms, conditions, laws, and regulations; and
- (d) maintain and pay premiums for comprehensive bodily injury, liability (professional and general) and property damage insurance in amounts sufficient to provide reasonable and adequate protection of both Successor and Contractor against liability which may arise in connection with the performance of the VA Contract.
- 4. <u>Contractor's Duties</u>. During the Term, Contractor shall use its commercially reasonable efforts to cooperate with Successor in pursuing the Novation Application including, but not limited to, using commercially reasonable efforts to provide Successor with all documents and information necessary for Successor to seek a novation of the VA Contract and signing any novation agreement and other forms reasonably requested by Successor or the Government. Contractor shall also take such actions reasonably required by Successor to facilitate Successor's performance of the VA Contract and the receipt of payments by Successor for services and goods rendered to the Government by Successor under the VA Contract including, but not limited to, submitting invoices prepared by Successor to the Government on behalf of Successor for goods and services rendered to the Government by Successor under the VA Contract, receiving payments on behalf of Successor on such invoices from the Government, and promptly remitting such payments to Successor.
- 5. <u>Compensation</u>. In consideration of the services to be rendered by Successor to Contractor hereunder, Successor shall be entitled to retain all revenues from goods and services provided by Successor during the Term to the Government under the VA Contract (the "*Compensation*"). Notwithstanding anything contained herein, the Compensation shall be deemed Accounts Receivable (as defined in the OTA) and subject to Section 6.10 of the OTA.
- 6. <u>Document Delivery</u>. Contractor shall deliver to Successor copies of all documents and information that are received by Contractor with respect to the VA Contract.
- 7. <u>No Partnership or Joint Venture or Agency</u>. Successor and Contractor are not partners, joint venturers or agents and nothing herein shall be so construed. Successor shall perform its duties hereunder solely as an independent contractor of Contractor.
- 8. <u>Indemnification</u>. Successor hereby agrees to indemnify and hold Contractor harmless from and against any claim, liability, loss, damage, cost, expense or other deficiency including, but not limited to, reasonable attorneys' fees and other legal costs and expenses in any way arising out of, resulting from, or relating to (i) Successor's breach of any representation or warranty made by Successor herein; or (ii) failure of Successor to perform any of its obligations as set forth herein.
- 9. <u>Amendments: Binding Effect.</u> Except as provided in Section 2 hereof, this Agreement shall not be modified or terminated except by an instrument in writing signed by both parties hereto or their respective successors or assigns, and shall be binding on and inure to the benefit of the parties hereto and their respective successors and assigns.
- 10. <u>Subcontracting and Assignment</u>. Successor shall not further subcontract or assign its rights or obligations under this Agreement without Contractor's express written consent.
- 11. <u>Counterparts</u>. This Agreement may be signed in one or more counterparts, each of which shall be deemed an original but all of which shall constitute one and the same instrument.

IN WITNESS WHEREOF, the parties have executed this Subcontract Pending Novation as of the date first above written.

Contractor:	Successor:			
				
By:	By:			
Name:	Name:			
Title:	Title:			

Exhibit 2.7

Excluded Assets

- (a) Any cash, cash equivalents, or bank accounts;
- (b) All Accounts Receivable, prepaid expenses, security deposits and other current assets of the Facility (excluding deferred tax assets);
 - (c) All National Contracts;
- (d) All Overhead and Shared Services, including any Contracts for or assets related to Overhead and Shared Services;
- (e) Licenses and permits that are not assignable or transferable, whether with or without third party consent, to Transferee;
- (f) Assets of Transferor disposed of in the ordinary course of business prior to the Effective Time; provided that Transferor shall not dispose of any material Assets without the prior written consent of Transferee (other than Inventory used at the Facility in the ordinary course of business, which may be used and disposed of provided that it shall also be replenished to a quantity that is required by Law);
- (g) Any management agreement between Transferor and Kindred or its Affiliates, as the case may be;
 - (h) All insurance policies and any claims and rights to proceeds thereunder;
- (i) The minute books and ownership records of Transferor, including all organizational documents, stock registers and such other Records of Transferor as they pertain to the ownership, organization, or existence of Transferor and duplicate copies of such records;
- (j) Any claims for refunds of Taxes and other governmental charges imposed on Transferor of whatever nature including, but not limited to, those with respect to the Facility or the business attributable to periods ending on or prior to the Closing Date;
- (k) All shares of any capital stock, membership interests or partner interests in any partnership, of Transferor;
 - (1) All of Transferor's email accounts;
 - (m) All rights of Transferor under this OTA or the other Transaction Document;
- (n) All insurance policies of Transferor or any of its Affiliates and all rights of every nature and description under or arising out of such insurance policies, including the right to make claims thereunder, to the proceeds thereof and to any insurance refunds relating thereto;
- (o) Transferor's Returns for periods up to and including the Closing Date and all rights of Transferor to any recoveries or refunds in respect of Taxes for periods up to and including the Closing Date, whether or not any refund of or credit for claims have been filed prior to the Closing Date;
 - (p) Transferor's attorney-client privilege;
 - (q) All Employee Benefit Plans (including Plans) and all assets related thereto;

- (r) Transferor's information technology systems, emails, software licenses, corporate minute books, records, marketing materials, policies and procedures, and all assets that are used at the corporate level and do not solely relate to the operations of the Business;
- (s) All claims or rights of Transferor with and among any other Transferor or amounts due from related parties;
- (t) All of Kindred's or any of its Affiliate's proprietary manuals, marketing materials, policy and procedure manuals, standard operating procedures and marketing brochures, and all data and studies or analyses generated for the benefit of the Facility;
- (u) All funds and accounts of all employee retirement, deferred compensation, health, welfare or benefit plans and programs, including assets representing a surplus or overfunding of any Employee Benefit Plan;
- (v) All unclaimed property of any third party as of the Closing, including, without limitation, property which is subject to applicable escheat Laws;
- (w) All assets of Transferor not used in connection with or held in whole or in part for use in connection with the Business;
- (x) The items of personal property brought to the Facility by employees of Transferor or its Affiliates that are not used or held for use with the Business and the operation of any of the Facility;
- (y) All tradenames, trademarks, service marks, domain names (URLs) and websites owned by Kindred or its Affiliates including, without limitation, any use of the names "Kindred" or "RehabCare," in whole or in part, or any derivation thereof, and all references to any of the foregoing on social media channels (including, without limitation, Facebook, Twitter and YouTube) associated with any or all of the Facility or Kindred or its Affiliates;
- (z) All files, charts, and other information relating to all Residents who previously occupied the Facility or used the Facility prior to the Effective Time and are not Residents of the Facility as of the Effective Time (including, but not limited to, all patient records, medical records, therapy records, pharmacy records, clinical records, and Resident Trust Funds records) for all periods prior to the date that is three (3) years before the Effective Time (collectively, "*Prior Records*" and together with the Current Records, "*Records*");
- (aa) And any asset transferred to Purchaser, its Affiliate or assignee under the terms of the Purchase Agreement;
- (bb) Any assets listed as an "Excluded Asset" pursuant to the terms of the Purchase Agreement;
 - (cc) All assets owned by RehabCare;
- (dd) all claims, rights, interests and proceeds (whether received in cash or by credit to amounts otherwise due to a third party) with respect to amounts overpaid by Transferor to any third party with respect to periods prior to the Closing (e.g., such overpaid amounts may be determined by billing audits undertaken by Transferor or Transferor's consultants) to the extent not offset against any underpayments by any applicable third party payor in respect of services rendered prior to the Closing;
- (ee) any receipts (i) relating to Transferor's Cost Reports or rights to settlements and retroactive adjustments on the same (whether resulting from an appeal by Transferor or otherwise) with

respect to time periods prior to the Closing, or (ii) which result from Transferor's pursuit of one or more appeals pertaining to a Government Reimbursement Program to the extent not offset against any overpayments by such Government Reimbursement Program in respect of services rendered prior to the Closing.

Exhibit 3.2(a)

BILL OF SALE

In consideration of One Dollar (\$1.00) and other go	od and valuable con	sideration, the receipt and
sufficiency of which are hereby acknowledged,	, a	(" <i>Transferor</i> "), does
hereby grant, bargain, sell, convey, assign and transfer all o	f Transferor's right, t	title and interest in and to
all and singular, the Assets to, a	("Transfere	ee"), and Transferee does
hereby purchase, take accept and assume, all of Transferd		
singular, the Assets. Capitalized terms used herein withou	t definition shall har	ve the meanings given to
them in that certain Operations Transfer and Surrender Agr	reement dated as of	, 2017, by
and among Transferor, Transferee, [Transferor Guaranty],	, a,	and Kindred Healthcare
Operating, Inc., a Delaware corporation (the "OTA").		

TO HAVE AND TO HOLD, all and singular, for Transferee's use and benefit, and Transferor hereby represents and warrants to Transferee that Transferor has full right, power and authority to sell the foregoing Assets and to make this Bill of Sale and that the foregoing Assets are free and clear of all liens and encumbrances except for the Permitted Encumbrances.

Nothing contained in this Bill of Sale shall be deemed to supersede or change any of the obligations, agreements, provisions, covenants, warranties or representations of Transferor or Transferee in the OTA, which obligations, agreements, provisions, covenants, warranties and representations shall remain in full force and effect to the full extent provided therein.

[Remainder of this Bill of Sale intentionally left blank – signature page to follow]

IN WITNESS WHEREOF, Transferor has caused this Bill of Sale to be duly executed as of the first date written above.

TRANSFEROR:		
Ву:		
Name:		
Title:		
TRANSFEREE:	8)	**
Ву:		
Name:		
Title:		

Exhibit 3.2(b)

ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND	ASSUMPTION AGREEM	ENT (this "Assignment") is made and
entered into as of	, 2017, by and between	, a
("Assignor"),	and	, a
("Assignee"). Capitalized terms used	herein without definition sha	Il have the meanings given to them in
the Operations Transfer and Surren	der Agreement dated	, 2017, by and among
Assignor, Assignee, and Kindred Hea	lthcare Operating, Inc., a Dela	ware corporation (the "OTA").

WITNESSETH:

WHEREAS, pursuant to the OTA, Assignor has agreed to sell, transfer, assign, and deliver to Assignee, and Assignee has agreed to accept from Assignor all existing contracts described in Exhibit A attached hereto and incorporated by reference herein (the "Assumed Contracts"); and

WHEREAS, pursuant to the OTA, Assignee has agreed to accept and assume Assignor's rights, claims, and interests with respect to the Assumed Contracts, and to assume the obligations under the Assumed Contracts which relate to the period of time after the Effective Time.

- **NOW, THEREFORE,** in consideration of the foregoing premises and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:
- Section 1. <u>Assignment and Assumption</u>. Effective as of the Closing, Assignor hereby transfers, assigns and delivers to Assignee all right, title and interest of Assignor in and to the Assumed Contracts.
- Section 2. <u>Assignee's Acceptance and Assumption</u>. Assignee hereby accepts such transfer, assignment and delivery of the Assumed Contracts from Assignor and expressly assumes any and all rights, responsibilities, obligations and liabilities of Assignor in connection with the Assumed Contracts which relate to the period of time after the Effective Time.
- Section 3. <u>Appointment</u>. Assignor hereby irrevocably appoints Assignee, its successors and assigns, as the attorney and agent of Assignor, in Assignor's name and stead, to enforce the provisions of the Assumed Contracts.
- Section 4. <u>Binding Effect</u>. This Assignment shall be binding on and inure to the benefit of Assignor and Assignee and their respective successors and assigns.
- Section 5. <u>Captions</u>. The captions of this Assignment are solely for the convenience of reference and shall not affect its interpretation.
- Section 6. <u>Counterparts</u>. This Assignment may be executed in counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument.
- Section 7. Governing Law. This Assignment shall be governed by and interpreted in accordance with the laws of the State of Delaware without giving effect to its conflict of law principles.

[Remainder of this Assignment intentionally left blank – signature page to follow]

IN WITNESS WHEREOF, the parties hereto have executed this Assignment and Assumption Agreement as of the day and year first above written.

ASSIGNOR:	
By:	
Its:	
ASSIGNEE:	
Ву:	
Name:	
T.	

Exhibit A

Assumed Contracts

Exhibit 3.2(c)

BRING DOWN CERTIFICATE

	<u>u</u>	, 2017
	, a	("[Transferee/Transferor]") hereby certifies that:
"OTA"), by an	Surrender Agreement dat	delivered pursuant to [Section 3.2(c)] of that certain Operations ed, 2017 (hereinafter referred to as the sferor],, a, and Kindred reporation.
IV/Article V] t qualified as to except to the e respect to a pa material respec	the OTA are true and corre- materiality, true and corre- xtent any such representation articular period, in which of	s and warranties of [Transferee/Transferor] set forth in [Article ct in all material respects (or, with respect to any representation ct) as of the date hereof as if made on and as of the date hereof, on or warranty expressly was made as of an earlier date or with ase such representation or warranty was true and correct in all representation qualified as to materiality, true and correct) as of
(iii) terms, agreeme date hereof.		has performed and complied, in all material respects, with all ons of the OTA to be performed or complied with by it as of the
ГРама	inder of this Pring Down (Contificate intentionally left blank signature nage to follow!

IN WITNESS WHEREOF, the undersigned has executed this Bring Down Certificate as of the first date written above.

[TRANSI	CEREE	TRAN	SFERU	ıkj:	
-		-			
By:					
By: Name:					
Title:					

Exhibit 6.10(f)

Wire Instructions

If to Kindred:

ABA# 021000021

Bank Name: JPMorgan Chase Bank

Bank Address: 270 Park Avenue, New York, NY 10017 Account Name: Kindred Healthcare Operating, Inc.

Account Number: 323272681

SWIFT CODE: CHASUS33

Kindred Wire Contacts:

Temesha Smith Keff Helstowski 502-596-2007

502-596-2971

If to Purchaser:

[to be provided by OTA Transferee within thirty (30) days following the Execution Date]

DISCLOSURE SCHEDULES (VERMONT - BIRCHWOOD)

These Disclosure Schedules are made and given pursuant to those certain Operations Transfer and Surrender Agreements by and between Kindred Healthcare Operating, Inc., a Delaware corporation ("Kindred"); Kindred's affiliated seller entities which operate the Facilities listed on Exhibit A of these Disclosure Schedules ("Transferors"); and Birchwood Operations LLC, a Vermont limited liability company ("Transferee"), dated June 30, 2017 (the "Agreement"). Capitalized terms used herein and not otherwise defined herein have the meanings ascribed to them in the Agreement.

The information contained in the Disclosure Schedules shall be deemed to qualify to the specific Section (or subsection, as appropriate) of the Agreement to which it corresponds, and shall be cumulative so that if the existence of the fact or item or its contents disclosed in any particular Schedule is relevant to any other Schedule, then such fact or item shall be deemed to be disclosed with respect to the other Schedule to the extent such relevance is reasonably apparent whether or not a specific cross-reference appears. The headings contained in the Disclosure Schedules are included for convenience only, and are not intended to limit the effect of the disclosures contained in such Schedule or to expand the scope of the information required to be disclosed in such Schedule. Descriptions of documents in the Schedules are summaries only and are qualified in their entirety by the specific terms of such documents. Matters reflected in the Disclosure Schedules are not necessarily limited to matters required by the Agreement to be reflected herein; additional matters are set forth for informational purposes and the fact that any item of information is disclosed in the Disclosure Schedules shall not be construed to mean that such information is required to be disclosed by the Agreement. Any information and the dollar thresholds set forth herein shall not be used as a basis for interpreting the term "material" or other similar terms in the Agreement or constitute an admission that such items are required to be disclosed under the Agreement.

Exhibit A Facility

Kindred Transitional Care and Rehabilitation – Birchwood Terrace Facility No. 559 43 Starr Farm Rd Burlington, VT 05408-1321

Schedule 2.1(e) Assumed Contracts

All Contracts listed on Schedule 4.12(a), subject to change until fifteen (15) days prior to Closing at Transferee's discretion per Section 2.1(e) of the Agreement.

Schedule 2.5(a) Ineligible Employees

Schedule 4.2 Notices

- (a) None.
- (b) See below.

DPNA

None.

CMPs

None.

Survey Deficiencies

See attached.

Survey Deficiencies

Facility Name	Exit Date	State	QIS	# of Tags	Def Free	SMART Training Completed	8 Months from Last Std Survey	9 Month from Last Std Survey	15 Month from Last Std Survey
Kindred Transitional Care and Rehabilitation- Birchwood Terrace	4/19/2017	VT	Y	0	Y	9/19/2016	12/19/2017	1/19/2018	7/19/2018

The surveys listed below (for which, as of the Execution Date, the Facility has not received certification that it has achieved substantial compliance) may result in the imposition of recommended or mandatory remedies (for, e.g., civil money penalties, denial of payment for new admissions, and termination of Medicare or Medicaid participation) in the future and/or on a retroactive basis and regardless of whether such remedies are specified below:

Facility	Start	Exit	CMS Letter	Date Certain	Survey Type	Result	FTag/KTag Scope/Severity	Proposed Remedies	Imposed Remedies
Kindred Transitional	5/10/	5/10/			Complaint	2567 Pending (Complaint &			
Care and	17	17				Self Rpt) - Verbal exit Both			
Rehabilitation-						Unsubstantiated			
Birchwood Terrace									

Schedule 4.3 Litigation

1. <u>Kindred Transitional Care and Rehabilitation-Birchwood Terrace, Vermont, Facility No. 559</u>: Two OAG investigations (abuse allegations) fall 2016. The self-reported events involved a staff member accused of unwanted physical touching of a resident. Information was provided to the OAG. Message left with OAG to obtain status.

Schedule 4.5(a) **Employee Benefits Plans**

2017 Plans

401(k)

T. Rowe Price

Deferred Compensation - Non Qualified Plan

T. Rowe Price

Kindred Healthcare Operating, Inc. Employee Medical and Welfare Benefits Plan

Medical

Anthem

Aetna

UHC

Prescription

ESI - Carve out

Wellness

Limeade

<u>H.S.A.</u>

Discovery

Commuter/Parking

Discovery

HMO

Kaiser CA Union

Kaiser CA Non-Union (High & Medium) **UHC** of **CA** Union

Advocacy

Health Advocate

Dental

CIGNA

Delta

Vision

Davis

UHC

Basic Life/AD&D

Unum

Supplemental Life - EE, Spouse & Child Benefit

Unum

Voluntary AD&D

Unum

STD & Buy-Up

Unum

LTD & Buy-Up

Unum

Accident

Unum

Critical Illness

Transamerica

Hospital Indemnity

Transamerica

Permanent Life

Unum

Identity Theft

InfoArmor

Legal

Hyatt Legal

Pet Insurance

VPI

Kindred Healthcare Operating, Inc. Cafeteria Plan FSA - General Purpose Healthcare, Limited Purpose Healthcare, Dependent

Care

Discovery

Auto/Home

MetLife

EAP

Unum

Discount Mall

Perkspot

Corporate Bonus Plans

Short Term Incentive Comp Plan Long Term Cash Incentive Plan Kindred Stock Incentive Plan

Division Specific Bonus Plans

AVPS Incentive Plan
ADS Incentive Plan
CL Incentive Plan
Support Center Insurance Specialist Bonus Plan
Business Office Bonus Plan
Every Moment Counts

Additional Fringe Benefits

Paid Time Off/Holidays
Tuition Reimbursment
ED/DNS - Take a Break on Us Housekeeping & Lawn Service

Retention Agreements

See attached chart.

Schedule 4.6(a) Employees

See attached.

Schedule 4.6(b) Employee Compensation

Schedule 4.6(e) Labor and Employment Disputes

Schedule 4.6(f) Collective Bargaining Agreements

Schedule 4.6(g) Labor Disputes

Schedule 4.6(i) At-Will Employment Exceptions

See disclosures set forth on Schedule 4.6(a).

Schedule 4.7(b) Requests for Repairs, Restorations or Alterations

Schedule 4.7(c) Encumbrances

1. JP Morgan Chase Bank, N.A., as a collateral agent, has a priority security interest in all Kindred assets.

Schedule 4.8(a)(i) Government Reimbursement Programs

(a) See Medicare and Medicaid numbers in the attached chart and the VA contract below:

MNumber	Contract Profile	Payor Parent Name	Payor Name	Contract Start Date
M0003665	M0003665.80002.VAMas_25.VA Birchwood	VETERANS ADMINISTRATION	VA Massachusetts	3/1/2013

(b) None.

							Kind	red Nursing	Facilities -	- Vermont					
									Li	icense					
															Initial
	Medicare Medicaid State		Facility Address NPI #		Licensee S	State License			n Licensed Medicar		CLIA Cert No.	Comments	Licensure		
rype		31	20mm	Facility Name	racinty Address	Mr1#	Licensee	#					4	Comments	
	Provider #	Provider#							Date	Date	Beds	Cert. Date			Date
				Kindred Transitional Care and	43 Starr Farm Rd,		Kindred							10 beds delicensed	
	1		11 8	Rehabilitation-Birchwood	Burlington, VT		Nursing Centers)	effective 4/1/2010; 6 beds	
	47-5003	47-5003	VT	Тептасе	05408	1659481943	East LLC	27-0000393	02/01/17	01/31/18	144	02/26/69	47D0670331	delicensed eff 4/1/11	
SNF			1												

$Schedule~4.8(a) (iv)\\ Threatened~Recoupment,~Suspensions,~Terminations~or~Restrictions$

See disclosures set forth on Schedule 4.2(b).

Schedule 4.8(a)(v) Audits

- (a) None.
- (b) and (c) see charts below.

MEDICAID COST REPORTS										
FYE	2011	2012	2013	2014	2015	2016	2017	NOTES		
Vermont	Completed	Completed	Completed	Completed	Open	Open	Open	Cost report audit occurs each year, no specific timeframe to begin. Nursing Care component rebases every two years; Resident Care, Indirect, and DON components rebase every 4 years; Ancillary & Property components rebase the next quarterly rate after the conclusion of an audit.		

MEDICARE COST REPORTS									
Medicare Provider Number	Fac	State	Name	Ending Date	Date of NPR				
Not Applicable	Not Applicable	Not Applicable	Not Applicable	Not Applicable	Not Applicable				

(d) See disclosures set forth on Schedule 4.2(b).

Schedule 4.8(b)(i) Licenses

See License disclosures set forth on the chart attached to Schedule 4.8(a)(i).

Schedule 4.8(b)(ii) Number and Type of Beds

See Licensed Beds attached.

ID#/FACILITY	Licensed Beds	Medicare Beds as of 1/1/17	Increases 1/1/17- 6/30/17	Decreases 1/1/17- 6/30/17	Medicare Beds	Dual Certification or Exception Detail	Effective Date or Certification Change	Beds (included in total)
559 KTC&R-Birchwood	144	144			144	100% DUAL	04/01/2011	144

Schedule 4.8(b)(iii) Reduction in Number of Beds

Schedule 4.8(b)(iv) Violations

See Survey Deficiencies contained in the attachment to Schedule 4.2(b).

Schedule 4.8(b)(v) Life Safety Code

See Survey Deficiencies contained in the attachment to Schedule 4.2(b).

Schedule 4.8(c)(i) Compliance

See disclosures as set forth on Schedule 4.2(b).

Schedule 4.8(c)(ii) Referral Laws

Schedule 4.8(d) Convictions/Exclusions

Schedule 4.8(e)-1 Audits; Settlements

See disclosures as set forth on Schedule 4.2(b) and Schedule 4.8(a)(iv).

Schedule 4.8(e)-2 Special Focus Facility

Schedule 4.10 Absence of Changes

Schedule 4.12(a) Proposed Assumed Contracts

See attached.

Fac. No.	Kindred Facility	State	Kindred Party	Vendor	Category	Contract type	Additional Contact Persons	Comments
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		TLC Nursing Associates LLC	Clinical	Staffing		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Ecolab	Non- Clinical	Dishmachine Lease		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Alliance Mechanical	Non- Clinical	Proactive Maintenance Agreement		
559	Kindred Transitional Care and Rehabilitation-	VT		Bayada Nurses, Inc. dba Bayada Hospice	Clinical	Inpatient Services Agreement	Kelli A. Marans, Esq.	
	Birchwood Terrace						Bayada Nurses, Inc.	
							101 Executive Dr. Ste 8	
							Moorestown, NJ 08057	
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Brook Field Service	Non- Clinical	Prevenative Maintenace Agreement		For contractual year 2013; silent on renewal terms
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Comcast	Non- Clinical	Services Agreement (cable)		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Excel Property Maintenance	Non- Clinical	Landscaping Services Agreement		For 2016 season
559	Kindred Transitional Care and	VT		Farrell Vending	Non- Clinical	Vending Services Agreement		

Fac. No.	Kindred Facility	State	Kindred Party	Vendor	Category	Contract type	Additional Contact Persons	Comments
	Rehabilitation- Birchwood Terrace					0.0		_ :
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Fire Tech	Non- Clinical	Agreement for Sprinkler Inspection Services		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Fletcher Allen Health Care, Inc.	MD	Medical Director Services Agreement		Dr. Zail S. Berry, MD
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Fletcher Allen Health Care, Inc.	Clinical	Professional Services Agreement - Psychiatry		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Fletcher Allen Health Care	Clinical	Laboratory and X-Ray Services Agreement		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Gauthier Trucking Company, Inc.	Non- Clinical	Solid waste disposal		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Haun Specialty Gases, Inc	Clinical	Agreement for Oxygen Supplies		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Healthcare Service Group	Non- Clinical	Certificate of Insurance		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Boylan, Joseph	Non- Clinical	Snowplowing		
559	Kindred Transitional Care and Rehabilitation-	VT		Med Waste Disposal	Non- Clinical	Agreement for Medical Waste Disposal		Expired

Fac. No.	Kindred Facility	State	Kindred Party	Vendor	Category	Contract type	Additional Contact Persons	Comments
	Birchwood Terrace							
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Millers Pest Control	Non- Clinical	Pest Control Service Agreement		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Mother Nature's Helper	Non- Clinical	Agreement for Plant Maintenance Services		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Phillips, Melinda	Non- Clinical	NCD Beauty/Barber Services Agreement		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		Simplex Grinnell	Non- Clinical	Fire alarm/sprinkler services		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		SSTA	Non- Clinical	Price Chart - hourly rate to a seat charge		
559	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	VT		UVM - Technical Services Program	Non- Clinical	Preventative Mainenance		

Schedule 4.17(a) Financial Statements

Schedule 4.18 Transferor Broker

- 1. Guggenheim Securities, LLC.
- 2. Healthcare Transactions Group, Inc.

Schedule 5.3 Transferee Broker

Schedule 6.3 Non-Competition Covenant

(a) The following facilities, until Seller relinquishes ownership of such facilities and the applicable Transferor transfers operations of such facilities:

<u>State</u>	Facility Name	Facility Address
AZ	Kindred Nursing and Rehabilitation-Hacienda	660 South Coronado Drive, Sierra Vista, AZ 85635
AZ	Kindred Transitional Care - Phoenix	1880 East Van Buren St, Phoenix, AZ 85006-3742
CA	Kindred Nursing and Healthcare - Livermore	76 Fenton Street, Livermore, CA 94550
CA	Kindred Nursing and Healthcare - Victorian	2121 Pine St, San Francisco, CA 94115
CA	Kindred Nursing and Healthcare-Bayberry	1800 Adobe Street, Concord, CA 94520-2313
CA	Kindred Nursing and Rehabilitation - Medical Hill	475 29th Street, Oakland, CA 94609
CA	Kindred Nursing and Rehabilitation - Ygnacio Valley	1449 Yganacio Valley Rd, Walnut Creek, CA 94959
CA	Kindred Nursing and Rehabilitation-Golden Gate	2707 Pine Street, San Francisco, CA 94115
CA	Kindred Nursing and Rehabilitation-Nineteenth Avenue	2043 19th Ave, San Francisco, CA 94116
CA	Kindred Nursing and Transitional Care - Pacific Coast	720 E. Romie Lane, Salinas, CA 93901
CA	Kindred Nursing and Transitional Care - Santa Cruz	1115 Capital Rd, Santa Cruz, CA 95062
CA	Kindred Nursing and Transitional Care-South Marin	1220 S. Eliseo Dr, Greenbrea, CA 94904
CA	Kindred Transitional Care and Rehabilitation - Valley Gardens	1517 Knickerbocker Drive, Stockton, CA 95210
CA	Kindred Transitional Care and Rehabilitation - Walnut Creek	1224 Rossmoor Parkway, Walnut Creek, CA 94595
CA	Kindred Transitional Care and Rehabilitation- Canyonwood	2120 Benton Drive, Redding, CA 96003-2151
CA	Kindred Transitional Care and Rehabilitation- Foothill	401 W Ada Ave, Glendora, CA 91741
CA	Kindred Transitional Care and Rehabilitation- Lawton	1575 7th Ave, San Francisco, CA 94122

State	Facility Name	Facility Address
CA	Kindred Transitional Care and Rehabilitation-Siena	11600 Education Street, Auburn, CA 95602
CA	Kindred Transitional Care and Rehabilitation-Smith Ranch	1550 Silveria Pkwy, San Rafael, CA 94903
CA	Kindred Transitional Care and Rehabilitation- Tunnell Center	1359 Pine St, San Francisco, CA 94109
CO	Kindred Nursing and Rehabilitation - Aurora	10201 East Third Ave, Aurora, CO
GA	Kindred Transitional Care and Rehabilitation- Lafayette	110 Brandywine Blvd, Fayetteville, GA 30214
ID	Kindred Nursing and Rehabilitation - Aspen Park	420 Rowe St, Moscow, ID 83843
ID	Kindred Nursing and Rehabilitation - Caldwell	210 Cleveland Blvd, Caldwell, ID 83605
ID	Kindred Nursing and Rehabilitation - Canyon West	2814 S. Indiana Ave, Caldwell, ID 83605
ID	Kindred Nursing and Rehabilitation - Mountain Valley	601 W. Camerson Ave, Kellogg, ID 83837
ID	Kindred Nursing and Rehabilitation - Nampa	404 N. Horton St, Nampa, ID 83651
ID	Kindred Nursing and Rehabilitation - Weiser	331 E. Park St, Weiser, ID 83672
ID	Kindred Transitional Care and Rehabilitation- Lewiston	3315 8th St, Lewiston, ID 83501
IN	Kindred Nursing and Rehabilitation-Valley View	333 W. Mishawaka Rd, Elkhart, IN
IN	Kindred Transitional Care and Rehabilitation-Allison Pointe	5226 E 82nd St, Indianapolis, IN
IN	Kindred Transitional Care and Rehabilitation- Bridgewater	14751 Carey Road, Carmel IN
IN	Kindred Transitional Care and Rehabilitation- Columbus	2100 Midway St, Columbus, IN
IN	Kindred Transitional Care and Rehabilitation-Dyer	2300 Great Lakes Dr, Dyer, IN
IN	Kindred Transitional Care and Rehabilitation-Eagle Creek	4102 Shore Dr, Indianapolis, IN
IN	Kindred Transitional Care and Rehabilitation- Greenfield	200 Green Meadows Dr, Greenfield, IN
IN	Kindred Transitional Care and Rehabilitation- Greenwood	377 Westridge Blvd, Greenwood, IN

State	Facility Name	Facility Address
IN	Kindred Transitional Care and Rehabilitation- Harrison	150 Beechmont Dr, Corydon, IN
IN	Kindred Transitional Care and Rehabilitation-Indian Creek	240 Beechmont Dr, Corydon, IN
IN	Kindred Transitional Care and Rehabilitation- Kokomo	429 Lincoln Rd, Kokomo, IN
IN	Kindred Transitional Care and Rehabilitation- Rolling Hills	3625 St Joseph Rd, New Albany, IN
IN	Kindred Transitional Care and Rehabilitation- Sellersburg	7823 Old Hwy #60, Sellersburg, IN
IN	Kindred Transitional Care and Rehabilitation- SouthPointe	4904 War Admiral Drive, Indianapolis, IN
IN	Kindred Transitional Care and Rehabilitation- Southwood	2222 Margaret Ave, Terre Haute, IN
IN	Kindred Transitional Care and Rehabilitation- Wedgewood	101 Potters Lane, Clarksville, IN
IN	Kindred Transitional Care and Rehabilitation- Wildwood	7301 E. 16th Street, Indianapolis, IN
KY	Heritage Manor Healthcare Center	401 Indiana Ave, Mayfield, KY 42066
KY	Kindred Nursing and Rehabilitation-Maple	515 Greene Drive, Greenville, KY 42345
MA	Clark House Nursing Center at Fox Hill Village	30 Longwood Drive, Westwood, MA 02090
MA	Kindred Nursing and Rehabilitation-Braintree	1102 Washington Street, Braintree, MA 02184
MA	Kindred Nursing and Rehabilitation-Harborlights	804 East 7th Street, Boston, MA 02127
MA	Kindred Nursing and Rehabilitation-Laurel Lake	620 Laurel Street, Lee, MA 01238
MA	Kindred Nursing and Rehabilitation-Tower Hill	One Meadowbrook Way, Canton, MA 02021
MA	Kindred Transitional Care and Rehabilitation-Avery	100 West Street, Needham, MA 02194
MA	Kindred Transitional Care and Rehabilitation- Country Estates	1200 Suffield Street, Agawam, MA 01001
MA	Kindred Transitional Care and Rehabilitation-Eagle Pond	1 Love Lane, South Dennis, MA 02660
MA	Kindred Transitional Care and Rehabilitation-	50 Indian Neck Road, Wareham, MA 02571

State	Facility Name	Facility Address
	Forestview	
MA	Kindred Transitional Care and Rehabilitation- Harrington	160 Main Street, Walpole, MA 02081
MA	Kindred Transitional Care and Rehabilitation- Highgate	10 CareMatrix Drive, Dedham, MA 02026
MA	Kindred Transitional Care and Rehabilitation- Highlander	1748 Highland Ave, Fall River, MA 02720
MA	Kindred Transitional Care and Rehabilitation- Westborough	8 Colonial Drive, Westborough, MA 01581
MA	Ledgewood Rehabilitation and Skilled Nursing Center	87 Herrick Street, Beverly, MA 01915
MA	Seacoast Nursing and Rehabilitation Center	292 Washington Street, Gloucester, MA 01930
МТ	Kindred Nursing and Rehabilitation - Parkview	200 North Oregon St, Dillon, MT
MT	Kindred Transitional Care and Rehabilitation-Park Place	1500 32nd St S, Great Falls, MT
NC	Kindred Nursing and Rehabilitation-Henderson	280 S Beckford Dr, Henderson, NC 27536
NC	Kindred Transitional Care and Rehabilitation- Elizabeth City	901 S. Halstead Blvd., Elizabeth City, NC 27909
NC	Kindred Transitional Care and Rehabilitation-Rose Manor	280 S Beckford Dr, Henderson, NC 27536
NH	Kindred Transitional Care and Rehabilitation- Greenbriar	55 Harris Road, Nashua, NH 03062
ОН	Kindred Nursing and Rehabilitation-Community	175 Community Drive, Marion, OH 43302
ОН	Kindred Transitional Care and Rehabilitation- LakeMed	70 Normandy Drive, Painesville, OH 44077
ОН	Kindred Transitional Care and Rehabilitation- Newark	75 McMillen Drive, Newark, OH 43055
ОН	Kindred Transitional Care and Rehabilitation- Stratford	7000 Cochran Rd, Glenwillow, OH 44139
ОН	Kindred Transitional Care and Rehabilitation-The Greens	1575 Brainard Road, Lyndhurst, OH 44124
TN	Kindred Nursing and Rehabilitation - Fairpark	307 North Fifth Street, Maryville, TN

State	Facility Name	Facility Address
TN	Kindred Nursing and Rehabilitation - Loudon	1520 Grove St, Loudon, TN
TN	Kindred Nursing and Rehabilitation - Northhaven	3300 North Broadway, Knoxille, TN
TN	Kindred Nursing and Rehabilitation - Smith County	112 Health Care Drive Carthage, TN
TN	Kindred Transitional Care and Rehabilitation- Maryville	1012 Jamestown Way, Maryville, TN
TX	Kindred Transitional Care and Rehabilitation- Grapevine	1005 IRA E. Woods Parkway, Grapevine, TX 76051
TX	Kindred Transitional Care and Rehabilitation- Mansfield Plaza	301 N. Miller Road, Mansfield, TX 76063
TX	Kindred Transitional Care and Rehabilitation- Ridgmar	6600 Lands End Court, Fort Worth, TX 76116
VA	Kindred Nursing and Rehabilitation-River Pointe	4142 Bonney Rd, Virginia Beach, VA 23452
VA	Kindred Transitional Care and Rehabilitation-Bay Pointe	1148 First Colonial Rd, VA Beach, VA 23454
VA	Kindred Transitional Care and Rehabilitation- Nansemond Pointe	200 West Constance Road, Suffolk, VA 23434
VA	Kindred Assisted Living – Nansemond Commons	200 West Constance Road, Suffolk, VA 23434
VT	Kindred Transitional Care and Rehabilitation- Birchwood Terrace	43 Starr Farm Rd, Burlington, VT 05408
VT	Starr Farm Nursing Center	98 Starr Farm Rd, Burlington, VT 05408
WA	Kindred Nursing and Rehabilitation-Arden	16357 Aurora Ave North, Seattle, WA 98133
WA	Kindred Transitional Care and Rehabilitation- Lakewood	11411 Bridgeport Way SW, Tacoma, WA 98499
WA	Kindred Transitional Care and Rehabilitation- Vancouver	400 East 33rd Street, Vancouver, WA 98663

(b) The following facilities to be retained by Seller and/or its Affiliates:

State	Facility Name	Facility Address
AZ	Kindred N&R Hacienda	660 S Coronado Dr, Sierra Vista, AZ 85635
CA	KH Brea SAU	875 N Brea Blvd, Brea, CA 92821

FL	KH South Florida Hollywood SAU	1859 Van Buren St, Hollywood, FL 33020
KY	KH Louisville SAU	1313 Saint Anthony Pl, Louisville, KY 40204
NC	KH Greensboro SAU	2401 Southside Blvd, Greensboro, NC 27406
NV	Kindred TC&R Spring Valley	5650 S Rainbow Blvd, Las Vegas NV 89118
NV	Kindred TC&R at KH Las Vegas Flamingo	2250 E Flamingo Rd, Las Vegas, NV 89119
TX	KH Dallas SAU	9525 Greenville Ave, Dallas, TX 75243
WA	KH Seattle First Hill SAU	1334 Terry Ave, Seattle, WA 98101
WA	KH Seattle Northgate SAU	10631 8 th Ave NE, Seattle, WA 98125

Schedule 6.10(e) Pending Medicaid Applicants

[To come at closing.]

Schedule 6.10(f)(i) Accounts Receivables by Resident

[See attached.] [To come at closing]

Schedule 7.2(h) UPL Program Transfers

ATTACHMENT 33

AMENDED AND RESTATED ASSIGNMENT AND ASSUMPTION AGREEMENT

VERMONT

THIS AMENDED AND RESTATED ASSIGNMENT AND ASSUMPTION AGREEMENT (this "Assignment") is entered into as of August 1, 2017 and effective as of June 30, 2017 (the "Assignment Effective Date"), by and among BM Eagle Holdings, LLC ("Assignor"), Starr Farm Operations LLC ("Starr Farm Assignee") and BIRCHWOOD PROP LLC ("Birchwood Assignee" and together with Starr Farm Assignee, "Assignees"). Capitalized terms used but not defined herein shall have the same meanings ascribed to such terms in the Purchase Agreement, as defined below.

RECITALS

WHEREAS, pursuant to the terms of that certain Asset Purchase Agreement (the "<u>Purchase Agreement</u>"), dated as of the Assignment Effective Date, by and between Assignor and Kindred Healthcare Operating, Inc. ("<u>Kindred</u>"), as Purchaser and Seller thereunder respectively, Purchaser agreed to purchase from Seller, among other things, the portfolio of skilled nursing facilities set forth on <u>Exhibit A</u> hereto (the "<u>Assigned Facilities</u>"); and

WHEREAS, Assignor and Birchwood Assignee previously entered into that certain Assignment and Assumption Agreement, dated June 30, 2017, pursuant to which Assignor assigned its interest under the Purchase Agreement in the Assigned Facilities to Birchwood Assignee; and

WHEREAS, the principals of Birchwood Assignee and Starr Farm Assignee now desire to amend and restate such Assignment and Assumption Agreement in its entirety as set forth herein so that Assignor's interest under the Purchase Agreement in the Assigned Facility known as Birchwood Terrace will be assigned to the Birchwood Assignee and Assignor's interest under the Purchase Agreement in the Assigned Facility known as the Starr Farm Nursing Center will be assigned to the Starr Farm Assignee, as set forth on Exhibit A hereto.

WHEREAS, the parties hereto desire to execute this Assignment to set forth the terms and conditions under which Assignor shall assign to Assignees, subject to the terms and conditions hereof, all of its right, title, and interest in and to the Purchase Agreement, solely with respect to the Assigned Facilities, including all real estate related thereto and as further described in the Purchase Agreement (collectively, the "Assigned Assets").

NOW THEREFORE, in consideration of the mutual premises contained herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

Assignment and Assumption.

(a) Provided that each Assignee delivers and pays the Assignment Purchase Price in accordance with Section 2(a) and subject to the terms of Section 1(c), Assignor hereby assigns, transfers and conveys to Birchwood Assignee, all of Assignor's rights under the Purchase

Agreement with respect to the Assigned Assets of the Assigned Facility known as Birchwood Terrace, and Assignor hereby assigns, transfers and conveys to Starr Farm Assignee, all of Assignor's rights under the Purchase Agreement with respect to the Assigned Assets of the Assigned Facility known as the Starr Farm Nursing Center, each as set forth on Exhibit A hereto.

- (b) Subject to the terms of <u>Section 1(c)</u>, each Assignee hereby accepts the assignment of Assignor's rights under the Purchase Agreement with respect to the Assigned Assets and assumes and agrees to fulfill, carry out and discharge all of Assignor's obligations, liabilities, duties, and terms and conditions of and applicable to Assignor's rights under the Purchase Agreement with respect to the Assigned Assets.
- (c) Notwithstanding any provision herein to the contrary, this Assignment shall not transfer, and Assignor shall retain, any and all of Assignor's rights, obligations or liabilities under Sections 2.1, 2.5(a)(i), 2.5(a)(x), 2.5(b), 2.5(c) (solely with respect to the first sentence), 3.4, 5.6, 6.9(e), 6.9(f), 10.1(a), 10.1(b), 10.1(c)(i), 10.1(d), 10.1(e), 10.1(f), 10.1(h), 10.2(b)(i), and each of 10.4(a), 10.4(b) and 10.4(c) (solely as each relates to the Termination Payment and the surrender of the Deposit), of the Purchase Agreement; provided, however, that upon the payment of the Assignment Purchase Price with respect to an Assigned Facility in accordance with this Assignment, the parties shall execute and deliver an assignment and assumption in the form attached hereto as Exhibit B (the "Assignment of Certain Retained Rights") pursuant to which Assignor shall assign to the applicable Assignee all rights, and such Assignee shall assume all obligations of Assignor, with respect to the Assigned Assets under Sections 2.1 and 3.4 of the Purchase Agreement with respect to such Assigned Facility. For the avoidance of doubt, Assignor shall remain obligated under the Purchase Agreement for the payment at the Closing of the portion of the Purchase Price allocated to the Assigned Facilities.

Purchase Price; Deposit.

In consideration of the assignment of the Assigned Assets, Assignees shall deposit or shall cause their affiliate(s) to deposit with Madison Title Company, LLC (the "Escrow Holder") an aggregate cash amount equal to \$8,500,000 (the "Assignment Purchase Price") prior to the Closing or first of two Closings (as applicable) with respect to the Assigned Assets, each of which shall occur in accordance with and subject to the terms and conditions of the Purchase Agreement. Not less than 5:00 PM EST one (1) Business Day prior to the each Closing with respect to the Assigned Assets (a "Payment Deadline"), Assignees will deliver to the Escrow Holder, by wire transfer of immediately available funds, the portion of the Assignment Purchase Price allocated to the Assigned Facilities being acquired by each Assignee at such Closing, less the amount of the Assignee Deposit being credited to the Assignment Purchase Price at such Closing pursuant to this Assignment, to be held in escrow in accordance with an escrow agreement substantially in the form attached hereto as Exhibit C (the "Purchase Price Escrow Agreement"). Time shall be of the essence with respect to Assignees' obligation to deliver and pay the Assignment Purchase Price. In the event that Assignees fail to make a payment of the Assignment Purchase Price by the applicable Payment Deadline, this Assignment shall automatically and immediately terminate with respect to all Assigned Facilities that are not vet Transferred Facilities, and Assignor shall be entitled to retain the Assignee Deposit (as defined below) as liquidated damages, but shall not have the right to sue for damages or pursue specific performance.

- (b) On or prior to the Assignment Effective Date, Assignees have made one or more deposits in an amount equal to \$425,000 (collectively, the "Assignee Deposit") in an escrow account with the Escrow Holder to be held pursuant to this Assignment and the Escrow Agreement dated May 30, 2017 (the "Deposit Escrow Agreement"). The Assignee Deposit shall serve as a portion of the Assignment Purchase Price or be paid in accordance with this Assignment and the Deposit Escrow Agreement, and to the extent applied to the Assignment Purchase Price, shall be applied at the final Closing with respect to the Assigned Facilities.
- 3. <u>Default</u>. In the event that Assignor fails to execute and deliver the Assignment of Certain Retained Rights in violation of its obligations hereunder, Assignees, in addition to all other rights and remedies, shall be entitled to specific performance.

4. Representations, Warranties and Covenants.

- (a) Assignor hereby covenants, warrants, and represents to Assignees that all of the representations and warranties of Assignor in the Purchase Agreement are, without condition or exception, incorporated and restated herein by Assignor as of the Assignment Effective Date as though fully stated in this Assignment, and such representations and warranties are true and correct in all material respects.
- (b) Each Assignee hereby represents and warrants to Assignor that all representations and warranties made by Assignor in the Purchase Agreement (specifically excluding the representations and warranties under Section 5.6 of the Purchase Agreement) are true and correct with respect to such Assignee to the same extent as if such representations and warranties were made by such Assignee as of the Assignment Effective Date; provided, however, that the representations and warranties set forth in Section 5.1(a) of the Purchase Agreement with respect to organizational matters shall be deemed made with respect to the entity type of such Assignee, if different from that of Assignor.
- (c) Each party hereby represents and warrants to the other party that this Assignment has been duly authorized by all necessary corporate or company action of such party and is enforceable in accordance with its respective terms with respect to such party, except as the enforceability may be subject to or limited by: (i) bankruptcy, insolvency, reorganization, arrangement, moratorium, fraudulent or preferential conveyance, or similar laws affecting the rights of creditors generally, or (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (d) Assignor represents and warrants to Assignees the following with respect to the Assigned Assets: (i) a true and complete copy of the Purchase Agreement as of the Assignment Effective Date, including all schedules and exhibits thereto, is attached hereto as Exhibit D; (ii) the Purchase Agreement is a valid, binding and enforceable obligation of Assignor, enforceable against Assignor in accordance with its terms, and is in full force and effect, except as the enforceability may be subject to or limited by: (1) bankruptcy, insolvency, reorganization, arrangement, moratorium, fraudulent or preferential conveyance, or similar laws affecting the rights of creditors generally, or (2) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law); (iii) neither Assignor nor, to Assignor's knowledge, any other party thereto is in default under or in violation of (or, to

Assignor's knowledge, is alleged to be in default under or in violation of), or has provided or received any written or, to Assignor's knowledge, oral notice of any intention to terminate, the Purchase Agreement; (iv) Assignor has not received any written or, to Assignor's knowledge, oral notice of any disputes under the Purchase Agreement; and (v) to Assignor's knowledge, there are no disputes pending or, threatened, under the Purchase Agreement.

- (e) Assignor represents and warrants to Assignees that the consideration received by Assignees pursuant to this Assignment, is full, adequate and independent consideration for the covenants and representations made by Assignees.
- (f) Except as set forth in this Assignment and the Purchase Agreement and the OTA for each Assigned Asset, each Assignee acknowledges, represents and warrants to Assignor that neither Assignor nor any other Person has made any representation or warranty, expressed or implied, as to the accuracy or completeness of any information regarding Kindred, the Assigned Assets or the Assigned Facilities furnished or made available to Assignees and their respective representatives. Each Assignee acknowledges that, except for the representations and warranties contained in the Purchase Agreement (and the OTA for each Assigned Asset) and this Assignment, neither Assignor nor any other Person has made, and such Assignee has not relied on, any other express or implied representation or warranty by or on behalf of the Assignor.
- (g) Prior to the Closing, Assignor shall comply with the reasonable instructions of Assignees, solely with respect to the Assigned Assets, as necessary for the Assignees to exercise the rights assigned to them under this Assignment.
- Each Assignee shall comply with the requirements of Sections 10.4(a) and 10.4(b) (h) of the Purchase Agreement, with respect to the Assigned Assets, to secure Substitute OTA Transferees in accordance with the timelines set forth therein and take such other actions as may be necessary or reasonably requested by Assignor from time to time to enable Assigned Facilities subject to either Section 10.4(a) or Section 10.4(b) to reach a Closing under the Purchase Agreement. Assignees shall not appoint a Substitute OTA Transferee without the prior written consent of Assignor (not to be unreasonably withheld). In the event Assignor determines in its reasonable discretion that an OTA Transferee or Substitute OTA Transferee selected by an Assignee is unable to fulfill its obligations under the OTA, upon written notice by Assignor to such Assignee of such determination, such Assignee shall replace such OTA Transferee or Substitute OTA Transferee in accordance with terms of this Assignment and the Purchase Agreement. In the event of a termination by Seller under Sections 10.4(a), 10.4(b) or 10.4(c) of the Purchase Agreement with respect to any Assigned Facility, this Assignment shall automatically terminate with respect to such Assigned Facility and the Assignor shall be entitled to retain the Assignee Deposit as liquidated damages, but shall not have the right to sue for damages or pursue specific performance.
- (i) Within ten (10) Business Days of the date on which an OTA is executed with respect to an Assigned Facility, each Assignee shall enter into a binding agreement with the applicable OTA Transferee, under which such Assignee shall have the right to cause such OTA Transferee to assign the OTA in the event the Assigned Facility has become an Operator Delay Facility or such OTA Transferee has otherwise materially breached its obligations under the

OTA in a manner which would reasonably be expected to cause the failure of any Closing Condition under the Purchase Agreement or the OTA.

(j) Each Assignee shall execute a joinder to the Escrow Agreement with respect to the Assigned Assets effective on the applicable Closing Date.

5. Indemnification.

- (a) Any and all indemnification claims by and between Assignees, on the one hand, and Kindred, on the other hand, arising in connection with the Assigned Facilities or the Assigned Assets shall be governed under the applicable indemnification provisions of the Purchase Agreement. Except as set forth in Section 5(b) of this Assignment, it is expressly understood and agreed that Assignees shall have no right to seek indemnification from Assignor for any matter whatsoever arising in connection with Assignees' purchase of the Assigned Assets or the Assigned Facilities.
- (b) Assignor shall indemnify, defend and hold Assignees harmless from and against any and all Losses arising out of or relating to any breach by Assignor of any representation, warranty, covenant or agreement made by Assignor in this Assignment.
- (c) Assignees shall jointly and severally indemnify, defend and hold Assignor harmless from and against any and all Losses arising out of or relating to any breach by either Assignee of any representation, warranty, covenant or agreement made by such Assignee in this Assignment.

6. Termination.

- (a) This Assignment may be terminated with respect to an Assigned Facility prior to the applicable Closing for such Assigned Facility (i) at any time by mutual consent of the Assignor and the applicable Assignee; (ii) automatically with respect to any Assigned Facility that is not a Transferred Facility, without any further action by the parties hereto, in the event of a termination of the Purchase Agreement with respect to such Assigned Facility pursuant to Sections 6.9 or 10.1 thereof; (iii) by the Assignor, by reason of the material breach by the applicable Assignee of its representations, warranties, covenants or other obligations set forth in this Assignment or the Purchase Agreement; and (iv) by the applicable Assignee, by reason of the material breach by Assignor of any representations, warranties, covenants or other obligations set forth in this Assignment.
- (b) In the event of a termination of this Assignment pursuant to Section 6.1(a)(i) or 6.1(a)(ii), the Assignee Deposit shall be returned to the applicable Assignee. Assignor agrees to promptly provide Escrow Holder with written instructions in the form required by Escrow Holder in connection with a return of the Assignee Deposit to Assignees.
- (c) In the event of a material breach that gives rise to a right of Assignor to terminate the Assignment pursuant to Section 6.1(a)(iii), Assignor shall have the right as its sole and exclusive remedy to retain the Assignee Deposit as liquidated damages and terminate this Assignment. Assignor shall not have the right to sue for damages or pursue specific performance.

(d) In the event of an uncured material breach that gives rise to a right of the applicable Assignee to terminate the Assignment pursuant to Section 6.1(a)(iv), such Assignee shall have the right to (i) recover the Assignee Deposit and terminate the Assignment and sue for damages (in which case Assignor agrees to promptly provide Escrow Holder with written instructions sufficient to cause Escrow Holder to release the Assignee Deposit to such Assignee) or (ii) pursue specific performance solely to require the consummation of the transactions contemplated under this Assignment (but not to obtain money for losses and the Assignee Deposit shall be left in escrow pending the outcome of any action for specific performance).

7. Miscellaneous.

- (a) <u>Delivery of Notices</u>. Assignor shall, within two (2) business days of receipt, deliver to Assignees a copy of any notices received by Assignor under Section 11.4 of the Purchase Agreement (and expressly including any Disclosure Updates delivered pursuant to Section 4.17 of the Purchase Agreement), or delivered by Assignor to Kindred under the Purchase Agreement, to extent related to, or concerning, the Assigned Facilities or the Assigned Assets. Without limiting the generality of the foregoing, Assignor shall, promptly following Assignor's learning of any default with respect to the Assigned Assets, notify Assignees of the existence of any such default by or on the part of Kindred and/or Assignor under the Purchase Agreement.
- (b) <u>Binding Effect</u>. All the covenants and agreements contained in this Assignment shall extend to and be binding upon the heirs, executors, administrators, successors and assigns of the parties hereto.
- (c) <u>Incorporation of Recitals</u>. The Recitals set forth at the beginning of this Assignment are incorporated in and made a part of this Assignment by this reference.
- Assumption Agreement, dated June 30, 2017, by and between Assignor and Birchwood Assignee and all prior understandings between the parties relating to the subject matter hereof. This Assignment may not be further amended or changed without the written consent of the parties hereto. Assignor agrees that it shall not enter into any amendment, restatement, or other modification to the Purchase Agreement that materially and negatively affects the Assigned Facilities or Assignees' obligations with respect to any of the Assigned Assets without Assignees' prior written consent.
- (e) <u>Governing Law</u>. This Assignment shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to the conflicts of law principles thereof.
- (f) <u>Disputes.</u> Any claims, controversies or disputes arising out of or related to this Assignment shall be governed by the terms and conditions set forth in the Purchase Agreement.
- (g) <u>Failure or Delay Not Waiver; Remedies Cumulative</u>. No failure or delay on the part of any party in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty, covenant or agreement herein, nor shall any single or partial exercise of any such right preclude other or

further exercise thereof or of any other right. Unless otherwise provided in this Assignment, all rights and remedies existing under this Assignment are cumulative to, and not exclusive of, any rights or remedies otherwise available.

- (h) No Assignment. No party shall assign its rights and obligations to this Assignment without the written consent of the parties hereto. Notwithstanding the foregoing, any written consent requested in connection with a party's assignment to one or more of its Affiliates shall not be unreasonably withheld. Assignor acknowledges and agrees that Assignees may assign the Assignment to certain of their respective Affiliates as specified in Exhibit 7(h) provided that such Assignee and its Affiliates shall remain jointly and severally liable for all obligations of such Assignee as set forth in this Assignment related to the Assigned Assets.
- (i) <u>Further Assurances</u>. The parties agree that at any time and from time to time, upon the written request of another party, each party will execute and deliver any and all further instruments and documents necessary or desirable to effectuate the transactions contemplated herein, or take any other actions as reasonably necessary to effectuate the foregoing.
- (j) Announcements. If reasonably requested by Assignees, Assignor agrees to provide Assignees with the opportunity to review and comment prior to the issuing of any initial press release pursuant to Section 6.4(a) of the Purchase Agreement, to the extent concerning this Assignment and the transactions contemplated hereby.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have executed this Assignment, which may be executed in multiple counterparts (including electronically-transmitted counterparts), all of which shall constitute one agreement, effective as of the date first above written.

ASSIGNOR

BM EAGLE HOLDINGS, LLC By: BM EAGLE MANAGER, LLC, its Manager

By:

Name: Elliot Mandelbaum Title: Authorized Signatory

ASSIGNEES

BIRCHWOOD PROP LLC

Name: Kri Erlichman

Title:

Authorized Signatory

STARR FARM OPERATIONS LLC

By:

Name: VAri Erlichman

Title:

Authorized Signatory

Exhibit A

Assigned Facilities

<u>To Birchwood Assignee:</u>
Kindred Transitional Care and Rehabilitation-Birchwood Terrace
43 Starr Farm Road
Burlington, VT 05401

To Starr Farm Assignee:
Starr Farm Nursing Center
98 Starr Farm Road
Burlington, VT 05401

Exhibit B

Assignment of Certain Retained Rights

Exhibit C

Form of Purchase Price Escrow Agreement

Exhibit D

Purchase Agreement

[See Attached]

Exhibit 7.1(h)

Assignees

None

ATTACHMENT 34



101 Park Ave Suite 1101 New York, NY 10178

January 22, 2018

Neil Gamss Housing & Healthcare Finance 1979 Marcus Avenue, Suite 210 New Hyde Park, NY 11042

Re: Kindred Transitional Care & Rehab - Birchwood Terrace

Dear Neil:

This term sheet is for discussion purposes only and is issued at a time when Customers Bank the "Lender" has not completed its credit review of the Borrower. All terms of the credit facility will be outlined in a comprehensive loan agreement if the credit facility is approved by Customers Bank. The loan agreement and related documentation may contain additional terms and conditions regarding matters not addressed in this term sheet.

Borrower (Co): Birchwood Prop, LLC, and Birchwood Operations, LLC

Guarantors The unconditional Joint and Several personal guarantees from

Milton Ostreicher and all members of Birchwood Prop, LLC, and

Birchwood OpCo, LLC

Purpose: Acquire the real estate and operations of the Kindred Transitional Care

& Rehab - Birchwood Terrace

Loan Amount: An aggregate principal amount of \$3,752,000 will be available upon

the terms and conditions hereinafter set forth.

\$2,752,000 senior secured Term Loan subject to the lesser of 80% As-

Is LTV or the loan amount

\$1,000,000 Capex Loan subject to an 80% As-Complete Senior LTV

when combined with Facility availability as later detailed.

Line Of Credit

Amount \$1,500,000

Line of Credit: Availability under the Revolver will be subject to a borrowing base

formula, which is equal to the sum of the lesser of (i) \$1,500,000 or ii) up

to 80% on all Eligible Accounts Receivable.

Eligible Accounts receivable would include the following:

· All Medicaid receivables less than 90 days.

All Medicare receivables less than 90 days.

All Insurance receivables less than 90 days.

This term sheet is for discussion purposes only and is issued at a time when Customers Bank has not completed its credit review of the borrower and the collateral to secure the credit facility outlined herein. All terms and conditions are subject to a satisfactory audit by Customers Bank. Furthermore, all terms of the credit facility will be outlined in a comprehensive Loan Agreement if the credit facility is approved by Customers Bank. The Loan Agreement and related documentation may contain additional terms and conditions regarding matters not addressed in this term sheet.

• Third party receivables less than 90 days.

Private Pay receivables and any Medicaid, Medicare, Managed Care or

Hospice receivables in dispute would be considered ineligible.

Additionally, all credit amounts in any of the eligible receivable categories

would be deducted from the "current"

Unused Fee:

An unused fee of .25% per quarter will be due on the unused portion of

the Line of Credit

Maturity:

Term Loan: 5 years

Line of Credit 12 months from closing

Amortization &

Structure:

Borrower will make monthly interest and principal payments to the lender

based upon a 25-year amortization schedule.

The borrower has the option to fund the principal payments into a sinking

fund held in escrow by the lender until the HUD takeout is complete

Advances on capital improvements will be subject to proper documentation requesting payment. Interest only will be charged on the \$1,000,000 allocated for capital improvements from the date of loan closing until the capital improvements have been substantially completed, but after 24 months, the

\$1,000,000 shall amortize on a 25-year schedule.

Interest Rate:

30 day LIBOR plus 325bps

Interest Rate Protection

Borrower will be required to provide Interest Rate Protection in the form of an interest rate swap, for 50% of the Mortgage Loan. The interest rate swap will be cross defaulted (i.e. defaults under the swap will constitute a default under the loan, and vice versa); cross terminated and cross

collateralized with the Mortgage Loan.

Commitment Fee:

A "Commitment Fee" of 1.0% of the sum of (a) the \$3,752 million Term Loan and Line of Credit (or such lesser. maximum amount approved for

the commitments approved including any amounts withheld at closing

Prepayment:

The credit facilities may be prepaid in whole (but not partially) provided that if prepaid before the 36-month anniversary of the closing, there will be a penalty ("Pre-payment Penalty") assessed of 1% of the maximum

amount of the facilities

Collateral:

The Term loan shall be evidenced and secured by 1) a first mortgage lien on the land and improvements now and hereafter acquired or

This term sheet is for discussion purposes only and is issued at a time when Customers Bank has not completed its credit review of the borrower and the collateral to secure the credit facility outlined herein. All terms and conditions are subject to a satisfactory audit by Customers Bank. Furthermore, all terms of the credit facility will be outlined in a comprehensive Loan Agreement if the credit facility is approved by Customers Bank. The Loan Agreement and related documentation may contain additional terms and conditions regarding matters not addressed in this term sheet.

construction thereon, 2) a chattel mortgage or security agreement creating first mortgage lien on the furnishings and equipment to be located at the projects and 3) an assignment of the lease with the operating entities. The term loan will shall also be secured by a lien on Birchwood OpCo, LLC's tangible and intangible personal property including but not limited to cash and deposit accounts: to the extent assignable under applicable law, licenses, certificates, permits and other governmental approvals necessary and required to operate the projects; the books and records of the Projects and accounts receivable. A Deposit Account Control Agreement will be required from the operator.

Additional Collateral:

Borrower may establish a sinking fund cash collateral account with Customers Bank. At closing, and each month thereafter through maturity, borrower will make payments into the cash collateralized account described in Amortization and Structure section. Upon closing and funding from HUD or any other lender, the cash collateral account will be released to the borrower.

Covenants:

Birchwood Prop, LLC, and Birchwood Operations, LLC Combined

DSCR not to be less than 1.25x before distributions and 1.10x after distributions.EBITDAR (excluding extraordinary income and expenses) less (a) replacement reserves of not less than \$350 per bed and (b) the amount, if any, by which management fees are less than 5 percent of effective gross income, divided by the principal and interest payments due Lenders for the applicable period.

The DSCR shall be calculated quarterly, commencing with the first full quarter immediately after the sixth month of closing the Mortgage Loan, and thereafter, commencing with June 30, 2018, the DSCR shall be tested on a trailing twelve month period. The DSCR for the twelve-month period ending December 31, 2018 shall be determined based on audited, fiscal year-end financial statements; all other periods will be tested on internally prepared financial statements.

Birchwood Operations, LLC shall demonstrate on a rolling four quarters basis, minimum average occupancy of 85%.

Management Fees are subject to subordination or curtailment in the event of Covenant default.

No additional indebtedness without Lender's written Consent

Ancillary Business Requirements:

Borrower will use Customers Bank for all accounts for this facility.

Other Expenses:

All legal fees, costs for third-party reports, disbursements and any other Lender out-of-pocket expenses for underwriting and closing the loan will be for Borrower's account and payable on the date of closing to the extent not paid for directly by the lender from the deposit referenced below.

This term sheet is for discussion purposes only and is issued at a time when Customers Bank has not completed its credit review of the borrower and the collateral to secure the credit facility outlined herein. All terms and conditions are subject to a satisfactory audit by Customers Bank. Furthermore, all terms of the credit facility will be outlined in a comprehensive Loan Agreement if the credit facility is approved by Customers Bank. The Loan Agreement and related documentation may contain additional terms and conditions regarding matters not addressed in this term sheet.

Documentation:

Documentation shall be satisfactory to Lenders and will include, but not be limited to, the terms and conditions included herein as well as provisions that are customary for a transaction of this type.

Conditions Precedent:

Lender site visit and meeting with Borrower's senior management. Additional operational, financial and regulatory compliance information may be required. Order and satisfactory review of a Phase one Environmental, Appraisal report, Property condition report, and all operating agreements, personal financial information of the guarantors and purchase agreements and any other documentation not mentioned.

Representations and Warranties:

The documents will contain those representations and warranties and covenants customarily found in transactions of this nature, and others appropriate to this transaction

Deposit

A deposit of \$40,000 will be required to cover underwriting cost upon execution of the term sheet. Any unused portion will be applied to the closing cost.

Closing:

TBD

Expiration Date:

This proposal will expire on 10 business days from the date of issuance unless previously accepted.

On behalf of Customers Bank, thank you for this opportunity and we look forward to continuing to work with you in regard to this financing request. If you have any questions concerning this matter, or if I may be of assistance in regard to any other issue, please do not hesitate to contact me at (212) 843-4549.

Sincerely,

Anthony Mai Senior Vice President Group Head, Healthcare

This term sheet is for discussion purposes only and is issued at a time when Customers Bank has not completed its credit review of the borrower and the collateral to secure the credit facility outlined herein. All terms and conditions are subject to a satisfactory audit by Customers Bank. Furthermore, all terms of the credit facility will be outlined in a comprehensive Loan Agreement if the credit facility is approved by Customers Bank. The Loan Agreement and related documentation may contain additional terms and conditions regarding matters not addressed in this term sheet.

NOTICES

If your application for business credit is denied, you have the right to a written statement of the specific reasons for the denial. To obtain the statement, please contact Customers Bank, Attn: Credit Administration Department

/ DLR, 99 Bridge Street, Phoenixville, PA 19460, 484-920-7028, or your Lender, within 60 days from the date you are notified of our decision. We will send you a written statement of reasons for the denial within 30 days of receiving your request for the statement.

When contacting the Credit Administration Department for a written statement of the specific reasons for the denial, please be prepared to offer the following information: (1) First and Last Name, (2) Date of the Notice of Declination, (3) Name of the Lender or Relationship Manager Who Worked with You, and (4) Current Mailing Address Where You Wish the Statement to be Delivered.

The Federal Equal Credit Opportunity Act prohibits creditors from discriminating against credit applicants on the basis of race, color, religion, national origin, sex, marital status, age (provided the applicant has the capacity to contract); because all or a part of the applicant's income derives from any public assistance program; or because the applicant has in good faith exercised any right under the Consumer Credit Protection Act. The Federal agency that administers compliance with this law concerning this creditor is the Federal Reserve Consumer Help Center, PO Box 1200, Minneapolis, MN 55480.



This information and the information provided on all accompanying financial statements and schedules is provided for the purpose of obtaining credit for the Applicant(s) or for the purpose of Applicant(s) guaranteeing credit for others. Applicant(s) acknowledge that representations made in this statement will be relied on by Creditor in its decision to grant such credit. This statement is true and correct in every detail and accurately represents the financial condition of the Applicant(s) on the date given below. Creditor is authorized to make all inquiries it deems necessary, either directly or through any agency employed by Lender for that purpose, to verify the accuracy of the information contained herein and to determine the creditworthiness of the Applicant(s). Applicant(s) will promptly notify Creditor of any subsequent changes which would affect the accuracy of this Statement. Creditor is further authorized to answer any questions about Creditor's credit experience with Applicant(s). Applicant(s) are aware that any knowing or willful false statements regarding the value of the above property for purposes of influencing the actions of Creditor can be a violation of federal law and may result in a fine or imprisonment or both. By signing below, each Applicant declares that he/she has read and understands the Notice Section above and, if applicable, has received the Reg B notification regarding denied credit, as well as made an election under the right to copy of appraisal or valuation.

IMPORTANT INFORMATION ABOUT PROCEDURES FOR OPENING A NEW ACCOUNT Institutions to obtain, verify, and record information that identifies each person who opens an account. What this means for you: When you open an account, we will ask for your name, address, date of birth, and other information that will allow us to identify you. We may also ask to see your driver's license or other identifying documents.

ATTACHMENT 35

Birchwood Operations, LLC and Birchwood Prop, LLC

COMBINED FINANCIAL STATEMENTS

Years Ending December 31, 2018 through 2020 (Projected) and
Year ending December 31, 2017 (Forecasted) and
Years Ended December 31, 2014 through 2016 (Historical)

Birchwood Operations, LLC and Birchwood Prop, LLC TABLE OF CONTENTS December 31, 2014 through 2016 (Historical) and 2017 (Forecasted) and 2018 through 2020 (Projected)

ACCOUNTANT'S COMPILATION REPORT								
FINANCIAL STATEMENTS								
Combined Balance Sheets under the Hypothetical Assumptions in Note 1	1							
Combined Statements of Income under the Hypothetical Assumptions in Note 1	2							
Combined Statements of Changes in Members' Equity under the Hypothetical Assumptions in Note 1	3							
Combined Statements of Cash Flows under the Hypothetical Assumptions in Note 1	4							
Summary of Significant Projection Assumptions and Accounting Policies	5							
SUPPLEMENTARY INFORMATION								
Combined Departmental Expense Schedules under the Hypothetical Assumptions in Note 1	9							



Vermont License #167

ACCOUNTANT'S COMPILATION REPORT

Ari Erlichman and Isaac Rubin Birchwood Operations, LLC and Birchwood Prop, LLC Burlington, VT 05408

Management is responsible for the accompanying projection of Birchwood Operations, LLC and Birchwood Prop, LLC, which comprises the projected combined balance sheets as of December 31, 2018, 2019, and 2020, and the projected statements of income, changes in members' equity, and cash flows for the years then ending, including the related summaries of significant assumptions and accounting policies in accordance with guidelines for the presentation of a projection established by the American Institute of Certified Public Accountants (AICPA).

Management also is responsible for the accompanying forecasted financial statements of Birchwood Terrace, which comprise the balance sheet as of December 31, 2017 and the related statements of income and members' equity and cash flows for the year then ended, and the related notes to the financial statements in accordance with accounting principles generally accepted in the United States of America.

Management also is responsible for the accompanying historical financial statements of Birchwood Terrace, which comprise the balance sheets as of December 31, 2014, 2015, and 2016 and the related statements of income and members' equity and cash flows for the years then ended, and the related notes to the financial statements in accordance with accounting principles generally accepted in the United States of America.

We have performed the compilation engagements in accordance with Statements on Standards for Accounting and Review Services promulgated by the Accounting and Review Services Committee of the AICPA. We did not examine or review the financial statements nor were we required to perform any procedures to verify the accuracy or completeness of the information provided by management. Accordingly, we do not express an opinion, a conclusion, nor provide any form of assurance on these financial statements.

Even if the CON approval from Green Mountain Care Board (GMCB) and financing occurred, there will usually be differences between the projection and actual results because events and circumstances frequently do not occur as expected, and those differences may be material. We have no responsibility to update this report for events and circumstances occurring after the date of this report.

The accompanying projections were previously complied by us and we stated in our report originally dated November 20, 2017 that we have not audited or reviewed the projections and accordingly do not express an opinion, a conclusion, nor provide any form of assurance on these financial statements. As disclosed in Note 8, management has changed certain assumptions and has restated its December 31, 2018 through 2020 (Projected) financial statements and has included a column reflecting forecasted financial statements as of and for the year ending December 31, 2017.

Ari Erlichman and Isaac Rubin Birchwood Operations, LLC and Birchwood Prop, LLC Page 2

The accompanying financial statements and this report are intended solely for the information and use of Birchwood Operations, LLC and Birchwood Prop, LLC and GMCB, and are not intended to be and should not be used by anyone other than these specified parties.

ittell Branagen & Saugat

St. Albans, Vermont January 17, 2018

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED BALANCE SHEETS UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

December 31, 2014 through 2016 (Historical), 2017 (Forecasted) and 2018 through 2020 (Projected)

ASSETS

		Seller istorical 2014	Seller Historical 2015	Seller Historical 2016	Seller Forecasted 2017	Buyer Projected 2018	Buyer Projected 2019	Buyer Projected 2020
CURRENT ASSETS					-		-	
Cash and cash equivalents	\$	19,800	\$ 5,973	\$ 57,833	\$ 61,127	\$ 267,236	\$ 498,623	\$ 752,409
Accounts receivables, net		1,862,772	1.415.546	1,646,392	1,679,320	1,712,906	1,747,164	1,782,108
Inventories		30,726	30,886	30,765	31,380	32,008	32,648	33,301
Insurance recoverables		332,700	336,254	481,596	491,228	2	~	:2
Other current assets		62,634	3,600	4,228	4,313	4,399	4,487	4,577
TOTAL CURRENT ASSETS		2,308,632	1,792,259	2,220,814	2,267,368	2,016,549	2,282,922	2,572,394
PROPERTY AND EQUIPMENT								
Land and land improvements		12,260	12,260	12,260	12,260	-	-	_
Buildings and improvements		-	341	390	_	3,213,785	3,313,785	3,413,785
Leasehold improvements		2,004,158	2,219,001	2,716,674	2,816,674	_	_	_
Equipment		1,151,446	1,230,518	1,340,381	1,390,381	350,000	400,000	450,000
Construction in progress		7,526	-	940	1000	-	*	
Contain action in progress		3,175,390	3,461,779	4,069,315	4,219,315	3,563,785	3,713,785	3,863,785
Less: Accumulated depreciation		2,592,478)	(2,748,262)	(2,900,820)	(3,056,429)	(98,916)	(224,974)	(363,176)
TOTAL PROPERTY, PLANT & EQUIPMENT		582,912	713,517	1,168,495	1,162,886	3,464,869	3,488,811	3,500,609
TOTAL THOSE EXTENSION OF THE STATE OF THE ST	-					-		
OTHER ASSETS								
Insurance recoverables		621,244	694.025	939,646	958,439	-		-
Patient fund accounts		38,653	41,505	37,193	37,937	38,696	39,470	40,259
TOTAL OTHER ASSETS		659,897	735,530	976,839	996,376	38,696	39,470	40,259
101/12 01/12/1/100210	-							
TOTAL ASSETS	\$	3.551.441	\$ 3,241,306	\$ 4.366.148	\$4,426,629	\$ 5.520.114	\$ 5.811.202	\$ 6.113.262
101/12/100210								
CURRENT LIABILITIES		LIABILITIE	S AND EQUIT	Y				
Accounts payable	\$	241,194	\$ 189,791	\$ 241,376	\$ 246,204	\$ 251,128	\$ 256,150	\$ 261,273
Salaries, wages and other compensation	•	185,633	219,676	222,496	226,946	231,485	236,115	240,837
Patient credit balances		117,402	41,716	76,970	78,509	80,080	81,681	83,315
Professional liability and workers compensation		332,700	336,254	481,596	491,228	-	=	
Line of Credit		402,100	(4)	;w;	101,000	800,000	400,000	
Current Portion of Long-term Debt			-)=	9	59,388	62,308	65,372
Other accrued liabilities		1,559	3,009	3,187	3,251	3,316	3,382	3,450
TOTAL CURRENT LIABILITIES		878,488	790,446	1,025,625	1,046,138	1,425,396	1,039,636	654,246
TO THE GOTTIEST EINBIETTIES).		
LONG-TERM LIABILITIES								
Mortgage payable, Less current portion			· **	1. - 1		2,555,036	2,492,728	2,427,356
Unamortized Debt Issuance costs			153	1000	-	(90,737)	(68,053)	(45,368)
Mortgage payable, less unamortized debt issuance costs					12	2.464.299	2,424,675	2,381,988
Patient fund accounts		38,653	41,505	37,193	37,937	38,696	39,470	40,259
Deferred rent		401,863	362,973	324,083	330,565	20,000	2	:=
Professional liability and workers compensation		621,244	694,025	939,646	977,608		. 2	
TOTAL LONG-TERM LIABILITIES		1,061,760	1.098,503	1,300,922	1,346,109	2,502,995	2,464,145	2,422,246
TOTAL EDITO-TERM EMBIETTES		7						
TOTAL LIABILITIES		1,940,248	1.888,949	2,326,547	2,392,247	3,928,391	3,503,781	3.076,493
TO TAE EIABIETTEO		114031400				7		
MEMBERS' EQUITY								
			: <u>\$</u> ()	525	12	1,591,723	2,307,421	3,036,769
Members' Equity Accumulated deficit		(2,896,936)	(3,787,137)	(5,189,681)	(6,565,617)	1,001,120	2,007,TET	5,555,765
Net contributions from Kindred Healthcare, Inc.		4,508,129	5,139,494	7,229,282	8,600,000			
TOTAL MEMBERS' EQUITY								
	_		-			1,591,723	2,307.421	3,036.769
TOTAL MEMBERS EQUIT	_	1,611,193	1,352,357	2,039,601	2,034,383	1,591,723	2,307,421	3,036,769

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED STATEMENTS OF INCOME

UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

For the Years Ended December 31, 2014 through 2016 (Historical), and the year ending 2017 (Forecasted) and the Years Ending 2018 through 2020 (Projected)

	Seller Historical 2014	Seller Historical 2015	Seller Historical 2016	Seller Forecasted 2017	Buyer Projected 2018	Buyer Projected 2019	Buyer Projected 2020
REVENUE							
Private room & board	\$ 938,791	\$ 960,343	\$ 1,110,988	\$ 1,133,208	\$ 993,698	\$ 1,007,013	\$ 1,020,328
Medicaid room & board	7,303,250	7,245,427	6,804,376	6,940,464	8,037,105	8,197,847	8,361,804
Medicare room & board	3,817,463	3,415,805	3,763,251	3,838,516	3,668,081	3,741,443	3,816,272
Other room & board	561,413	661,398	828,706	845,280	947,326	966,272	985,598
Ancillary & other patient revenue	417,642	427,414	414,441	422,730	431,184	439,808	448,604
Less: Provision for Bad Debts	13,038,559 (174,631)	12,710,387 (256,544)	12,921,762 (64,825)	13,180,197 (66,122)	14,077,394 (140,774)	14,352,383 (143,524)	14,632,606 (146,326)
Less. Flovision for Bad Debts	(174,001)	(200,011)	(01,020)	(00,122)	(110,111)	(110,021)	
INCOME FROM PATIENT CARE	12,863,928	12,453,843	12,856,937	13,114,076	13,936,620	14,208,860	14,486,280
EXPENSES							
Administrative & general	3,566,011	3,811,438	3,773,941	3,835,252	3,572,872	3,676,460	3,740,996
Property and related expenses	1,935,051	1,838,412	1,842,540	1,879,018	523,523	552,964	567,368
Plant operation and maintenance	522,124	536,872	524,057	534,538	545,228	556,133	567,255
Dietary	805,247	805,258	848,244	865,209	882,513	900,162	918,166
Laundry and linen	180,334	181,202	188,933	192,712	176,882	180,419	184,027
Housekeeping	269,924	271,214	281,646	287,279	263,499	268,769	274,144
Nursing	4,736,745	4,721,470	4,884,081	4,941,645	5,040,479	5,141,289	5,244,114
Therapy services	1,058,981	989,569	1,063,226	1,084,490	818,450	834,820	851,516
Other services	731,802	714,471	<u>853,765</u>	870,840	888,257	906,023	924,144
TOTAL EXPENSES	13,806,219	13,869,906	14,260,433	14,490,983	12,711,703	13,017,039	13,271,730
OPERATING INCOME (LOSS)	(942,291)	(1,416,063)	(1,403,496)	(1,376,907)	1,224,918	1,191,821	1,214,550
OTHER REVENUE							
Miscellaneous	618	24	584	596	608	620	632
Interest income	1,682	329	<u>368</u>	375	383	391	398
TOTAL OTHER REVENUE	2,300	353	952	971	990	1,010	1,030
INCOME TAX BENEFIT	176,378	525,509					
NET INCOME (LOSS)	\$ (763,613)	\$ (890,201)	\$ (1,402,544)	\$ (1,375,936)	<u>\$1,225,908</u>	\$ 1,192,831	\$ 1,215,580

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED STATEMENTS OF CHANGES IN MEMBERS' EQUITY UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

For the Years Ended December 31, 2014 through 2016 (Historical), and the year ending 2017 (Forecasted) and the Years Ending 2018 through 2020 (Projected)

	Seller Historical 2014	Seller Historical 2015	Seller Historical 2016	Seller Forecasted 2017	Buyer Projected 2018	Buyer Projected 2019	Buyer Projected 2020
Beginning Balance	\$ (2,133,323)	\$ (2,896,936)	\$ (3,787,137)	\$ (5,189,681)	\$ -	\$ 1,591,723	\$ 2,307,421
Equity Contributions	Nex	7 .	ā		856,178	62	2
Distributions	3#8	. P.	-		(490,363)	(477,132)	(486,232)
Net Income (loss)	(763,613)	(890,201)	(1,402,544)	(1,375,936)	1,225,908	1,192,831	1,215,580
Ending Balance	\$ (2,896,936)	\$ (3,787,137)	\$ (5,189,681)	\$ (6,565,617)	\$ 1,591,723	\$ 2,307,421	\$ 3,036,769

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED STATEMENTS OF CASH FLOWS UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

For the Years Ended December 31, 2014 through 2016 (Historical), and the year ending 2017 (Forecasted) and the Years Ending 2018 through 2020 (Projected)

	Seller Historical 2014	Seller Historical 2015	Seller Historical 2016	Seller Forecasted 2017	Buyer Projected 2018	Buyer Projected 2019	Buyer Projected 2020
CASH FLOWS FROM OPERATING ACTIVITIES							
Net Income (loss)	\$ (763,613)	\$ (890,201)	\$ (1,402,544)	\$ (1,375,936)	\$ 1,225,908	\$ 1,192,831	\$ 1,215,580
Adjustments to Reconcile Net Income (Loss) to							
Net Cash Provided by Operating Activities							
Depreciation & Amortization	253,132	155,784	152,558	155,609	121,600	148,743	160,886
Provision for doubtful accounts	205,060	256,544	64,825	66,122	140,774	143,524	146,326
Other	1,306	-	1.00		*	-	-
(Increase) decrease in:							
Accounts receivable	(350,738)	190,682	(295,671)	(99,049)	(1,853,680)	(177,782)	(181,269)
Inventory and other assets	(59,923)	(17,461)	(391,470)	(29,125)	(36,407)	(728)	(743)
Increase (decrease) in:							
Accounts Payable	(62,522)	(32,357)	41,592	4,828	251,128	5,023	5,123
Salaries, wages and other compensation	(57,621)	34,043	2,820	4,450	231,485	4,630	4,722
Patient credit balances and other accrued liabilities	217,895	(36,791)	387,505	55,678	83,395	1,668	1,701
NET CASH PROVIDED (USED) BY							
OPERATING ACTIVITIES	(617,024)	(339,757)	(1,440,385)	(1,217,424)	164,203	1,317,908	1,352,326
OI EIGHING NOTHING							
CASH FLOWS FROM INVESTING ACTIVITIES	10-50/2012/02/20	NEED BOOK	(500 100)	(450 000)	(0.500.705)	(450,000)	(450,000)
Purchases of property & equipment	_(127,680)	(298,576)	(592,163)	(150,000)	(3,563,785)	(150,000)	(150,000)
CASH FLOWS FROM FINANCING ACTIVITIES							
Proceeds from new debt					2,671,028		<u> </u>
Principal Payments of Long-Term Debt					(56,604)	(59,388)	(62,308)
Capital Contributions					856,178		
Debt Issuance Costs					(113,421)	5	2
Proceeds from Line of Credit					1,000,000		
Principal Payments on Line of Credit					(200,000)	(400,000)	(400,000)
Distributions to owners for taxes					(490,363)	(477,132)	(486,232)
Net increase in contributions due from K.H., Inc.	721,849	624,506	2,084,408	1,370,718	(,		-
Net increase in contributions due from terra, mo.			-				
NET CASH PROVIDED (USED) BY					8755555331	00/00/00/00/00/00	12202-2020
OPERATING ACTIVITIES	721,849	624,506	2,084,408	1,370,718	3,666,818	(936,520)	(948,540)
NET INCREASE (DECREASE) IN CASH	(22,855)	(13,827)	51,860	3,294	267,236	231,387	253,786
		40.000	5.070	F7 000		267 226	408 623
CASH AT BEGINNING OF YEAR	42,655	19,800	5,973	57,833		267,236	498,623
CASH AT END OF YEAR	\$ 19,800	\$ 5,973	\$ 57,833	\$ 61,127	\$ 267,236	\$ 498,623	\$ 752,409
SUPPLEMENTARY DISCLOSURES							
Interest paid	\$ -	\$ -	\$	\$	\$ 156,199	\$ 167,896	\$ 136,015
Transfers of property and equipment to Kindred	\$ (4,333)	\$ 6,859	\$ 5,380	\$	\$	\$ -	\$ -
Property and equipment purchases payable	\$ 9,254	\$ (19,046)	\$ 10,801	\$ -	\$ -	\$ -	\$ -
1 topots and odolphion becauses because							

December 31, 2014 through 2016 (Historical) and 2017 (Forecasted) and 2018 through 2020 (Projected)

NOTE 1 NATURE AND LIMITATIONS OF PROJECTIONS

The accompanying projections assume that the Company obtains approval from GMCB for a certificate of need (CON) to purchase an existing 144 bed nursing home in Burlington, Vermont and can obtain financing for the purchase. These financial projections present, to the best of management's knowledge and belief, the Company's expected financial position, results of operations, and cash flows for the years ending December 31, 2018 through 2020 if it obtains CON approval and financing. Accordingly, the projections reflect its judgment as of January 17, 2018 the date of these projections, of the expected conditions, and its expected course of action given those hypothetical assumptions. See note 8 relating to restatement of the projections from our report originally dated November 20, 2017.

The presentation is designed to assist GMCB in its decision regarding CON approval and should not be considered to be a presentation of expected future results. Accordingly, these projections may not be useful for other purposes. The assumptions disclosed herein are those that management believes are significant to the projections. Even if the projected assumptions are attained, there will usually be differences between projected and actual results, because events and circumstances frequently do not occur as expected, and those differences may be material.

NOTE 2 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Organization and Combination:

Birchwood Operations, LLC has been organized as the entity that will operate the 144 bed nursing home facility. They will lease the property from Birchwood Prop, LLC (a related organization) which will be organized to own the property being purchased. These attached projected financial statements are presented as one combined entity with all eliminating entries being reflected.

Nature of Operations:

The Company will continue to provide nursing home care and short term rehabilitation for up to 144 residents in the Burlington, Vermont area.

Inventories:

Inventories are stated at the lower of cost or market. Cost is determined on the first-in, first-out (FIFO) basis.

Property, Plant and Equipment:

Property, plant and equipment is recorded at cost and depreciation thereon is computed by the straight-line method over the assets estimated useful life.

Revenues:

A significant amount of revenues are from Medicaid and Medicare reimbursements.

Estimates

The preparation of financial statements in conformity with generally accepted accounting principles require management to make estimates and assumptions that affect certain reported amounts and disclosures. Accordingly, actual results could differ from those estimates.

December 31, 2014 through 2016 (Historical) and 2018 through 2020 (Projected)

NOTE 3 PROJECT FUNDING AND CAPITALIZATION

The projections assume that the purchase will be funded by approximately 20% of the purchase price (\$667,757) from the owner's equity and the company will borrow 80% of the purchase price (\$2,671,028) through conventional financing with an amortization over 25 years at an estimated interest rate of LIBOR plus 3.25 basis points (for a total estimate of 4.81% for these projections). This will fund the \$3,338,785 purchase price. The Loan is estimated to balloon in 5 Years and the buyers are contemplating refinancing to a HUD loan at that time.

Additional financing costs estimated at \$113,421 and additional costs associated with obtaining the CON approval estimated at \$75,000 will be funded by the owners as equity contributions for a total estimated equity contribution of \$856,178.

It is also anticipated that any cash shortfall in the first year of operation will be covered with by a Line of Credit in the amount of \$1,500,000 estimated to have the same interest rate as the mortgage above and will be paid back as cash flow allows. For the purpose of these projections, it is anticipated that the owner's will have to borrow from the line of credit in the first year for cash flow purposes and that it will be paid back as cash flow permits by the end of the third year.

NOTE 4 REVENUE ASSUMPTIONS

All revenue assumptions are based on management's best judgment about circumstances and conditions at the time these projections were prepared and are not all inclusive.

Census - Overall census numbers are projected to remain the same as they were in the first Quarter of 2017 annualized at 92.29% occupancy. No changes in overall occupancy or in the patient mix of that occupancy percentage are projected. The mix of approximately 14.1% Medicare, 73.7% Medicaid, 5.5% Private and 6.7% VA and other insurance is projected to remain the same throughout the projections.

Projected Census	2018	2019	2020
Private VA & Other Insurances Medicaid	2,663 3,227 35,763	2,663 3,227 35,763	2,663 3,227 35,763
Medicare	6,852	6,852	6,852
Total	48,505	48,505	48,505

December 31, 2014 through 2016 (Historical) and 2018 through 2020 (Projected)

NOTE 4 REVENUE ASSUMPTIONS (continued)

Rates – Private rates are anticipated to increase \$5.00 per day annually from a beginning average of \$368.15 per day to cover normal inflationary costs. VA & Other insurance rates are anticipated to continue with their current average rates inflated annually by 2% to cover normal inflationary costs. Medicaid rates are projected at the October 2017 current Medicaid Rate of \$218.12 plus an estimated Stepped up capital rate increase of \$2.25 inflated 2% annually to cover normal inflationary costs. Medicare rates are based on the Current average rate of \$535.33 and are expected to increase 2% annually to cover normal inflationary costs. Private and Part B ancillaries are expected to increase 2% per year using the 2016 revenues as a base.

Provision for Bad Debts – The projections estimate that the provision for bad debts will be approximately 1% of revenues

NOTE 5 EXPENSE ASSUMPTIONS

All expense assumptions are based on management's best judgment about circumstances and conditions at the time these projections were prepared and are not all inclusive.

Overall expenses – except where otherwise indicated below, expenses are projected using the current facility's historical 2016 costs increased annually by an estimated 2.0% for inflation.

Interest costs – Mortgage and Line of credit interests are calculated based on amortization schedules for projected debt as described in Note 3 above.

Depreciation – calculated based on allocation of the \$3,338,785 purchase price and other purchase costs of \$75,000 plus an annual increase for normal equipment and furnishings of \$50,000 annually and improvements of \$100,000 annually. Lives on all depreciable assets are set using the American Hospital Association's estimated useful lives guide.

Amortization of Debt Issuance costs – calculated based on amortizing projected financing costs of \$113,421 being amortized over the 5 year life of the loan.

Management fee – the current owner's management fee has been replaced with a management fee of 5% of revenues before bad debts per year that will cover administrative, accounting and oversight provided by a related management company.

General insurance and Worker's Compensation Insurance will not be self-funded. The estimated premiums for these insurances are management's best estimate based on the current information at the time of these projections.

Corporate – Integrated Marketing – This cost related marketing costs passed down to Birchwood relating to marketing done by the national management company and won't be continuing under new ownership and are therefore not included in these projections.

December 31, 2014 through 2016 (Historical) and 2018 through 2020 (Projected)

NOTE 5 EXPENSE ASSUMPTIONS (continued)

Contracted Services – Laundry – Management's best estimate of the cost for these services under new ownership will be \$3.60 per day.... inflated by 2% for normal inflationary cost increases.

Contracted Services – Housekeeping – Management's best estimate of the cost for these services under new ownership will be \$5.40 per day.... inflated by 2% for normal inflationary cost increases.

Contracted Therapy Services – management estimates that it will be able to realize a \$37.16 per day savings for each Medicare, Medicare HMO, and Other Insurance patient day from the cost the previous owner was incurring. Costs are then inflated 2% for annual inflation

Bed Tax – it is not anticipated the bed tax will increase throughout these projections as it has remained the same for the last several years.

Facility Rent- Ground Lease – Facility rent will go away with the purchase of the Facility. However, the current ground lease will continue and be assigned to management at closing. The remaining terms of this lease are \$2,000 per year through June 30, 2062. Management is in negotiations to try to extend this lease.

NOTE 6 2017 FORECAST ASSUMPTIONS

The assumptions for the forecast of the seller's 2017 activity are based on the seller's actual 2016 results and reflect no significant changes in census, revenues or expenses other than 2% inflationary increases throughout.

NOTE 7 DISTRIBUTIONS TO OWNERS

Distributions to owners to cover income taxes on profits passed through to them are estimated at 40% of profits. For purposes of this projection book income is estimated to approximate taxable income.

NOTE 8 RESTATEMENT OF FINANCIAL STATEMENTS

In our report dated November 20, with respect to the Projected 2018 through 2020 financial statements, management's assumptions included that part of the purchase price of \$3,338,785 would include \$100,000 for Land. The projections have been updated to reflect that the purchase price does not include Land and therefore that \$100,000 has been allocated to the cost of the building. It also reflects that there will be an assumption at closing of the ground lease that the seller has with the landowner as described in note 5. Also, financing terms have been updated as indicated in note 3.

The financial statements have also been updated to include Forecasted statements as of and for the year ending December 31, 2017.

N.	2			(4)	*	
		SUPPLEMENT	ARY INFORMATIO	N		
		99				

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED DEPARTMENTAL EXPENSE SCHEDULES UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

For the Years Ended December 31, 2014 through 2016 (Historical), and the year ending 2017 (Forecasted) and the Years Ending 2018 through 2020 (Projected)

	Hi	Seller storical 2014	F	Seller listorical 2015	ı	Seller Historical 2016	Fo	Seller recasted 2017	Buyer Projected 2018		P	Buyer rojected 2019	Buyer Projected 2020	
ADMINISTRATIVE & GENERAL														
Salary - Administrator	\$	140,541	\$	172,453	\$	150,140	\$	153,143	\$	156,206	\$	159,330	\$	162,517
Salary - Other Admin		280,816		268,853		259,436		264,625		269,918		275,316		280,822
Office supplies & postage		43,898		44,876		50,929		51,948		52,987		54,047		55,128
Communications		43,232		44,016		51,223		52,247		53,292		54,358		55,445
Travel & meetings		20,486		18,613		31,396		32,024		32,664		33,317		33,983
Advertising		22,505		64,596		74,657		76,150		77,673		79,226		80,811
Licenses & dues		22,333		19,485		25,923		26,441		26,970		27,509		28,059
Professional services		52,981		60,671		60,335		61,542		62,773		64,028		65,309
Insurances - general		51,785		48,046		58,602		59,774		60,000		61,200		62,424
Insurance - Worker's Comp		176,597		154,698		233,294		237,960		232,289		236,935		241,674
Employee benefits		308,754		313,673		340,048		346,849		353,786		360,862		368,079
Payroll taxes		464,312		477,402		495,129		505,032		545,435		589,070		636,196
Miscellaneous		34,885		23,055		24,786		25,282		25,788		26,304		26,830
Employee physicals		5,072		18,482		13,116		13,378		13,646		13,919		14,197
Seminars/inservices		6,715		18,437		34,152		34,835		35,532		36,243		36,968
Medicaid Assessment		708,412		708,412		708,412		708,412		708,412		708,412		708,412
Purchased services		38,247		128,977		126,820		129,356		131,943		134,582		137,274
Penalties		90,160		500		700		714		728		743		758
Line of Credit Interest		2		(2)		(54)				28,960		43,440		14,480
Corporate - Integrated Marketing		319,419		327,696		282,326		287,973		3.		-		38
Management fees		734,861		898,497		752,517		767,567		703,870		717,619		731,630
TOTAL ADMINISTRATIVE & GENERAL	\$ 3	566,011	\$	3,811,438	\$	3,773,941	\$ 3	3,835,252	\$:	3,572,872	\$ 3	3,676,460	\$	3,740,996
PROPERTY & RELATED EXPENSES														
Depreciation expense	\$	253,132	\$	155,784	\$	152,558	\$	155,609	\$	98,916	\$	126,059	\$	138,202
Mortgage interest	Ψ	200,102	Ψ	100,701	•	(%)	•	920	•	127,239	•	124,456	•	121,535
Mortgage interest - amortization of debt isuance costs				100				120		22,684		22,684		22,684
Facilty Rent - Ground Lease	1	410,753		1,417,296		1,427,163		1,455,706		2,000		2,000		2,000
Equipment Rent		73,340		59,608		58,381		59,549		60,740		61,955		63,194
Taxes		176,712		183,280		185,776		189,492		193,282		197,148		201,091
Insurance		21,114		22,444		18,662		18,662		18,662		18,662		18,662
Modranos	-	1	_											
TOTAL PROPERTY & RELATED EXPENSESS	\$ 1	935,051	\$	1,838,412	\$	1,842,540	\$	1,879,018	\$	523,523	\$	552,964	\$	567,368
PLANT OPERATION & MAINTENANCE														
Salary - Maintenance	\$	87.289	\$	92,624	\$	89,798	S	91,594	\$	93,426	\$	95,295	\$	97,201
Supplies	Ψ	8,265	•	10,907	*	9,331	*	9,518	•	9,708	•	9,902	7	10,100
Purchased services		132,085		145,919		140,096		142,898		145,756		148,671		151,644
Gas, fuel & oil		42,203		41,359		35,612		36,324		37,050		37,791		38,547
Electricity		155,427		155,067		151,995		155,035		158,136		161,299		164,525
Water & sewer		79,173		73,335		76,862		78,399		79,967		81,566		83,197
Garbage		17,682		17,661		20,363		20,770		21,185		21,609		22,041
Gaibage	_	,002	_	,551	-		_		_			,		<u> </u>
TOTAL PLANT OPERATION & MAINTENANCE	\$	522,124	\$	536,872	\$	524,057	\$	534,538	\$	545,228	\$	556,133	\$	567,255

Birchwood Operations, LLC and Birchwood Prop, LLC COMBINED DEPARTMENTAL EXPENSE SCHEDULES UNDER THE HYPOTHETICAL ASSUMPTIONS IN NOTE 1

For the Years Ended December 31, 2014 through 2016 (Historical), and the year ending 2017 (Forecasted) and the Years Ending 2018 through 2020 (Projected)

		Seller Historical 2014		Seller Historical 2015	Seller Historical 2016		F	Seller Forecasted 2017		Buyer Projected 2018		Buyer Projected 2019		Buyer Projected 2020
DIETARY														
Dietary salaries	\$	440,344	\$	446,649	\$	486,224	\$	495,948	\$	505,867	\$	515,984	\$	526,304
Food		321,908		321,663		320,332		326,739		333,274		339,939		346,738
Supplies & other expenses	ç —	42,995	-	36,946		41,688	_	42,522	-	43,372		44,239	-	45,124
TOTAL DIETARY	\$	805,247	\$	805,258	\$	848,244	\$	865,209	\$	882,513	\$	900,162	<u>\$</u>	918,166
LAUNDRY & LINEN														
Contracted services	\$	176,550	\$	178,739	\$	186,757	\$	190,492	\$	174,618	\$	178,110	\$	181,672
Supplies & Other Expenses		3,784	_	2,463	_	2,176	_	2,220	_	2,264	_	2,309	_	2,355
TOTAL LAUNDRY & LINEN	\$	180,334	\$	181,202	\$	188,933	\$	192,712	\$	176,882	\$	180,419	<u>\$</u>	184,027
HOUSEKEEPING														
Contracted services	\$	264,825	\$	267,049	\$	280,135	\$	285,738	\$	261,927	\$	267,166	\$	272,509
Supplies & Other Expenses		5,099	_	4,165	_	1,5 <u>11</u>	_	1,541	_	1,572	_	1,603	_	1,635
TOTAL HOUSEKEEPING	\$	269,924	\$	271,214	\$	281,646	\$	287,279	\$	263,499	\$	268,769	\$_	274,144
NURSING														
Salaries - Nurses (RN)	\$	1,468,718	\$	1,548,109	\$	1,637,875	\$	1,670,633	\$	1,704,046	\$	1,738,127	\$	1,772,890
Salaries - Nurses (LPN)		1,008,276		1,036,755		976,119		995,641		1,015,554		1,035,865		1,056,582
Salaries - Nurses (Other)		1,823,104		1,703,206		1,779,293		1,814,879		1,851,177		1,888,201		1,925,965
Salaries - Director of Nursing		83,568		88,991		96,603		98,535		100,506		102,516		104,566
Medical director		76,855		76,639		70,506		71,916		73,354		74,821		76,317
Nursing supplies & other costs		276,224		267,770		284,354		290,041		295,842		301,759		307,794
Contracted nursing services	-		_		:::	39,331	=	40,118	_	40,920	_	41,738	_	42,573
TOTAL NURSING	\$	4,736,745	\$	4,721,470	<u>\$</u>	4,884,081	\$	4,941,645	<u>\$</u>	5,040,479	<u>\$</u>	5,141,289	\$_	5,244,114
THERAPY SERVICES														
Salaries - therapy	\$	2	\$	8,856	\$	9,015	\$	9,195	\$	9,379	\$	9,567	\$	9,758
Contracted therapy		1,041,286		965,529		1,048,563		1,069,534		803,195		819,259		835,644
Therapy supplies	-	17,695	_	15,184	_	5,648	_	5,761	_	5,876	_	5,994	_	6,114
TOTAL THERAPY SERVICES	\$	1,058,981	\$	989,569	\$	1,063,226	\$	1,084,490	\$	818,450	\$	834,820	\$	851,516
OTHER SERVICES														
Salaries - activities	\$	159,702	\$	166,527	\$	174,541	\$	178,032	\$	181,593	\$	185,225	\$	188,930
Salaries - social services		88,174		98,426		103,206		105,270		107,375		109,523		111,713
Supplies		25,200		31,745		31,552		32,183		32,827		33,484		34,154
Pharmacy/X-Ray/Lab		284,641		280,609		315,750		322,065		328,506		335,076		341,778
Purchased services - pharmacy		10,014		10,835		10,557		10,768		10,983		11,203		11,427
Purchased services - activities		3,338		3,565		4,297		4,383		4,471		4,560		4,651
Other services		160,733	-	122,764	-	213,862	-	218,139	_	222,502	_	226,952	_	231,491
TOTAL OTHER SERVICES	\$	731,802	\$	714,471	\$	853,765	\$	870,840	\$	888,257	\$	906,023	\$	924,144

ATTACHMENT 39

Percentage of short-stay residents who improved in their ability to move around on their own (Higher Better)		1) Facility has implemented a comprehensive oversight in the admission assessments and documentation to capture an accurate picture of admission function. 2) The Physical and Occupational therapists proactively meet weekly with the IDT team to capture earlier detection of any signs of decline in function. 3) The weekly IDT risk meeting reviews alternative plans /interventions to improve function. 4) Review of results of this QM is done monthly at QAPI committee to determine if plan is improving outcomes.
Percentage of short-stay residents who report moderate to severe pain. (Lower Better)	-6.4	Facility will establish an initiative with the Interdisciplinary team including the Physicians and Nurse practitioners for pain management on an ongoing basis including the following: 1) Establish a pain scoring system that reflects the patient's actual experiences. 2) As pain ratings are subjective and each person's reliability in rating pain, the Facility plans to use alternative pain scales recommended by the QIO to help reflect the actual patient effects. 3) Quantifying the timing of routine pain med compared to PRN med is given may shed some light on the process of collecting pain ratings and its improvement. 4) QAPI will include utilizing the pain scale provided by the QIO which has more definitions of what each level of pain represents to help quantify and treat appropriately. 5) Educate staff, patients and families on new pain scale to assist in achieving this goal. 6) Review the results of the monthly QMs with the QAPI committee to determine if the above pain management plan is improving the quality management of overall pain.
Percentage of short-stay residents with pressure ulcers that are new or worsened. (Lower Better)	-0.04	1) One patient has had a worsening in his/her wound and the facility continue to work toward zero declines in condition 2) The wound care nurse and the medical team work hand in hand on treatment protocols approved by the wound care association 3) The social service team and psych services work to encourage patients who make choices that may not be in the best of their overall health maintenance (wound care included). 4) The results of wound management is being reviewed monthly by the QAPI committee to ensure positive outcomes are obtained.
Percentage of short-stay residents who newly got an antipsychotic medication.Percentage of short-stay residents who got an antipsychotic medication for the first time during this nursing home admission. (Lower Better)	-2.1	1) 2018 plan was already started on November 28, 2017 with the new final rule phase 2 use of PRN psychotropics has also encouraged Facility to go to the next step of overall psychotropic utilization. 2) The Facility has discontinued antipsychotic medications in last 30 days which have not been being utilized consistently. The committee will continue to review at the weekly risk meeting with Physician/NP oversight to ensure appropriate use, target behaviors are identified and discuss alternative interventions where appropriate 3) The plan is to utilize psychotropic medications as last option and more implementation of non-pharmacological approaches specific to each individual patient needs as mentioned in above behavior affecting other QM. 4) The Facility will educate current staff, new hires, patients and families on above plan to achieve this goal. 5) Review the results of the monthly QMs with the QAPI committee to determine if the above plan is improving the quality management of short and long stay residents utilizing antipsychotics.
Long Stay		

Percentage of long-stay residents who report moderate to severe pain. (Lower Better)	-8.6	Facility will establish an initiative with the Interdisciplinary team including the Physicians and Nurse practitioners for pain management on an ongoing basis including the following: 1) Establish a pain scoring system that reflects the patient's actual experiences. 2) As pain ratings are subjective and each person's reliability in rating pain, the Facility plans to use alternative pain scales recommended by the QIO to help reflect the actual patient effects. 3) Quantifying the timing of routine pain med compared to PRN med is given may shed some light on the process of collecting pain ratings and its improvement. 4) QAPI will include utilizing the pain scale provided by the QIO which has more definitions of what each level of pain represents to help quantify and treat appropriately. 5) Educate staff, patients and families on new pain scale to assist in achieving this goal. 6) Review the results of the monthly QMs with the QAPI committee to determine if the above pain management plan is improving the quality management of overall pain.
Percentage of long-stay low-risk residents who lose control of their bowels or bladder. (Lower Better)	-5.3	1) The Facility will continue to work on LNA documentation to properly capture the correct bowel and bladder status. Accurate and complete documentation greatly contributes to MDS capturing significant declines. 2) The SDC /designee will re-educate and oversee the q shift documentation to avoid errors in documentation and missed opportunities to improve outcomes. 3) The results of this oversight will be reviewed monthly at QAPI to determine outcome of plan.
Percentage of long-stay residents who were physically restrained. (Lower Better)	-0.4	1) One resident at the Facility in the last six months has had a restraint. The Facility's goal is to have no restraints but unfortunately this resident has failed multiple alternative interventions. 2) This resident is being assessed daily for changes in condition that that may allow the facility to discontinue this intervention. 3) The DNS/designee will ensure staff are aware of goal to be restraint free and for no resident to have a restraint implemented without an IDT team comprehensive evaluation. 4) The results of the restraint free goal will be reviewed monthly at QAPI.
Percentage of long-stay residents whose ability to move independently worsened. (Lower Better)	-5.6	1) Facility will continue to review with the IDT team, on a weekly basis, at its risk meetings any patient who has had a decline in any ADLs or decline in independent mobility to create a plan to help improve status. 2) Once the plan is established the team communicates the plan to the LNAs/care givers to ensure a comprehensive approach to improving ADL/mobility outcomes. If end of life goal would be care & comfort. (Often seen with Alzheimer's/ terminal patients) 3) The QM is also reviewed monthly by the committee to determine if current plan is impacting outcome ADLs or independent mobility.
Percentage of long-stay residents whose need for help with daily activities has increased. (Lower Better)	-6.6	1) Facility will continue to review with the IDT team, on a weekly basis, at its risk meetings any patient who has had a decline in any ADLs or decline in independent mobility to create a plan to help improve status. 2) Once the plan is established the team communicates the plan to the LNAs/care givers to ensure a comprehensive approach to improving ADL/mobility outcomes. If end of life goal would be care & comfort. (Often seen with Alzheimer's/ terminal patients) 3) The QM is also reviewed monthly by the committee to determine if current plan is impacting outcome ADLs or independent mobility.
Percentage of long-stay residents who lose too much weight. (Lower Better)	-1.3	1) Facility will have a weekly risk weight meeting with dietician and IDT team to identify any resident who is at risk before weight loss occurs and implement supplements or other resident specific interventions. 2) New Food service chef to work closely with dietician on more comprehensive finger food options for Alzheimer residents or other residents who have difficulties sitting to eat either related to cognitive status or movement disorders. 3) The plans for each resident identified at these risk meetings will be shared with staff/ patients and families to assist in achieving this goal. 4) Facility will review the results of the monthly QMs with the QAPI committee to determine if the plan is improving the quality management of weight loss.

Percentage of long-stay residents who got an antianxiety or hypnotic medication. (Lower Better)	-8.4	1) Facility has implemented the new final rule phase 2 use of PRN psychotropics has also encouraged Facility to go to the next step of overall psychotropic utilization 2) The Facility has already discontinued several Anxiolytic & Hypnotic medications in last 30 days which have not been being utilized consistently after review at the weekly risk meeting with Physician/NP oversight. The committee will continue to review weekly for appropriate use, identify target behaviors and discuss alternatives including non-pharmacological interventions 3) The plan is to utilize psychotropic medications as a last option and more implementation of non-pharmacological approaches specific to each individual resident needs as mentioned in above behavior affecting other QM. 4) The Facility will continue to educate current staff, new hires, patients and families on above plan to achieve this goal. 5) Review the results of the monthly QMs with the QAPI committee to determine if the above plan is improving the quality management of anxiolytic and hypnotic use.
Percentage of long-stay residents who got an antipsychotic medication. (Lower Better)	-21.1	1) Facility has implemented the new final rule phase 2 use of PRN psychotropics has also encouraged Facility to go to the next step of overall psychotropic utilization. 2) The Facility has discontinued antipsychotic medications in last 30 days which have not been being utilized consistently. The committee will continue to review at the weekly risk meeting with Physician/NP oversight to ensure appropriate use, target behaviors are identified and discuss alternative interventions where appropriate 3) The plan is to utilize psychotropic medications as last option and more implementation of non-pharmacological approaches specific to each individual patient needs as mentioned in above behavior affecting other QM. 4) The Facility will continue to educate current staff, new hires, patients and families on above plan to achieve this goal. 5) Review the results of the monthly QMs with the QAPI committee to determine if the above plan is improving the quality management of short and long stay residents utilizing antipsychotics.

ATTACHMENT 40



DEPARTMENT OF DISABILITIES, AGING AND INDEPENDENT LIVING

Division of Licensing and Protection
HC 2 South, 280 State Drive
Waterbury VT 05671-2060
http://www.dail.vermont.gov
Survey and Certification Voice/TTY (802) 241-0480
Survey and Certification Fax (802) 241-0343
Survey and Certification Reporting Line (888) 700-5330
To Report Adult Abuse: (800) 564-1612

April 27, 2017

Ms. Alecia Dimario, Administrator Kindred Transitional Care & Rehab Birchwood Terrace 43 Starr Farm Rd Burlington, VT 05408-1321

Provider ID #: 475003

Dear Ms. Dimario:

The Division of Licensing and Protection completed a survey at your facility on **April 19, 2017**. The purpose of the survey was to determine if your facility was in compliance with Federal participation requirements for nursing homes participating in the Medicare/Medicaid programs. This survey found that your facility was in substantial compliance with the participation requirements. Congratulations to you and your staff.

Sincerely,

Pamela M. Cota, RN Licensing Chief

amleMCota PN

Enclosure



DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

An Edichman

PRINTED: 04/27/2017 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:				E CDNSTRUCTION	(X3) DATE SURVEY COMPLETED	
		475003	B. WING		04/19/2017	
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	S 4	,		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	TEMENT OF DEFICIENCIES Y MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTI (EACH CORRECTIVE ACTION SHOUL CROSS-REFERENCED TO THE APPRO DEFICIENCY)	D BE COMPLETION	
F 000	INITIAL COMMEN	TS	F 000			
	conducted an annu	ensing and Protection lal recertification survey There were no regulatory llt.				
					•	
	į				Î	
) 	ı		*			
					· ·	
		1981				
	Σi.					
LABORATOR'	Y DIRECTOR'S OR PROVI	DER/SUPPLIER REPRESENTATIVE'S SI	GNATURE	TITLE	(X6) DATE	

Any deficiency statement ending with an asterisk (*) denotes a deficiency which the institution may be excused from correcting providing it is determined that other safeguards provide sufficient protection to the patients. (See instructions.) Except for nursing homes, the findings stated above are disclosable 90 days following the date of survey whether or not a plan of correction is provided. For nursing homes, the above findings and plans of correction are disclosable 14 days following the date these documents are made available to the facility. If deficiencies are cited, an approved plan of correction is requisite to continued program participation.

Art Erlighman Zenith Gure HG: Jul 18, 2017 13:34

Division of Licensing and Protection

HC 2 South, 280 State Drive Waterbury VT 05671-2060

http://www.dail.vermont.gov

Survey and Certification Voice/TTY (802) 241-0480

To Report Adult Abuse: (800) 564-1612

Survey and Certification Fax (802) 241-0343

Survey and Certification Reporting Line: (888) 700-5330

May 5, 2016

Ms. Alecia Dimario, Administrator Kindred Transitional Care & Rehab 43 Starr Farm Rd Burlington, VT 05408-1321

Provider ID #: 475003

Dear Ms. Dimario:

The Division of Licensing and Protection completed a survey at your facility on April 27, 2016. The purpose of the survey was to determine if your facility was in compliance with Federal participation requirements for nursing homes participating in the Medicare and Medicaid programs. This survey found the most serious deficiency in your facility to be a pattern of deficiencies that constitute no actual harm with potential for more than minimal harm that is not immediate jeopardy as evidenced by the attached CMS-2567 whereby corrections are required. All references to regulatory requirements contained in this letter are found in Title 42, Code of Federal Regulations.

Plan of Correction (POC)

A POC for the deficiencies, which is your allegation of compliance, must be received by May 17, 2016. Failure to submit an acceptable POC by May 17, 2016 may result in imposition of remedies or termination of your provider certification. Your POC must contain the following:

- What corrective action will be accomplished for those residents found to have been affected by the deficient practice;
- How you will identify other residents having the potential to be affected by the same deficient practice and what corrective action will be taken;
- What measures will be put into place or what systemic changes you will make to ensure that the deficient practice does not recur; and,
- How the corrective actions will be monitored to ensure the deficient practice will not recur, i.e., what quality assurance program will be put into place.
- The dates corrective action will be completed.

An Erlichman Zenith Care HG kii 18, 2617 13:3

The remedies, which will be imposed if substantial compliance has not been achieved by May 27, 2016, will include the following:

Denial of Payment for New Admissions effective July 27, 2016

An Enforcement Cycle has been initiated based on the citation of deficiencies at a "D" level or greater at your facility. All statutory/mandatory enforcement remedies are effective based on the beginning survey of the Enforcement Cycle. Your Enforcement Cycle began with the April 27, 2016, survey. All surveys conducted after April 27, 2016, with deficiencies at a "D" level or greater become a part of this Enforcement Cycle. The enforcement cycle will not end until substantial compliance is achieved for all deficiencies from all surveys within an enforcement cycle. Facilities are expected to achieve and maintain continuous substantial compliance. If you do not achieve substantial compliance, the CMS Regional Office and/or State Medicaid Agency must deny payments for new admissions. We are also recommending to the CMS Regional Office and/or State Medicaid Agency that your provider agreements be terminated on October 27, 2016 if substantial compliance is not achieved by that time. A change in the seriousness of the deficiencies on May 27, 2016 may result in a change in the remedy selected.

Allegation of Compliance

If you believe these deficiencies have been corrected, you may contact Suzanne Leavitt, RN, MS, Assistant Division Director, Division of Licensing and Protection with your written credible allegation of compliance. If you choose and so indicate, the POC may constitute your allegation of compliance. We may accept the written allegation of compliance and presume compliance until substantiated by a revisit or other means. In such a case, the recommended remedy listed above would not be imposed at that time.

If, upon the subsequent revisit, your facility has not achieved substantial compliance, a civil money penalty may be imposed by the CMS Regional Office beginning on the last day of survey and continue until substantial compliance is achieved. Additionally, the CMS Regional Office will impose the other remedies indicated above or revised remedies, if appropriate.

Informal Dispute Resolution

In accordance with §488.331, you have one opportunity to question cited deficiencies through an informal dispute resolution process. To be given such an opportunity, you are required to send your written request, along with the specific deficiencies being disputed, and an explanation of why you are disputing those deficiencies, to Suzanne Leavitt, RN, MS, Assistant Division Director, Division of Licensing and Protection. This written request must be received by this office by May 17, 2016. An incomplete informal dispute resolution process will not delay the effective date of any enforcement action.



Opportunity for Independent Informal Dispute Resolution (IIDR)

If you have already requested an Informal Dispute Resolution (IDR) from the State Agency, your request for IIDR will only be allowed if it is made before the State's IDR is completed. If you chose to request an IIDR with an Independent Panel, your written request for an IIDR must be sent to Suzanne Leavitt, RN, MS, State Survey Agency Director. The State Survey Agency will forward your request to the IIDR Panel, and they will inform you when and how the IIDR will be conducted. Your request for IIDR must be made no later than 10 calendar days from the date of your receipt of this letter.

Sincerely,

Pamela M. Cota, RN Licensing Chief

PamlaMCota DN

Enclosure:

	F ISOLATED DEFICIENCIES WHICH CAUSE	PROVIDER#	MULTIPLE CONSTRUCTION A, BUILDING:	DATE SURVEY					
NO HARM WIT FOR SNFs AND	H ONLY A POTENTIAL FOR MINIMAL HARM NFs	475003	B. WING	COMPLETE: 4/27/2016					
	VIDER OR SUPPLIER TRANSITIONAL CARE & REĤAB BIRCE	STREET ADDRESS, CITY, STATE, ZIP CODE 43 STARR FARM RD BURLINGTON. VT							
ID PREFIX TAG	SUMMARY STATEMENT OF DEFICIENCIES								
F 247	483.15(e)(2) RIGHT TO NOTICE BEF	ORE ROOM/ROOM	MATE CHANGE						
	A resident has the right to receive notic	e before the resident'	s room or roommate in the facility	is changed.					
	This REQUIREMENT is not met as ev Based on staff interview and record revi sample of 31 (Resident # 78) of a room	iew, the facility failed	I to notify 1 applicable resident in blude:	the stage 2					
	Per staff interview with a facility Social Worker (SW) on 4/26/16 at 11:43 AM, Resident # 78 had a roc change on 4/6/16. There is no written evidence in the clinical record that verbal or written notice was g prior to the room change. During the interview, the SW confirmed there is no evidence of written or venotice of room change.								
F 388	483.40(c)(3)-(4) PERSONAL VISITS BY PHYSICIAN, ALTERNATE PA/NP								
	Except as provided in paragraphs (c)(4) the physician personally.) and (f) of this section	on, all required physician visits mu	st be made by					
	At the option of the physician, required visits by the physician and visits by a place accordance with paragraph (e) of this second	sician, required visits in SNFs, after the initial visit, may alternate between personal and visits by a physician assistant, nurse practitioner or clinical nurse specialist in ph (e) of this section.							
	This REQUIREMENT is not met as ex Based on record review and confirmed have required physician visits that alter findings include the following:	iled to ensure that 2 of 31 sampled al physician and the nurse practition	l residents oner. The						
	1. Per medical record review for Resident The resident was seen by the nurse practice.	ent #50, s/he was last etitioner on 12/28/15	t seen by the attending physician of 2/1/16 and 4/28/16.	n 11/16/15.					
	2. Per medical record review for Resident on 10/21/15. The resident was seen by	ent #166, evidences t the nurse practitione	hat s/he was last seen by the attender on 11/19/15, 1/22/16 and 3/24/16	ling physician 5.					
	Per interview with the Unit Manger, co visits with the nurse practitioner as req		e that the attending physician did n	ot alternate					

Any deficiency statement ending with an asterisk (*) denotes a deficiency which the institution may be excused from correcting providing it is determined that other safeguards provide sufficient protection to the patients. (See instructions,) Except for nursing homes, the findings stated above are disclosable 90 days following the date of survey whether or not a plan of correction is provided. For nursing homes, the above findings and plans of correction are disclosable 14 days following the date these documents are made available to the facility. If deficiencies are cited, an approved plan of correction is

Jul 18, 2017 13:34

The above isolated deficiencies pose no actual harm to the residents

031099

If continuation sheet | I of I

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:			(X2) MULTIPLE A BUILDING _	CONSTRUCTION		E SURVEY 1PLETED
		475003	B. WING		04/	27/2016
	PROVIDER OR SUPPLIER D TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	43	REET ADDRESS, CITY, STATE, ZIP CODE STARR FARM RD JRLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	TEMENT OF DEFICIENCIES / MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRE (EACH CORRECTIVE ACTION SHI CROSS-REFERENCED TO THE APF DEFICIENCY)	OULD BE	(X5) COMPLETION DATE
F 000	INITIAL COMMEN	TS	F 000	u		
	conducted an unan recertification surve following regulatory result.	censing and Protection inounced onsite annual ey on 4/25/26 - 4/27/16. The y violations were cited as a	5 050			
F 253 SS=E	483.15(h)(2) HOUS MAINTENANCE S	ERVICES	F 253			
	maintenance servi	rovide housekeeping and ces necessary to maintain a nd comfortable interior.				
	by: Based on observa provide housekeep to maintain a sanitainterior in resident	NT is not met as evidenced tion, the facility failed to sing and maintenance services ary, orderly and comfortable rooms and bathing areas on A, e findings include the				
	presence of the Nu	tour on 4/26/16 in the irsing Home Administrator and irector the following were				
	and bureaus were chipped and peelin Walls are noted to with gouges and cowalls. Waste paper plastic bag enclosus themselves were for accumulated dried	ent Rooms with built in closets found to have rough edges, ag paint and missing knobs. have peeling paint, sheet rock ove base sloughing from the er baskets were found without ures, therefore the baskets bound to be heavily soiled with materials. Oxygen se were found to have				

Any deficiency statement ending with an asterisk (*) denotes a deficiency which the institution may be excused from correcting providing it is determined that other safeguards provide sufficient protection to the patients. (See instructions.) Except for nursing homes, the findings stated above are disclosable 90 days following the date of survey whether or not a plan of correction is provided. For nursing homes, the above findings and plans of correction are disclosable 14 days following the date these documents are made available to the facility. If deficiencies are cited, an approved plan of correction is requisite to continued program participation.

FORM CMS-2567(02-99) Previous Versions Obsolete

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

Event ID: F88B11 Enirchman Facility ID: 475003

TITLE

If continuation sheet Page 1 of 11

(X6) DATE

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES Comb Com HC

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

CENTER	13 FOR MEDICARE	& WILDICAID SLIVICES	111 6 1	0.33			
	TEMENT OF DEFICIENCIES DEPLAY OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA (X2) PROVIDER/SUPPLIER/CLIA (X3) PROVIDER/SUPPLIER/CLIA (X4) PROVIDER/SUPPLIER/SU		l ' '		E CONSTRUCTION		E SURVEY PLETED
		475003	B, WING			04/2	27/2016
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER		43	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD URLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENCY	TEMENT OF DEFICIENCIES / MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREF TAG		PROVIDER'S PLAN OF CORRECTION (EACH CORRECTIVE ACTION SHOULD CROSS-REFERENCED TO THE APPROP DEFICIENCY)	BE	(X5) COMPLETION DATE
F 279 SS=D	accumulated dust, C Wing: 10 Reside and bureaus were for chipped and peelin Walls are noted to with gouges and cowalls. Waste pape plastic bag enclosus themselves were for dried accumulated concentrators in us accumulated dust, Special Care Unit: in closets and bure edges and chipped noted to have peelis loughing from the were found without therefore baskets the avily soiled with B and C Wing Bath were found to have located towards the was found to have the floor and on the were noted to be so Confirmation was rethat the resident roneed of cosmetic rowaste paper basket need cleaning. 483.20(d), 483.20(d)	debris and dried on spills. ent Rooms with built in closets found to have rough edges, g paint and missing knobs. have peeling paint, sheet rock ove base sloughing from the r baskets were found without res, therefore baskets bund to be heavily soiled with materials. Oxygen e were found to have debris and dried on spills. 5 Resident Rooms with built aus were found to have rough and peeling paint. Walls are ing paint and cove base walls. Waste paper baskets plastic bag enclosures, hemselves were found to be dried accumulated materials. Ining/Shower Rooms: Both e cracked tiles on the walls e floor. C Wing shower room dirty soiled laundry resting on the towel racks. Privacy curtains biled as well. made at the end of the tour oms, and shower rooms are in epairs. Also confirmed that its and oxygen concentrators. k)(1) DEVELOP		279			

An Edichman

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 2 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

Artellactic D

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

		(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:	(X2) MULTIPLE A. BUILDING _	E CONSTRUCTION	(X3) DATE COME	SURVEY
		475003	B. WING		04/2	7/2016
	PROVIDER OR SUPPLIER D TRANSITIONAL CA	ARE & REHAB BIRCHWOOD TER	43	REET ADDRESS, CITY, STATE, ZIP CODE STARR FARM RD URLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC	ATEMENT OF DEFICIENCIES Y MUST BE PRECEDED BY FULL LSC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECT (EACH CORRECTIVE ACTION SHOU CROSS-REFERENCED TO THE APPRO DEFICIENCY)	LD BE	(X5) COMPLETION DATE
F 279	to develop, review comprehensive plate The facility must deplan for each reside objectives and time medical, nursing, an eeds that are ideassessment. The care plan must are to be furnished resident's highest and psychosocial of \$483.25; and any be required under due to the resident's \$483.10, including under §483.10(b)(-1)	the results of the assessment and revise the resident's an of care. levelop a comprehensive care lent that includes measurable etables to meet a resident's and mental and psychosocial ntified in the comprehensive of the describe the services that do attain or maintain the practicable physical, mental, well-being as required under services that would otherwise §483.25 but are not provided t's exercise of rights under the right to refuse treatment	F 279			
	by: Based on record if facility failed to as	review and staff interview the sure that a plan of care was sident (R #94) in a stage 2				
	area on the Left 44 (Minimum Data Se assessment tool), weekly skin check on 12/15/15, the repressure ulcer. The facility from 1/5/16	Resident #94 has a pressure th toe. In reviews of the MDS et - a comprehensive Nursing Assessments, and s since his/her first admission esident is coded not to have a e resident was absent from the 5 to 3/17/16 when s/he returned alcers. The resident was				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID:F88B11

Facility ID: 475003

If continuation sheet Page 3 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES Zerollo Care In Care

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

NAME OF PROVIDER OR SUPPLIER KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER (X4) ID PREFIX TAG (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) F 279 Continued From page 3 discharged to the hospital on 4/8/16 for fever and altered mental status. R #94 was readmitted on 4/11/16 and the Nursing Admission Assessment Skin inspection notes a pressure area which is staged as Unstageable on the L 4th Toe. The Admission assessment MDS notes the Unstageable pressure ulcer on the 4th toe.	(X3) DATE SURVEY COMPLETED		
KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER (X4) ID PREFIX TAG (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) F 279 Continued From page 3 discharged to the hospital on 4/8/16 for fever and altered mental status. R #94 was readmitted on 4/11/16 and the Nursing Admission Assessment Skin inspection notes a pressure area which is staged as Unstageable on the L 4th Toe. The Admission assessment MDS notes the	04/27/2016		
PREFIX TAG (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) F 279 Continued From page 3 discharged to the hospital on 4/8/16 for fever and altered mental status. R #94 was readmitted on 4/11/16 and the Nursing Admission Assessment Skin inspection notes a pressure area which is staged as Unstageable on the L 4th Toe. The Admission assessment MDS notes the			
discharged to the hospital on 4/8/16 for fever and altered mental status. R #94 was readmitted on 4/11/16 and the Nursing Admission Assessment Skin inspection notes a pressure area which is staged as Unstageable on the L 4th Toe. The Admission assessment MDS notes the			
In a review of the record although the care plan does have an initiated Problem of Impaired Skin Integrity there are no prevention or care interventions found in the plan of care. Per interview on 4/27/16 at 1:40 PM, the Charge Nurse on the Unit confirmed that there was no care plan with interventions developed for a new pressure area discovered on 4/11/18. F 309 SS=D Each resident must receive and the facility must provide the necessary care and services to attain or maintain the highest practicable physical, mental, and psychosocial well-being, in accordance with the comprehensive assessment and plan of care.			
This REQUIREMENT is not met as evidenced by: Based on staff interview and record review, the facility failed to ensure 1 of 4 applicable residents in the stage 2 sample of 31 (Resident #154) received the necessary care to maintain the highest practicable well being. Findings include: Per record review, Resident #154 had an			

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 4 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

An Edichman Zenith Care HG PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION		(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:	A. BUILDING	E CONSTRUCTION	(X3) DATE SURVEY COMPLETED	
		475003	B. WING		04/	27/2016
	PROVIDER OR SUPPLIER D TRANSITIONAL C	ARE & REHAB BIRCHWOOD TER	43	REET ADDRESS, CITY, STATE, ZIP CODE S STARR FARM RD URLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC	ATEMENT OF DEFICIENCIES BY MUST BE PRECEDED BY FULL LSC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTION (EACH CORRECTIVE ACTION SHOUL CROSS-REFERENCED TO THE APPROF DEFICIENCY)	D BE	(X5) COMPLETION DATE
F 353 SS=D	unwitnessed fall of evidence that staff Signs (NVS) after policy states that if monitor NVS ever every 30 minutes 2 hours. On 4/27/2 Nurses (DNS) combeen done after the they have not beed 483.30(a) SUFFICE PER CARE PLAN The facility must if provide nursing armaintain the higher and psychosocial determined by resindividual plans of the facility must provide nursing armaintain the higher and psychosocial determined by resindividual plans of the facility must provide all resident care plans: Except when waiv section, licensed repersonnel. Except when waiv section, the facility nurse to serve as duty.	in 3/30/16. There is no if performed Neurological Vital the unwitnessed fall. Facility if the fall is unwitnessed by 15 minutes for 1 hour, then for 1 hour, then every hour for 16 at 12:25 PM, the Director of firmed that NVS should have be unwitnessed fall and that in done. SIENT 24-HR NURSING STAFF is ave sufficient nursing staff to not related services to attain or est practicable physical, mental, well-being of each resident, as ident assessments and it care. For ovide services by sufficient of the following types of t				
	This REQUIREME by:	ENT is not met as evidenced				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID:F88B11

Facility ID: 475003

If continuation sheet Page 5 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES Zandto Care 1963

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION A. BUILDING				SURVEY PLETED		
		475003	B. WING		04/2	27/2016
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	4:	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD URLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENCY	TEMENT OF DEFICIENCIES 'MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTIOI (EACH CORRECTIVE ACTION SHOULD CROSS-REFERENCED TO THE APPROPI DEFICIENCY)	BE	(X5) COMPLETION DATE
F 353	record review, the there was sufficient well-being of each provision of care as plans of care. Finding 1). Per resident into #28 stated during the has requested to us during the night shi are too busy to ass commode and that although that is not elimination. S/he states urine all night long S/he states that on his/her urine as long finally had to request that s/he waited so	I resident interviews and facility failed to assure that a staff to assure the highest resident and assure the ecording to the resident's regident and assure the ecording to the resident's regident interview on 4/26/16, Resident resident interview that s/he se his/her bedside commode ft and has been told that they ist him/her to the bedside s/he must use the bedpan his/her preferred method of tates that s/he has told staff infortably use the bedpan. As that s/he has "held" his/her to avoid using the bedpan. The night recently s/he had held g as s/he could but that s/he set the bedpan. S/he relates long that s/he had edpan and that his/her clothing	F 353			
F 371 SS=E	Director of Nurses aware that they are efforts to recruit bo Nurses Aides (LNA 483.35(i) FOOD PR		F 371			
	considered satisfact authorities; and	om sources approved or story by Federal, State or local distribute and serve food				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 6 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

Ari Erlichman enth Caru HG PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:	l ' '	SLE CONSTRUCTION 3	(X3) DATE SURVEY COMPLETED	
		475003	B. WING_		04/	27/2016
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER		STREET ADDRESS, CITY, STATE, ZIP CODE 43 STARR FARM RD BURLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	TEMENT OF DEFICIENCIES / MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTIO (EACH CORRECTIVE ACTION SHOULD CROSS-REFERENCED TO THE APPROF DEFICIENCY)) BE	(X5) COMPLETION DATE
F 371	Continued From pa under sanitary cond	-	F 37	1		
	by: Based on observa interview the facilit the nourishment kit are stored under sa findings include the Per tour on 4/26/16 Units in the presen (RD) the following 1. Dementia unit k have 2 multi-servir chips and a multi-	of the kitchenettes on all 3 ce of the Registered Dietician was observed: itchen cabinet was found to all bags of partially used potato erving bag of partially used ealed nor dated as to the day ducts were opened. drawer was found to store os containing a white powder. cation as to the contents in the date when the powder was put				
	with the following of 4/25/16. 3 Styrofo dated 4/23/16. The with a dried sticky	lates: 3/27/16, 4/23/16 and am containers labeled jelly e refrigerator is heavily soiled substance making it difficult to lower drawers. Confirmation				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 7 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES

An Erlienman CENTERS FOR MEDICARE & MEDICAID SERVICES Zamilli Care RG

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:			E CONSTRUCTION	(X3) DATE SURVEY COMPLETED		
		475003	B WING		04/2	27/2016
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	4	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD CURLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	NTEMENT OF DEFICIENCIES Y MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTIO (EACH CORRECTIVE ACTION SHOULE CROSS-REFERENCED TO THE APPROP DEFICIENCY)) BE	(X5) COMPLETION DATE
F 371	that the refrigerator cleaning. They als	age 7 e Unit Manager and the RD r is very dirty and needs to confirm that the food should as per the date on the	F 371			
F 514 SS=D	Dementia Unit and foods need to be stated 483.75(I)(1) RES	med the findings on the B Wing and voiced that the tored and labeled properly. *LETE/ACCURATE/ACCESSIB*	F 514			
	each resident in ac professional standa complete; accurate	naintain clinical records on accordance with accepted ards and practices that are aly documented; readily stematically organized.				
	information to iden the resident's asse services provided;	ening conducted by the State;				
	by: Based on staff into review, the facility and accurate clinic acceptable profess 1 of 31 residents in Resident #77, the Per medical record	NT is not met as evidenced erview and medical record failed to maintain complete cal records in accordance with sional standards of practice for the sampled group. For findings include the following:				
		6 with diagnoses to include				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID:F88B11

Facility ID: 475003

If continuation sheet Page 8 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES And Erdictionners CENTERS FOR MEDICARE & MEDICAID SERVICES Zonith Care HG

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

	OF DEFICIENCIES OF CORRECTION	(X1) PROVIDER/SUPPLIER/CLIA Juli 1 IDENTIFICATION NUMBER:	` '	E CONSTRUCTION		E SURVEY PLETED
		475003	B. WING		04/	27/2016
	PROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	- 4	STREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD BURLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	TEMENT OF DEFICIENCIES YMUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTIO (EACH CORRECTIVE ACTION SHOULD CROSS-REFERENCED TO THE APPROP DEFICIENCY)	BE	(X5) COMPLETION DATE
F 514	the right foot with p toes, End Stage Re on Renal Dialysis a Pulmonary Disease. Per review of the D Record for Resider evidences docume Registered Nurse (bumped his toe betours-2nd toe stump Area cleaned and a please follow up that Licensed Practical Resident #77's 2nd approximately 4:15 wound was treated. Per observation of there is no evidence notified of the newlevidence that wour Confirmation is mathere is no docume physician regarding acquired wound or Per interview with the lack of communication is mathere is no docume physician regarding acquired wound or Per interview with the lack of communication is mathere is no docume physician regarding acquired wound or Per interview with the lack of communication is mathered.	mputation, Chronic Ulcer of partial amputation of multiple penal Disease with dependency and Chronic Obstructive end. Dialysis Communication of the 477 dated 4/25/16, on the Dialysis RN) ["Resident #77 must have tween your facility and tween your facility and tween your facility and tween your facility and the skin tear, bleeding. In the penaltiple of the stump on 4/25/16 at the stump on 4/25/16 at the medical record on 4/26/15 and dressed. The medical record on 4/26/15 are that the physician was by acquired wound nor is there and care was conducted. The the Unit Manager that the physician to the gresident #77's newly treatment there of. The LPN on 4/27/16 regarding inication to the physician	F 514	DEFICIENCY		
F9999	of documentation p	y acquired wound and the lack pertaining to the wound, nade by the LPN stating ["I just	F9999			
	7.13 Nursing Servi	ices				

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 9 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

An Emeriman Zendir Cam HG PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

(X3) DATE SURVEY STATEMENT OF DEFICIENCIES (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION COMPLETED AND PLAN OF CORRECTION IDENTIFICATION NUMBER: A. BUILDING B. WING 475003 04/27/2016 STREET ADDRESS, CITY, STATE, ZIP CODE NAME OF PROVIDER OR SUPPLIER 43 STARR FARM RD KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER **BURLINGTON, VT 05408** PROVIDER'S PLAN OF CORRECTION SUMMARY STATEMENT OF DEFICIENCIES (X5) COMPLETION (X4) ID PREFIX (EACH CORRECTIVE ACTION SHOULD BE (EACH DEFICIENCY MUST BE PRECEDED BY FULL **PREFIX** DATE CROSS-REFERENCED TO THE APPROPRIATE REGULATORY OR LSC IDENTIFYING INFORMATION) TAG TAG DEFICIENCY) F9999 Continued From page 9 F9999 provide nursing and related services to attain or maintain the highest practicable physical, mental and psychosocial well-being of each resident, as determined by resident assessments and individual plans of care or as specified by the licensing agency. (d)Staffing Levels. The facility shall maintain staffing levels adequate to met resident needs. (1) At a minimum, nursing facilities must provide: (i) no fewer than 3 hours of direct care per resident per day, on a weekly average, including nursing care, personal care and restorative nursing care, but not including administration or supervision of staff; and of the three hours of direct care, no fewer than 2 hours per resident per day must be assigned to provide standard LNA care(such as personal care, assistance with ambulation, feeding, etc.)performed by LNAs or equivalent staff and not including meal preparation, physical therapy or the activities program. This requirement is NOT MET as evidenced by: Based on record review and Resident and Staff interviews, the facility failed to assure that there was sufficient staff to assure the highest well-being and care according to the resident's plans of care. Findings include: 1. In a review of the facility staffing patterns for February, March and April of 2016 it is found that the facility failed to meet the state regulatory requirement for 2 hours per day per resident of direct (LNA) care on 10 days in February, 9 days in March, and 6 of 23 reported days in April. Additionally the Total number of care hours of 3 hours per day per resident were not met on 8 days in February, 9 days in March, and 5 of 23

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 10 of 11

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

ANT EQUIPMENT

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		(X2) MULTIPLE CONSTRUCTION A. BUILDING			(X3) DATE SURVEY COMPLETED	
		475003	B. WING		04/	27/2016
	PROVIDER OR SUPPLIER D TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	4	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD BURLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	TEMENT OF DEFICIENCIES / MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECT (EACH CORRECTIVE ACTION SHOU CROSS-REFERENCED TO THE APPRO DEFICIENCY)	LD BE	(X5) COMPLETION DATE
F9999	reported days in Ap In an interview on a Director of Nurses aware that they are		F9999			

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID:F88B11

Facility ID: 475003

If continuation sheet Page 11 of 11



DEPARTMENT OF DISABILITIES, AGING AND INDEPENDENT LIVING

Division of Licensing and Protection
HC 2 South, 280 State Drive
Waterbury, VT 05671-2060
http://www.dail.vermont.gov

Survey and Certification Voice/TTY (802) 241-0480 Survey and Certification Fax (802) 241-0343 Survey and Certification Reporting Line: (888) 700-5330

To Report Adult Abuse: (800) 564-1612

May 18, 2016

Ms. Alecia Dimario, Kindred Transitional Care & Rehab Birchwood Terrace 43 Starr Farm Rd Burlington, VT 05408-1321

Dear Ms. Dimario:

Enclosed is a copy of your acceptable plans of correction for the survey conducted on April 27, 2016. Please post this document in a prominent place in your facility.

We may follow-up to verify that substantial compliance has been achieved and maintained. If we find that your facility has failed to achieve or maintain substantial compliance, remedies may be imposed.

Sincerely,

Pamela M. Cota, RN

Mak Chafe

Licensing Chief

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		A BUILDING			(X3) DATE SURVEY COMPLETED		
		475003	B. WING_			04/2	27/2016
	OVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER		STREET ADDRESS, CITY, STATE, ZIF 43 STARR FARM RD BURLINGTON, VT 05408	CODE		
(X4) ID PREFIX TAG	(EACH DEFICIENCY	TEMENT OF DEFICIENCIES MUST BE PRECEDED BY FULL SC TOENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF ((EACH CORRECTIVE ACTI CROSS-REFERENCED TO T DEFICIENC'	ON SHOULD HE APPROPE	BE	(X5) COMPLETION DATE
F 253 M SS=E Tris Trib R p ttri O B a cl V W W p ttri C C Tri S Tri S Tri S Tri S Tri S Tri S Tri S Tri S Tri S T	ponducted an unantecertification survestillowing regulatory esult. 83.15(h)(2) HOUS MAINTENANCE SETT IN ESULTENANCE SETT IN E	ensing and Protection nounced onsite annual by on 4/25/26 - 4/27/16. The violations were cited as a EKEEPING & EKVICES ovide housekeeping and es necessary to maintain a aid comfortable interior. It is not met as evidenced ion, the facility failed to ng and maintenance services by, orderly and comfortable boms and bathing areas on A, findings include the sing Home Administrator and rector the following were not Rooms with built in closets bund to have rough edges, and paint and missing knobs. The baskets were found without res, therefore the baskets und to be heavily soiled with materials. Oxygen a were found to have	F 26	Preparation and/or execute correction daes not constitute agreement by the provider alleged facts or conclusion statement of deficiencies. is prepared and/or execute provisions of federal and statement of deficiencies. F 253 Walls and closets with note were patched and painted. Oxygen concentrators and immediately assessed and of the Director of Nursing/or educate staff regarding the related to equipment use and The Director of Nursing/or conduct monthly rounds to being maintained. All results the QAPI meeting for 3 modern compliance. The Director of Maintenance Designee will complete modernify any areas of needed will be established so that refor completion. The Director of Maintenance Designee will work with the develop a systematic plan for repairs in patient rooms and	ante admission of the truth is set forth. The plan of the truth is set forth. The plan of the district of the plan of the	hof the in the footreet transition or hof the in the footreet transition of the footreet transition of the footreet transition or district	ee 2016 s es ers ers ed
Alle	anla	KCU EXECUTION EXECUTION OF SIGN	tive	Director	5	1131	II.

Any deficiency statement ending with an asterisk (*) denotes a deficiency which the institution may be excused from correcting providing it is determined that other safeguards provide sufficient protection to the patients. (See instructions.) Except for nursing homes, the findings stated above are disclosable 90 days following the date of survey whether or not a plan of correction is provided. For nursing homes, the above Indings and plans of correction are disclosable 14 days following the date these documents are made available to the facility. If deficiencies are cited, an approved plan of correction is requisite to continued program participation.

FORM CMS-2567(02-99) Previous Versions Obsolete

Event IO: F88B11

Facility ID: 475003

If continuation sheet Page 1 of 11

An Eritanman Zenim Care HG

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIÉR/CLIA IDENTIFICATION NUMBER:		(X2) MULTIPLE CONSTRUCTION A BUILDING			(X3) DATE SURVEY COMPLETED		
		475003	B. WING			04/27	/2016
NAME OF PROVIDER		RE & REHAB BIRCHWOOD TER		43	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD BURLINGTON, VT 05408		
	CH DEFICIENCY	TEMENT OF DEFICIENCIES Y MUST BE PRECEDED BY FULL SC IDENTIFYING INFORMATION)	ID PREFI TAG		PROVIDER'S PLAN OF CORRECTID (EACH CORRECTIVE ACTION SHOULD CROSS-REFERENCED TO THE APPROP DEFICIENCY)	BE C	(X5) COMPLETION DATE
C Wing and bu chipper Walls a with go walls. plastic themse dried a concern accum. Special in close edges noted the slough were for the slow was for the floor were not confirmation that the need of waste proceed of the slow waste proced of the slow waste procedure.	ulated dust, g: 10 Reside reaus were if d and peelin are noted to buges and cr Waste pape bag enclosue elves were for comulated strators in us ulated dust. I Care Unit: ets and bure and chipped to have peelit ing from the bund without re baskets the could be so I towards the und to have bund to have or and on the oted to be so mation was ne e resident ro of cosmetic re paper baskel leaning. (d), 483.20(f)	debris and dried on spills. Int Rooms with built in closets bund to have rough edges, g paint and missing knobs. have peeling paint, sheet rock we base sloughing from the reskets were found without res, therefore baskets bund to be heavily soiled with materials. Oxygen e were found to have debris and dried on spills. 5 Resident Rooms with built aus were found to have rough and peeling paint. Walls are ng paint and cove base walls. Waste paper baskets plastic bag enclosures, nemselves were found to be dried accumulated materials. Ing/Shower Rooms: Both cracked tiles on the walls of floor. C Wing shower room dirty soiled laundry resting on towel racks. Privacy curtains oiled as well. Inade at the end of the tour oms, and shower rooms are in epairs. Also confirmed that is and oxygen concentrators.		253	reviewed with the QAPI committee months to ensure compliance. The ED is responsible for overall complete parameters and the ED is responsible for overall complete parameters. Fasa POC accepted 5/17/16 RT Fasa PoC accepted 5/17/16 RT Fasa PoC accepted 5/17/16 RT House audits have been completed or residents with potential/actual skin in issues to ensure that care plans have developed. The SDC/designee completed re-edu on developing care plans for resident potential/actual skin integrity issues. The DNS/designee will complete ran care plan audits on resident's with potential/actual skin integrity impairmenthly x 90 days. The results of the audits will be reviewed at the monthly meeting for 3 months to ensure complete complete parameters.	ronthly x 3 repliance. Yemblaus for the control of	en) pmc

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID; F88811

Facility ID: 475003

If continuation sheet Page 2 of 11

Jul 18 2017 19 19

DEPARTMENT OF HEALTH AND HUMAN SERVICES

FORM APPROVED OMB NO. 0938-0391 CENTERS FOR MEDICARE & MEDICAID SERVICES (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION (X3) DATE SURVEY STATEMENT OF DEFICIENCIES IDENTIFICATION NUMBER: COMPLETED AND PLAN OF CORRECTION A BUILDING 475003 B. WING 04/27/2016 STREET ADDRESS, CITY, STATE, ZIP CODE NAME OF PROVIDER DR SUPPLIER 43 STARR FARM RD KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER **BURLINGTON, VT 05408** (X5) COMPLETION DATE SUMMARY STATEMENT OF DEFICIENCIES PROVIDER'S PLAN OF CORRECTION 1D (X4) ID (EACH CORRECTIVE ACTION SHOULD BE (EACH DEFICIENCY MUST BE PRECEDED BY FULL PREFIX PREFIX REGULATORY OR LSC IDENTIFYING INFORMATION) CROSS-REFERENCED TO THE APPRIDPRIATE TAG TAG DEFICIENCY) F 279 Continued From page 2 F 279 A facility must use the results of the assessment to develop, review and revise the resident's comprehensive plan of care. The facility must develop a comprehensive care plan for each resident that includes measurable objectives and timetables to meet a resident's medical, nursing, and mental and psychosocial needs that are identified in the comprehensive assessment. The care plan must describe the services that are to be furnished to attain or maintain the resident's highest practicable physical, mental, and psychosocial well-being as required under §483.25; and any services that would otherwise be required under \$483.25 but are not provided due to the resident's exercise of rights under §483.10, including the right to refuse treatment under §483.10(b)(4). This REQUIREMENT is not met as evidenced by: Based on record review and staff interview the facility failed to assure that a plan of care was developed for 1 resident (R #94) in a stage 2 sample of 31. Findings include: Per record review. Resident #94 has a pressure area on the Left 4th toe. In reviews of the MDS (Minimum Data Set - a comprehensive assessment tool), Nursing Assessments, and weekly skin checks since his/her first admission on 12/15/15, the resident is coded not to have a pressure ulcer. The resident was absent from the

FORM CMS-2567(02-99) Previous Versions Obsolete

facility from 1/5/16 to 3/17/16 when s/he returned with no pressure ulcers. The resident was

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 3 of 11

PRINTED: 05/05/2016

An Erlichmun Zenith Gare HG rul 18, 2017, 13 33

DEPARTMENT OF HEALTH AND HUMAN SERVICES
CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		CONSTRUCTION		(X3) DATE SURVEY COMPLETED	
		475003	B. WING			27/2016	
	ROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	43	(REET ADDRESS, CITY, STATE, ZIP CO I STARR FARM RD URLINGTON, VT 05408	DDE		
(X4) ID PREFIX TAG	(EACH DEFICIENC)	NTEMENT OF DEFICIENCIES YMUST BE PRECEDED BY FULL SC IDEN TIPYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF COF (EACH CORRECTIVE ACTION CROSS-REFERENCED TO THE DEFICIENCY)	SHOULD BE	(X5) COMPLETION DATE	
F 309	altered mental stat 4/11/16 and the Nu Skin inspection no staged as Unstage Admission assessr Unstageable press In a review of the redoes have an initia Integrity there are interventions found interview on 4/27/ Nurse on the Uniticare plan with inte pressure area disc 483.25 PROVIDE HIGHEST WELL E Each resident mus provide the necess or maintain the hig mental, and psych accordance with the and plan of care. This REQUIREME by: Based on staff int facility failed to en residents in the staff #154) received the	nospital on 4/8/16 for fever and us. R #94 was readmitted on using Admission Assessment tes a pressure area which is able on the L 4th Toe. The ment MDS notes the ure ulcer on the 4th toe. record although the care plan ated Problem of Impaired Skin no prevention or care in the plan of care. Per 16 at 1:40 PM, the Charge confirmed that there was no reventions developed for a new overed on 4/11/18. CARE/SERVICES FOR	F 279	F 309 Resident # 154 no longer resfacility. All residents with un-witnes the potential to be affected. The SDC/designee has re-ed licensed nurses on the facility policy/procedure with emph un-witnessed falls. The DNS/designee will committee to compliance. The DNS is responsible for	sed falls have lucated the ty fall lasis on NVS for aplete post fall x 90 days. The le reviewed at the o ensure	r	
	include:	, Resident #154 had an		compliance. F309 PDC accepted 5/17		y Ril PMC	

FORM CMS-2567(02-99) Previous Versions Obsolete

Event 10: F88B11

Facility ID: 475003

If continuation sheet Page 4 of 11

Ari Erlichman Zenith Care HG Ari 18: 2017 | 3:33

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT	OF DEFICIENCIES F CORRECTION	(X1) PROVIDER/SUPPLIER/CLA IDENTIFICATION NUMBER:		E CONSTRUCTION		SURVEY PLETED
		475003	B. WING		04/	27/2016
	ROVIDER OR SUPPLIER TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	43	REET ADDRESS, CITY, STATE, ZIP CODE S STARR FARM RD URLINGTON, VT 05408		
(X4) IO PREFIX TAG	FACH DEFICIENC	ATEMENT OF DEFICIENCIES Y MUST BE PRECEDEO BY FULL SC IDENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORREC (EACH CORRECTIVE ACTION SHOU CROSS-REFERENCED TO THE APPR DEFICIENCY)	JLD BE	(X5) COMPLETION DATE
F 353 \$S=D	unwitnessed fall or evidence that staff Signs (NVS) after policy states that if monitor NVS every every 30 minutes 2 hours. On 4/27/1 Nurses (DNS) combeen done after the they have not been done after they have not been 483.30(a) SUFFIC PER CARE PLAN. The facility must he provide nursing an maintain the higher and psychosocial determined by resindividual plans of the facility must provide nurse for each opersonnel on a 24 care to all resident care plans: Except when waity section, licensed repersonnel. Except when waity section, the facility nurse to serve as duty.	n 3/30/16. There is no performed Neurological Vital the unwitnessed fall. Facility the fall is unwitnessed y 15 minutes for 1 hour, then for 1 hour, then every hour for 6 at 12:25 PM, the Director of firmed that NVS should have a unwitnessed fall and that n done. EENT 24-HR NURSING STAFF Serve sufficient nursing staff to ad related services to attain or est practicable physical, mental, well-being of each resident, as ident assessments and		F 353 Resident # 28 needs are being me plan. All residents have the potential to affected. The DNS/designee has been active recruiting for licensed nurses and nursing assistants. Current strategutilizing local as well as multiple recruitment options. The facility contracted with temporary staffing to provide direct care for the facility has an active Recruitment Retention committee. The ED/designee will audit staffing Results of audits will be reviewed inouthly QAPI committee for 3 mensure compliance. The ED is responsible for overall compliance. F353 POC accepted 5 17 16 for the staffing results of audits will be reviewed the staffing results of audits will be reviewed to the staffing results of audits w	ely licensed gies include other state has g agencies ity. The and ng weekly, i at the nonths to	
	This REQUIREMED by:	ENT IS not met as evidenced		1000 100 mm 41-0. 1. 16-	0	

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID; F68B11

Facility ID: 475003

If continuation sheet Page 5 of 11

Ari Erlichman Zenith Gare HG Jul 18, 2017 13,33

DEPARTMENT OF HEALTH AND HUMAN SERVICES

CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

	OF DEFICIENCIES F CORRECTION	(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:	(X2) MULTIPLE CONSTRUCTION A. BUILDING		(X3) DATE SURVEY COMPLETED	
		475003	B, WING		04/2	27/2016
	ROVIDER OR SUPPLIER O TRANSITIONAL CA	RE & REHAB BIRCHWOOD TER	43	TREET ADDRESS, CITY, STATE, ZIP CODE 3 STARR FARM RD URLINGTON, VT 05408		
(X4) ID PREFIX YAG	(EACH DEFICIENC	ATEMENT OF DEFICIENCIES Y MUST BE PRECEDED BY FULL, LSC (DENTIFYING INFORMATION)	ID PREFIX TAG	PROVIDER'S PLAN OF CORRECTI (EACH CORRECTIVE ACTION SHOUL CROSS-REFERENCED TO THE APPRO DEFICIENCY)	D BE	(X5) COMPLETION DATE
F 353	Based on staff and record review, the there was sufficient well-being of each provision of care a plans of care. Find 1). Per resident int #28 stated during the night share too busy to assocommode and the although that is no elimination. S/he states urine all night long S/he states that or his/her urine as lor finally had to request that s/he waited so	d resident interviews and facility falled to assure that it staff to assure the highest resident and assure the ccording to the resident's ings include: erview on 4/26/16, Resident the resident interview that s/he se his/her bedside commode ift and has been told that they sist him/her to the bedside t s/he must use the bedpan at his/her preferred method of states that s/he has told staff omfortably use the bedpan. As that s/he has "held" his/her to avoid using the bedpan. The night recently s/he had held as as s/he could but that s/he est the bedpan. S/he relates bedpan and that his/her clothing edpan and that his/her clothing	F 353			
F 371 SS=E	In an interview on Director of Nurses aware that they are efforts to recruit by Nurses Aides (LN/483.35(i) FOOD P STDRE/PREPARE The facility must - (1) Procure food for considered satisfa authorities; and	4/26/16 at 2:30 PM the (DNS) stated that the facility is e understaffed despite many oth Nurses and Licensed As). ROCURE, E/SERVE - SANITARY	F 371			

DEPARTMENT OF HEALTH AND HUMAN SERVICES

FORM APPROVED OMB NO, 0938-0391 CENTERS FOR MEDICARE & MEDICAID SERVICES (X3) DATE SURVEY STATEMENT OF DEFICIENCIES (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION COMPLETED **IDENTIFICATION NUMBER:** AND PLAN OF CORRECTION A. BUILDING _ B. WING 04/27/2016 475003 STREET ADORESS, CITY, STATE, ZIP CODE NAME OF PROVIDER OR SUPPLIER 43 STARR FARM RD KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER **BURLINGTON, VT 05408** PROVIDER'S PLAN OF CORRECTION (X5) COMPLETION DATE SUMMARY STATEMENT OF DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) (EACH CORRECTIVE ACTION SHOULD BE PREFIX PRÉFIX CROSS-REFERENCED TO THE APPROPRIATE TAG TAG DEFICIENCY F 371 F 371 F 371 Continued From page 6 May 26, 2016 under sanitary conditions Food in the nourishment room/refrigerator that was not dated, unsealed was discarded immediately. The refrigerators in the nourishment rooms were all cleaned and placed on a cleaning schedule. This REQUIREMENT is not met as evidenced Alt residents have the potential to be by: affected. Based on observation and confirmed by staff interview the facility falled to ensure that food in the nourishment kitchens on A, B and C Wings The SDC/designee has re-educated the staff are stored under sanitary conditions. The on food storage/cleaning guidelines per findings include the following: policy. Per tour on 4/26/16 of the kitchenettes on all 3 The Culinary/Hospitatity Manager/designee Units in the presence of the Registered Dietician will complete regular audits of nourishment (RD) the following was observed: rooms for 1 month and then weekly audits for 60 days. Results of these audits will be 1. Dementia unit kitchen cabinet was found to reviewed at the monthly QAPI meeting to have 2 multi-serving bags of partially used potato ensure compliance. chips and a multi-serving bag of partially used The ED is responsible for overall vanilla wafers unsealed nor dated as to the day compliance. any of the food products were opened. F371 POCaccepted 5/19/16 RTYENDLOG PN PN 2. B Wing cabinet drawer was found to store multiple souffle cups containing a white powder. There is no identification as to the contents in the souffle cups or the date when the powder was put in use. The cups are marked 2 T. 3. C Wing refrigerator was found to have multiple residents' personal food items (fried

FORM CMS-2567(02-99) Previous Versions Obsolete

chicken-cherry peppers) in disposable containers with the following dates: 3/27/16, 4/23/16 and 4/25/16. 3 Styrofoam containers labeled jelly dated 4/23/16. The refrigerator is heavily soiled with a dried sticky substance making it difficult to open and close the lower drawers. Confirmation

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 7 of 11

PRINTED: 05/05/2016

Zemith Care HG Jul 18, 2017 13 33

DEPARTMENT OF HEALTH AND HUMAN SERVICES

FORM APPROVED CENTERS FOR MEDICARE & MEDICAID SERVICES OMB NO. 0938-0391 (X3) DATE SURVEY (X2) MULTIPLE CONSTRUCTION (X1) PROVIDER/SUPPLIER/CLIA STATEMENT OF DEFICIENCIES COMPLETED AND PLAN OF CORRECTION IDENTIFICATION NUMBER: A. BUILDING_ 475003 04/27/2016 STREET ADDRESS, CITY, STATE, ZIP CODE NAME OF PROVIDER OR SUPPLIER 43 STARR FARM RD KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER BURLINGTON, VT 05408 PROVIDER'S PLAN OF CORRECTION (X5) COMPLETION SUMMARY STATEMENT OF DEFICIENCIES (X4) ID (EACH CORRECTIVE ACTION SHOULD BE (EACH DEFICIENCY MUST BE PRECEDED BY FULL PREFIX PREFIX CROSS-REFERENCED TO THE APPROPRIATE REGULATORY OR LSC IDENTIFYING INFORMATION). TAG TAG DEFICIENCY F 371 F 371 Continued From page 7 is made by both the Unit Manager and the RD that the refrigerator is very dirty and needs cleaning. They also confirm that the food should of been discarded as per the date on the containers. The RD also confirmed the findings on the Dementia Unit and B Wing and voiced that the foods need to be stored and labeled properly. F 514 May 26, 2016 F 514 F 514 483.75(I)(1) RES SS=D RECORDS-COMPLETE/ACCURATE/ACCESSIB Resident # 77 physician has been notified LF regarding skin tear. Appropriate documentation was completed in the medical The facility must maintain clinical records on record. each resident in accordance with accepted The LPN involved was re-educated on the professional standards and practices that are policy related to contact of physician and the complete; accurately documented; readily need for appropriate documentation. accessible; and systematically organized, House audit of resident's care plans with The clinical record must contain sufficient potential/actual skin integrity issues has been information to identify the resident; a record of completed to ensure no other residents are the resident's assessments: the plan of care and services provided; the results of any affected by this practice. preadmission screening conducted by the State; The SDC/designee has re-educated licensed and progress notes. nurses on MD notification and documentation requirements. This REQUIREMENT Is not met as evidenced bv: The DNS/designee will complete random Based on staff interview and medical record audits on documentation on residents with review, the facility failed to maintain complete skin integrity issues monthly x 90 days. The and accurate clinical records in accordance with results of these audits will be reviewed at the acceptable professional standards of practice for monthly OAPI committee x3 months to 1 of 31 residents in the sampled group. For ensure compliance. Resident #77, the findings include the following:

FORM CMS-2567 (02-99) Previous Versions Obsolete

Per medical record review Resident #77 was

admitted on 2/20/16 with diagnoses to include

Event ID: F88811

Facility ID: 475003

compliance.

The DNS is responsible for overall

F514 POC accepted 5/17/16 PTrembley If continuation sheet Page 8 of 11

PRINTED: 05/05/2016

Ad Erlichman Zenith Care HG Jul 18 (2017 13 32)

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES PRINTED: 05/05/2016 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION (X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:		(X2) MULTIPLE CONSTRUCTION A BUILDING			COMPLETED	
		475003	B. WING			/27/2016
	PROVIDER OR SUPPLIER D TRANSITIONAL C	ARE & REHAB BIRCHWOOD TER		STREET ADDRESS, CITY, STATE, ZIP COL 43 STARR FARM RD BURLINGTON, VT 05408		
(X4) ID PREFIX TAG	(EACH DEFICIENC	TATEMENT OF DEFICIENCIES CYMUST BE PRECEDED BY FULL LSC IDENTIFYING INFORMATION)	IO PREFII TAG	PROVIDER'S PLAN OF CORF ((EACH CORRECTIVE ACTION S CROSS-REFERENCED TO THE A DEFICIENCY)	HOULD BE	(X5) COMPLETION DATE
F 514	the right foot with toes, End Stage Fon Renal Dialysis Pulmonary Disease. Per review of the Record for Reside evidences docum Registered Nurse bumped his toe bours-2nd toe stun Area cleaned and please follow up to Licensed Practica Resident #77's 2r approximately 4: wound was treated Per observation of there is no evident notified of the new Confirmation is not there is no docum physician regarding the new of documentation confirmation was did not do it"]. FINAL OBSERVATATION TO THE PROPERTY OF THE PROPER	amputation, Chronic Ulcer of partial amputation of multiple Renal Disease with dependency and Chronic Obstructive se. Dialysis Communication ent #77 dated 4/25/16, lentation from the Dialysis (RN) ["Resident #77 must have etween your facility and inp with skin tear, bleeding. If antibiotic ointment applied thanks."] Per observation at Nurse (LPN) evaluated ind toe stump on 4/25/16 at 15 PM at the facility. The ed and dressed. If the medical record on 4/26/15 ince that the physician was why acquired wound nor is there und care was conducted. In the LPN on 4/27/16 regarding incoming the physician was acquired wound and the lack in pertaining to the wound, is made by the LPN stating ["I just ATIONS		999		

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID; F88811

Facility ID: 475003

If continuation sheet Page 9 of 11

An Erlichman Zenith Care HG Jul 18: 2017 13:33

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED: 05/05/2016 FORM APPROVED OMB NO 0938-0391

	OF DEFICIENCIES F CORRECTION	(X1) PROVIDER/SUPPLIER/CLIA IDENTIFICATION NUMBER:	(X2) MULTIPLE CONSTRUCTION A. BUILDING			(X3) DATE SURVEY COMPLETED	
		475003	B. WING,	_		04/2	27/2016
NAME OF PROVIDER OR SUPPLIER KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER (X4) ID SUMMARY STATEMENT OF DEFICIENCIES PREFIX (EACH DEFICIENCY MUST BE PRECEDED BY FULL		ID PREFI)	43 B	TREET ADDRESS, CITY, STATE, ZIP CODE S STARR FARM RD URLINGTON, VT 05408 PROVIDER'S PLAN OF CORRECTIO (EACH CORRECTIVE ACTION SHOULD	BE	(X5) COMPLETION DATE	
TAG	REGULATORY OR L	SC IDENTIFYING INFORMATION)	TAG		CROSS-REFERENCED TO THE APPROP DEFICIENCY)	RIAIE	
F9999	provide nursing and maintain the higher and psychosocial widetermined by residindividual plans of licensing agency. (d) Staffing Levels adec (1) At a minimum, (i) no fewer than 3 resident per day, on ursing care, personursing care, personursing care, but no supervision of staff direct care, no fewer per day must be as LNA care(such as pambulation, feeding equivalent staff and preparation, physic program. This requirement is Based on record reinterviews, the facility and care plans of care. Finding the facility failed to requirement for 2 higher than 1. In a review of the February, March are the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of care of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to requirement for 2 higher than 3 of the facility failed to failed the facility failed to fa	d related services to attain or st practicable physical, mental vell-being of each resident, as dent assessments and care or as specified by the The facility shall maintain quate to met resident needs. nursing facilities must provide: hours of direct care per in a weekly average, including shall care and restorative of including administration or and of the three hours of er than 2 hours per resident is igned to provide standard personal care, assistance with a getc.) performed by LNAs or do not including meal at therapy or the activities. NOT MET as evidenced by: wiew and Resident and Staff lity failed to assure that there to assure the highest according to the resident's	F99		Resident # 28 needs are being met p plan. All residents have the potential to be affected. The DNS/designee has been actively recruiting for licensed nurses and lic nursing assistants. Current strategie utilizing local as well as multiple off recruitment options. The facility has contracted with temporary staffing a to provide direct care for the facility facility has an active Recruitment and Retention committee. The ED/designee will audit staffing Results of audits will be reviewed at monthly QAPI conumittee for 3 monensure compliance. The ED is responsible for overall compliance. F1911 POL accepted 517116 Rt	censed sinclude ner state segencies. The ad	

An Edichman Jul 18 2017 13:33

DEPARTMENT OF HEALTH AND HUMAN SERVICES

FORM APPROVED OMB NO. 0938-0391 CENTERS FOR MEDICARE & MEDICAID SERVICES (X3) DATE SURVEY COMPLETED (X1) PROVIDER/SUPPLIER/CLIA (X2) MULTIPLE CONSTRUCTION STATEMENT OF DEFICIENCIES IDENTIFICATION NUMBER: AND PLAN OF CORRECTION A, BUILDING B. WING 04/27/2016 475003 STREET ADDRESS, CITY, STATE, ZIP CODE NAME OF PROVIDER OR SUPPLIER 43 STARR FARM RD KINDRED TRANSITIONAL CARE & REHAB BIRCHWOOD TER **BURLINGTON, VT 05408** SUMMARY STATEMENT OF DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) PROVIDER'S PLAN OF CORRECTION (X5) COMPLETION ID PREFIX (X4) ID PREFIX TAG (EACH CORRECTIVE ACTION SHOULD BE DATE CROSS-REFERENCED TO THE APPROPRIATE TAG DEFICIENCY) F9999 F9999 Continued From page 10 reported days in April. In an interview on 4/26/16 at 2:30 PM the Director of Nurses (DNS) stated that the facility is aware that they are understaffed despite many efforts to recruit bot Nurses and Licensed Nurses Aides 2. Also see F353.

FORM CMS-2567(02-99) Previous Versions Obsolete

Event ID: F88B11

Facility ID: 475003

If continuation sheet Page 11 of 11

PRINTED: 05/05/2016

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEFICIENCIES AND PLAN OF	(X1) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION	(X2) MULTIPLE CONSTRUCTION A. BUILDING B. WING	(X3) DATE SURVEY COMPLETED			
CORRECTION	NUMBER	B. WING	04/23/2015			
NAME OF PROVIDER OF SUI	335505 PLIER	STREET ADDRESS, CITY, S'	TATE, ZIP			
HIGHLAND CARE CENTER		91 31 175TH STREET JAMAICA, NY 11432	91 31 175TH STREET			
For information on the nursing l	home's plan to correct this deficien	cy, please contact the nursing home or the state survey agency.				
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF DOR LSC IDENTIFYING INFORM	DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED I	BY FULL REGULATORY			
F 0205 Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	Tell the resident or the resident' will hold the resident's bed. **NOTE-TERMS IN BRACKET Based on record review and intervibat specified the duration of the not provided with written or verb family members interviewed, Resident #76. This resulted in no harm with pote The finding is: Resident #76 is [AGE] years old visit dependent upon staff for all Activ The Nursing Notes dated 3/18/15 notes documented that the husbar care. There was no documented expolicy from the Social Work Dep The facility's Bed hold Policy do acute care setting, the Social Work resident's current bed hold status representative. On 4/22/15 during the family intervited in the was never informed of the visits his wife almost daily. On 4/23/15, at approximately 11:0 assigned to the resident sends a le hold. On 4/23/15, at approximately 12:2 which the bed hold policy is explicative the letter left in the chart	s representative in writing how long the nursing home S HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY riew, the facility did not ensure that the resident's family member bed hold. Specifically when a resident was transferred to the hosp al information concerning the facility's bed hold policy. This was ential for more than minimal harm that was not immediate jeopard with [DIAGNOSES REDACTED]. The resident has long term and rities of Daily Living. documented that the resident was sent to hospital per the Doctor's ad was informed by Nursing that the resident was sent to a local h vidence in the medical record that a written notice was sent to the	received written notice ital the spouse was evident for 1 out of 3 dy. d short term memory problems and order. Also, the Nursing ospital for immediate of amily of the bed hold in its transferred to an regarding the esent to the designated of the hospital in March He stated that he that the Social Worker uration of the bed hold letter in tals. There is no hard the Social Worker			
F 0280 Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	Allow the resident the right to p care plan. **NOTE- TERMS IN BRACKET Based on record review and intervinput during care planning activit planning for the resident. This wa This resulted in no actual harm withe finding is: Resident #76 is [AGE] years old vdependent upon staff for all Activities of Da On 4/21/15 the resident's husband CPM. He further stated that he dithe medical record was reviewed to the Care Plan Meeting (CPM) On 4/23/15, at approximately 1:10 note usually. He stated that Resid December, 2014. He reviewed the	articipate in the planning or revision of the resident's S HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY frew the facility did not ensure that the resident's representative w ies. Specifically Resident #76's spouse was not invited to particip is evident for 1 out of 28 residents reviewed for Quality of Care of the potential for more than minimal harm that was not immediate with [DIAGNOSES REDACTED]. She has long term and short to ily Living, was interviewed, He stated that he visits almost every day and he d not receive a letter or a phone call inviting him to the CPM, and there was no documented evidence that the Social Worker in held in December, 2014. My the Social Worker Assistant was interviewed and stated that I ent #76's husband visits daily and he invited him in person to the c chart for his documentation but did not present documented evi- person. Additionally, there is no note written by any other assign	** as allowed to provide ate in the care oncerns. ecopardy, erm memory problems and is was never invited to the evited the resident's husband he sends a letter or writes a e annual CCP in dence that he wrote a note or			
F 0371	Store, cook, and serve food in a	safe and clean way				
Level of harm - Minimal harm or potential for actual harm Residents Affected - Many	Specifically, two refrigerators, or tray line was not in proper workin This resulted in no actual harm withe findings are: Observed at 4/20/15 tour during in refrigerator is 70 degrees. The no Another observation at 4/20/15 at An interview with the Food service that she put the nourishments and was not in use and they put the no Interviewed Food Service Directed morning, her early supervisor cal dairy/milk refrigerator should be The Food Service Supervisor was	erviews the facility did not ensure that food was stored under san that stores cartons of milk and one refrigerator that was used for a conditions. This had the potential to affect the whole facility, the potential of more than minimal harm, and the potential of more than minimal harm, and the potential of more than minimal harm. It is a round 10:30 am the dairy/milk refrigerator temperature unishments in the prep refrigerator was cool to touch. 12:35 PM dairy/milk refrigerator is at 48 degrees and milk was 4 to worker was immediately conducted after observation on initial milk containers in the prep refrigerator about a half an hour ago. For (FSD) on 4/20/15 after initial tour, who stated that she did not cled out and she did not have an opportunity to check the temperature around 38 to 40 degrees and the staff keeps coming in and out of interviewed on 4/20/15 at 12:40 PM. The supervisor stated he charted the staff that possibly the total degrees. The supervisor continued to state that possibly the	e at 48 degrees and prep 7-48 degrees. tour of 4/20/15, who stated The prep refrigerator check temperatures this ures. The FSD stated the the refrigerator. ecked last night the prep			

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

PRINTED:12/18/2017

CENTERS FOR MEDICARE	& MEDICAID SERVICES		FORM APPROVED OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION	(XI) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUCTION A. BUILDING B. WING	(X3) DATE SURVEY COMPLETED 04/23/2015
NAME OF PROVIDER OF SU	335505 PPLIER	STREET ADDRE	SS, CITY, STATE, ZIP
HIGHLAND CARE CENTER	RINC	91 31 175TH STR JAMAICA, NY 1	
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing home or the state surve	
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF D OR LSC IDENTIFYING INFORM	DEFICIENCIES (EACH DEFICIENCY MUST BE P MATION)	RECEDED BY FULL REGULATORY
F 0371	(continued from page 1)	ice company at this time to check out refrigerators.	The refrigerators temperatures
Level of harm - Minimal harm or potential for actual harm	should be between 38 to 40 degre	es. The FSD tested the milk and it was initially at 48 ed the appropriate temperature is 40 degrees,	
Residents Affected - Many F 0441	Have a program that investigate	s, controls and keeps infection from spreading.	
Level of harm - Minimal harm or potential for actual harm	followed. Specifically,1) two (2) during the Medication Pass. 2) Fa	iew and interviews the facility did not ensure that Int Licensed Practical Nurses (LPN) were observed as the cility staff were observed not properly wearing a fac	hey performed hand hygiene incorrectly
Residents Affected - Few	unit, in accordance with New Yor This resulted in no actual harm wi	k State Flu Mask Regulation of July 2013. th potential for more than minimal harm.	·
Residents Affected - Few	The finding is: 1) During the Medication pass on Resident #325. At approximately she applied soap and water to her and dried them with a brown pape On 4/21/15 at approximately 4:18 with friction and soap for 15 to 20 nd 4/21/15 at approximately 4:30 resident #197 with a hand held in running water after administering using friction, then inmediately Pon 4/21/15 at approximately 4:41 15-20 seconds under running water after administering chiefiction. Then rinse her hands und 2) On 4/20/15, at approximately 1:5th floor while serving residents of below nose upon exiting the Diniobserved as she went into DR, ret below her nose. On 4/20/15 at approximately, 12:1 she did not receive the flu shot and that the residents are located. On 4/22/15, at approximately 11:1 Therapy area towards the elevator. On 4/22/15 at approximately 11:1 therapy area towards the elevator. On 4/22/15, at approximately 11:2 because he did not receive the flu areas that the residents are in. He On 4/22/15, at approximately 2:29 her chin. On 4/22/15 at approximately 2:29 her chin. On 4/22/15 at approximately of Influenza by health care worke persons are defined as healthcare utilized by patients. Such individithe facility's policy on hand wash hands with the volume of the ham	4/21/15 at approximately 4:04 PM, LPN #1 was obseved as she washed he hands and briefly used friction. She immediately place towel. She stated to this State Agent that she wash PM LPN #1 was interviewed and stated that she wash PM LPN #1 was interviewed and stated that she did seconds and then rinse with running water. PM LPN #2 was observed as she administered oral rhaler. At approximately 4:34 PM: the LPN was obse medications. She was observed as she washed her halt them under the running water and dried them will PM LPN #2 was interviewed and stated that she did er. She stated that she is required to wash her hands it running water, dry them with a paper towel and to 2:04 PM a CNA was observed wearing a blue face in turing lunch. At approximately 12:15 PM, the CNA mag Room to dispose of used food and paper items on noved another resident's tray from the table and exist the supposed to wear the mask when she enters the lower of the state	er hands. This State Agent observed her as acced her hands under the running water ed her hands for 15 seconds. mot realize that she must wash her hands medication (pills) and assisted the rved as she performed hand hygiene under ands briefly with soap and water while h a brown paper towel. In't wash her hands using friction for for 15-20 seconds using soap and urn off the water using a paper towel. mask below her nose while in Dining Room on was observed wearing the blue mask a resident 's tray. Lastly, she was ited the DR with the blue paper mask was wearing the mask. The CNA stated that building and when she is in an area from the hallway outside the Physical ow his nose and there were several stated that he is supposed to wear the mask hat he is to wear the blue paper mask below and stated that she did not receive the se is to reduce the risk of transmission inst the influenza virus. 1. Covered ith residents, and or areas highly ne procedures: b. Thoroughly cover the er and vigorously rub hands together for at

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

			OMB NO, 0938-0391
DEFICIENCIES AND PLAN OF	XI) PROVIDER / SUPPLIER CLIA DENNTIFICATION JUMBER	(X2) MULTIPLE CONSTRUCTION A. BUILDING B. WING	(X3) DATE SURVEY COMPLETED 05/16/2016
NAME OF PROVIDER OF SUPPI	35505 TER	ETREET A	DDRESS, CITY, STATE, ZIP
HIGHLAND CARE CENTER IN		91 31 175T	TH STREET
For information on the nursing hor	ne's plan to correct this deficience	ey, please contact the nursing home or the state	te survey agency.
	JMMARY STATEMENT OF D R LSC IDENTIFYING INFORM		T BE PRECEDED BY FULL REGULATORY
Level of harm - Minimal harm or potential for actual harm Residents Affected - Few Residents Affected - Few The did do	Make sure that each resident esident's entire drug/medication's entire drug/medication's entore drug/medication's entore as a construction of the	s drug regimen is free from unnecessary den is managed and monitored to achieve hig S HAVE BEEN EDITED TO PROTECT COWN and interview, the facility did not ensure direction of the property of the property of the state of the property of th	ghest well being. NoFIDENTIALITY** hat a resident's drug regimen was free from treceive gradual dose reductions to see if esidents reviewed for unnecessary atted the resident had severely impaired cognition and esident had [DIAGNOSES REDACTED]. The MDS of the most of the diagnostic service was observed. It was observed in the day room sitting in a lors were observed. It was observed observed. It was observed for conditionally drug use dated gait disturbance, movement disorder, and tardive ests of medications; Administer medications as on; Observe for changes in function status. It this 76 y/o (year old), for psychiatric instory of dementia, GERD, DM, [MEDICAL CONDITION]. Patient complained of seeing bugs on dotationally dependent of the observation of the ob

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

CENTERS FOR WEDICARE	CONTENTANT BEKVICES		OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION	(X1) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUCTION A. BUILDING B. WING	(X3) DATE SURVEY COMPLETED 05/16/2016
	335505		
NAME OF PROVIDER OF SU HIGHLAND CARE CENTER		STREET ADDR 91 31 175TH SI JAMAICA, NY	
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing home or the state sur	vey agency.
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I OR LSC IDENTIFYING INFOR	DEFICIENCIES (EACH DEFICIENCY MUST BE MATION)	PRECEDED BY FULL REGULATORY
F 0329	(continued from page 1)	and the state of t	A AC TO LO
Level of harm - Minimal harm or potential for actual harm	last week on 5/13/16. The Psychi impaired. He stated the resident v Depression. The	chiatrist was interviewed and stated that he received atrist stated he relies on staff for information becat was on [MEDICATION NAME] for [MEDICAL C	use the resident is cognitively CONDITION and [MEDICATION NAME] for
Residents Affected - Few	NAME] be	esident had been stable, he recommended the [ME orks as a consultant and because of the volume of a	
		be assessed and re-evaluated. He further stated he	
F 0431	Maintain drug records and proj to accepted professional standa	perly mark/label drugs and other similar productds.	cts according
Level of harm - Minimal harm or potential for actual harm	Based on observation, record revi not expired. Specifically, Heparin The findings are:	iew and interview, the facility did not ensure that m i IV flush syringes were expired. This was evident	nedications and other biologicals were for 1 of 8 units (Unit #8),
Residents Affected - Few	On 5/10/16 at approximately 10:0 box contained three Monoject Pre Lock were observed with a manu On 5/10/16 at 11:00 am, the Regi using it here, I do not know why Coordinator is responsible for che On 5/12/16 at 10:00 am, the RN It stated that any expired items shot	0 am, the 8th floor nursing unit medication storage filled Syringes containing Heparin 50 USP Units a facturer's expiration date of 02/2016. Stered Nurse (RN) manager was interviewed and s these drugs are still kept in emergency box. The Recking the emergency box for contents and expiration. Service Coordinator was interviewed and stated ald be returned to the Director of Nursing's office w	5 ml (milliliters) Heparin Lock Heparin tated we don't use them anymore. No one is N Manager further stated the In-Service on dates. that there was a mistake, she further
	Heparin flushes left on the nursin maybe it was mixed up with other The Policy and Procedure dated F	or of Nursing was interviewed and stated that she is gunits. She further stated that the facility does not r stock medications. ebruary 2016 documented that a list of contents wichecks the list on the box for expired medications.	use Heparin flushes anymore, and th expiration dates is maintained outside

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

		-4		OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES	(X1) PROVIDER / SUPPLIER / CLIA	(X2) MULTIPLE CONSTRUCT	ION	(X3) DATE SURVEY
DEFICIENCIES	CLIA	A. BUILDING		COMPLETED
AND PLAN OF CORRECTION	IDENNTIFICATION NUMBER	B. WING		08/04/2017
CONCECTION				
TALL OF BROTHDER OF SIT	335505		because a papagod dieni de	A STE STE
NAME OF PROVIDER OF SU			STREET ADDRESS, CITY, ST.	ATE, ZIP
HIGHLAND CARE CENTER	RINC		91 31 175TH STREET	
			JAMAICA, NY 11432	
	home's plan to correct this deficien			
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I	DEFICIENCIES (EACH DEFICIE	ENCY MUST BE PRECEDED B	Y FULL REGULATORY
	OR LSC IDENTIFYING INFORM No Health Deficiencies Found	MATION)		
	No Health Deficiencies Found			
	1			
l I				
	l			
	l			
			*	
LABORATORY DIRECTOR'S	OR PROVIDER/SLIPPLIER	TITLE	(X6) D	ATE
REPRESENTATIVE'S SIGNA	TURE	IIIDD	(2KO) D.	

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

			OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES	(X1) PROVIDER / SUPPLIER	(X2) MULTIPLE CONSTRUCTION	(X3) DATE SURVEY COMPLETED
DEFICIENCIES	I/ CLIA	A. BUILDING	
AND PLAN OF CORRECTION	IDENNTIFICATION NUMBER	B. WING	02/04/2015
Condition			
NAME OF PROVIDER OF SUI	335449	OTDEET ADI	DRESS, CITY, STATE, ZIP
ACHIEVE REHAB AND NU	RSING FACILITY	170 LAKE ST LIBERTY, N	REET V 12754
E i C	handa da da anno della deficion	cy, please contact the nursing home or the state s	
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I OR LSC IDENTIFYING INFORI	DEFICIENCIES (EACH DEFICIENCY MUST E	BE PRECEDED BY FULL REGULATORY
	No Health Deficiencies Found	WATION	
	No ricalin Denoicieles i build		
LABORATORY DIRECTOR'S REPRESENTATIVE'S SIGNA	S OR PROVIDER/SUPPLIER TURE	TITLE	(X6) DATE

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

				OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION	(X1) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUCT A, BUILDING B. WING	ION	(X3) DATE SURVEY COMPLETED 03/22/2016
U.A. C. D. D. O. U.D. C. O.	335449		OTHERT ADDRESS CHEV STA	TC 710
NAME OF PROVIDER OF SUI ACHIEVE REHAB AND NUI			STREET ADDRESS, CITY, STA 170 LAKE STREET LIBERTY, NY 12754	ile, ZiP
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing hom	e or the state survey agency.	
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF D OR LSC IDENTIFYING INFORM	DEFICIENCIES (EACH DEFICIE MATION)	NCY MUST BE PRECEDED BY	FULL REGULATORY
F 0431		perly mark/label drugs and other	r similar products according	
Level of harm - Minimal harm or potential for actual harm	to accepted professional standar Based on observations and intervi- and 1-West), it was determined the standards of practice and manufac	ews conducted during review of the	ne medication storage in two of for not dated when opened, in accord	ur nursing units (1-East lance with acceptable
Residents Affected - Some	observed: 1-East: 0n 3/22/16 at approximately 1:00 written, neither on the bottle or or The unit medication Licensed Pra- have been labeled with the date w	PM, an opened multidose vial of I n the box, when the vial was openectical Nurse (LPN) was interviewed then it was opened and promptly d	Lantus insulin was noted to be oped: did this time and stated that the v	ened and there was no date
	Humalog insulin pen were noted containers when they were opene The unit medication LPN was inte provide any explanation as to wh	rviewed at 1:30 PM as to why the	pen or the vials were not dated w	or their respective

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

DEFICIENCIES AND PLAN OF CORRECTION	/ CLIA IDENNTIFICATION NUMBER	A. BUILDING B. WING	COMPLETED 06/05/2017		
	335449				
NAME OF PROVIDER OF SUP ACHIEVE REHAB AND NUR		STREET ADDRESS, CITY, S 170 LAKE STREET LIBERTY, NY 12754	TATE, ZIP		
For information on the nursing h	home's plan to correct this deficien	cy, please contact the nursing home or the state survey agency,			
	SUMMARY STATEMENT OF D OR LSC IDENTIFYING INFORM	DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED MATION)	BY FULL REGULATORY		
Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	Give notice to the resident before a room or roommate change. **NOTE- TERMS IN BRACKETS HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY** Based on interviews and record review conducted during a recertification survey, the facility did not inform 1 out of 31 residents (#65) reviewed during Stage 2 of the survey of a room change prior to the actual change on multiple occasions. The findings are: Resident #65 has [DIAGNOSES REDACTED]. The Quarterly Minimum Data Set (a resident assessment tool) of 5/8/17 revealed that the resident has no cognitive impairment with dulty decision making. The resident was interviewed on 5/31/17 at 11:00 AM during Stage 1 of the survey and stated that she was moved to another room twice without notice. The facility policy and procedure for Room(mate) Change revised on 11/2016 stated that the resident who is being moved and their representative will be promptly informed of the room change verbally prior to room being changed as well as the reason for the change. The resident and the resident's representative will be afforded the opportunity to make objections to the room change, which the facility will document and take into consideration, prior to a room change. The Social Worker (SW) was interviewed on 6/2/17 at 9:45 AM and stated that the room transfer process in the facility was that the resident is care planned for the room change, then it is documented in the medical record. The resident and the family are then informed of the room changes. Review of the resident's clinical record revealed no documented evidence of any discussion with the resident and the family about the multiple room and bed changes. The SW was unable to provide any documented evidence that a care plan was developed.				
F 0253	Provide housekeeping and main	tenance services.			
Level of harm - Potential for minimal harm Residents Affected - Some	Based on observation and intervie maintenance and housekeeping so out of 4 facility units (1 East and the aforementioned units were dis off the wall. The findings include: Inspection of the facility was cond 1. Unit 1 East Rooms 102, 103, 104, 105, 106, 10 conditions: peeling and or soiled a scuff marks and gouges in walls, 2. Unit 1 West The whitish gray protecting vinyl color discoloration. The affected 146 ad 150.	we conducted during a recertification survey, the facility did not e prvices were provided to maintain a sanitary, orderly and well-mat 1 West). It was determined that walls, doors, and baseboard trims colored, scraped and with scuff marks. The wall papers are soiled tucted on 5/31/17, 6/1/17 and 6/5/17 and the following conditions 107, 108, 109, 110, 111, 112, 113, 114, 117, and 118 exhibited on and discolored wall papers; scraped or peeling paint on walls and coverings on the lower ends of resident room doors that face the rooms included but were not limited to rooms 130, 132, 133, 134 erviewed on 6/5/17 at 1:10 PM regarding the above observations at and that if possible the door coverings will be cleaned by the head	intained environment in 2 sin multiple rooms in d, discolored and peeling swere observed: e or more of the following /or baseboard trims; hallway have blackish , 135, 136, 141, 144, 145, and stated that there were		
	Allow the resident the right to pa	articipate in the planning or revision of the resident's			
Level of harm - Potential for minimal harm . Residents Affected - Some	**NOTE- TERMS IN BRACKET Based on record review and interv for 1 out of 2 residents (#207) rev restore the residents' previous leve The findings are: Resident #207 is an [AGE] year of for physical and occupational therapy The initial comprehensive care pla related to impaired immobility. The achieve the written goal were to a symptoms of breakdown, i.e. redr The resident was assessed on the A that the resident showed no cogni locomotion; used a walker and wl incontinent of urine (defined in M The above comprehensive care pla improve the resident's level of cor The resident was subsequently ass incontinent of urine (defined in M continued to show no cognilive in The comprehensive care plan was	in for incontinence dated 1/12/17 revealed that the resident had us he goal for the resident was to not develop any pressure sore. The pply protective skin care, encourage fluid intake, monitor conditi	direvise the care plan nationence to potentially dimitted to the facility on [DATE] rinary incontinence interventions to on of skin for sign and of 1/19/17 which indicated inbulation and g; and was occasionally in incontinence or to urinary continence. resident had become frequently tone continent voiding), rolcomotion.		
	The Certified Nurse Aide (CNA # that the resident at times did ask to The Registered Nurse (RN) who c	pisodes, dides in the Care Tracker showed that the resident continued to be 1) assigned to the resident on the day shift was interviewed on 6/o be toileted and was not able to transfer from the bed independed ompleted the MDS assessments was interviewed on 6/5/17 at 10 and was asked to explain the assessment process and how it is line.	2/17 at 11:45 AM and stated ntly. 30 AM regarding the decline in		

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

CENTERS FOR MEDICARE	& MEDICAID SERVICES		FORM APPROVED OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION	(X1) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUCTION A. BUILDING B. WING	(X3) DATE SURVEY COMPLETED 06/05/2017
	335449	Name (Control of Control of Contr	
NAME OF PROVIDER OF SUI ACHIEVE REHAB AND NUI		STREET ADDRESS, (170 LAKE STREET LIBERTY, NY 12754	
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing home or the state survey ag	gency.
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I OR LSC IDENTIFYING INFORI	DEFICIENCIES (EACH DEFICIENCY MUST BE PREC MATION)	EDED BY FULL REGULATORY
F 0280 Level of harm - Potential for minimal harm Residents Affected - Some	medical or cognitive reasons for t	e checks and validates the Care Tracker then speaks to the decline noted. After this determination, the resident's one process was not implemented for the resident.	ne Nurse Manager to determine the care plan would be reviewed
F 0315	Make sure that each resident what catheter, and receive proper s	o enters the nursing home without a catheter is not gi ervices to prevent urinary tract infections and restore	iven
Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	Based on interview and record rev 1 out of 3 residents reviewed for extent possible. The findings are:	S HAVE BEEN EDITED TO PROTECT CONFIDENT iew, the facility did not ensure that the necessary treatme arinary incontinence (#207) to maintain or improve the red female with [DIAGNOSES REDACTED]. The resider	ent and services were provided to esident's continence level to the
	for physical and occupational therapy The initial comprehensive care plarelated to impaired immobility. The achieve the written goal were to a symptoms of breakdown, i.e. redr. The resident was assessed on the Athat the resident showed no cognilocomotion; used a walker and wincontinent of urine (defined in M. The above care plan revealed no stresident's level of continence. Nei The resident was subsequently assincontinent of urine (defined in M. continued to show no cognitive in The comprehensive care plan was the plan of care with no changes rurinary incontinence episodes. The documentation by the nurse at The Certified Nurse Aide (CNA # that the resident's level of continence iplan of care. The RN stated that medical or cognitive reasons for the medical or cognitive reasons for the comprehensive care plan of care.	r services, in for incontinence dated 1/12/17 revealed that the residence goal for the resident was to not develop any pressure spply protective skin care, encourage fluid intake, monitor	nt had urinary incontinence sore. The interventions to r condition of skin for sign and ent tool) of 1/19/17 which indicated arring, ambulation and r toileting; and was occasionally tinence or to improve the rry continence. that the resident had become frequently it aleast one continent voiding), alation or locomotion, and the plan was to continue ionale for the increase in ed to be currently incontinent. ed on 6/2/17 at 11:45 AM and stated dependently. If at 10:30 AM regarding the decline in vit is linked to the resident's lurse Manager to determine the
F 0371	Store, cook, and serve food in a	safe and clean way	
Level of harm - Minimal harm or potential for actual harm Residents Affected - Some	hat the dishwasher registered the food-borne illness. The findings are: An initial inspection of the kitcher The temperature gauge of the disk cycle of the machine for proper se below 150 degrees F. The dietary the temperature of the dishwashin this procedure and directed the su stated that he had not taken the temperature of the dishwashing machine tempers 75/30/17 prior to the above observe. The acting Food Service Director test the temperature. This test strip degrees F which is the minimum in The AFSD then requested mainter. The maintenance director arrived into the machine. A follow-up visit to the kitchen was	(AFSD) was present at that time of the observation and we was designed to change from white to black when the to emperature required for hot water sanitization. The color	asher was observed to be in use, ures were being achieved for each not move and was recorded to be ewed at that time if she had checked the one responsible for doing was immediately interviewed and was recorded for the morning of was asked to use a test strip to emperature reached at least 180 or of the test strip did not change, the was allowing cold water to get eachfast dishes and utensils were
.41			

DEPARTMENT OF HEALTH AND HUMAN SERVICES CENTERS FOR MEDICARE & MEDICAID SERVICES

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

				OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES	(X1) PROVIDER / SUPPLIER	(X2) MULTIPLE CONSTRUCTION		X3) DATE SURVEY COMPLETED
DEFICIENCIES	(X1) PROVIDER / SUPPLIER / CLIA	(X2) MULTIPLE CONSTRUCTION A, BUILDING B. WING	1	COMPLETED
AND PLAN OF CORRECTION	IDENNTIFICATION NUMBER	B. WING		09/08/2016
Coldice Holy				
NAME OF PROVIDER OF SU	335726	етре	ET ADDRESS, CITY, STA	TE 210
		I		IE, ZIP
BEACON REHABILITATIO	N AND NURSING CENTER	ROCK	EACH 113TH STREET CAWAY PARK, NY 11694	be
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing home or th		
(X4) ID PREFIX TAG		EFICIENCIES (EACH DEFICIENCY N		EIII I BECLII ATODV
(A4) ID FREFIX TAG	OR LSC IDENTIFYING INFORM	MATION)	MOST DE LYECEDED DI	FULL REGULATOR I
	No Health Deficiencies Found			
	1			
				=

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

ta .			OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF	(X1) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUCTION A, BUILDING B, WING	(X3) DATE SURVEY COMPLETED 09/17/2014
CORRECTION			
NAME OF PROVIDER OF SUI	335726 PPLIER	STREET ADDRESS, CITY, ST	ATE, ZIP
BEACON REHABILITATION		140 BEACH 113TH STREET ROCKAWAY PARK, NY 116	KC.
For information on the nursing	home's plan to correct this deficien	cy, please contact the nursing home or the state survey agency.	
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I OR LSC IDENTIFYING INFORT	DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED E MATION)	IY FULL REGULATORY
F 0282 Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	**NOTE- TERMS IN BRACKET Based on observation, record revit a resident with a Pressure Ulcer (2 This was evident for one of three (Resident #18). Specifically, the [REDACTED]. The Finding is: Resident #18 was admitted to the REDACTED]. The resident was a The Minimum Data Set (MDS) as also documented that the resident On 9/17/14 at 8:30 AM the treatm observed to use a wound cleanser Alginate and cover with a Clean The physician's orders [REDACT On 9/17/14 at 10:00 AM the LPN she was using the right solution w locate Dakins solution on the unit On 9/17/14 at 10:30 AM the Assis was no Dakins solution available On 9/17/14 at 11:100 AM the Phart solution for Resident #18 since 5 On 9/17/14 at 11:15 AM the Direct receiving his supplies from Medic Dakins solution had been deliver the facility at this time. On 9/17/14 at 11:20 PM the Regist	EDJ. The physician's orders [REDACTED]. who rendered the treatment to Resident # 18 was interviewed. The when she used the wound cleanser instead of the Dakins solution. It where Resident # 18 resided, stant Director of Nursing Services (ADNS) completed a facility to in the facility, macist was interviewed. The Pharmacist stated that the pharmacy It /31/14. Iter of Nursing Services (DNS) was interviewed. The DNS stated care Part B. The DNS was unable to confirm this. The DNS further date the facility as stock on 7/1/14. The DNS also confirmed that the facility as stock on 7/1/14. The DNS also confirmed that the facility is stock on 7/1/14. The DNS also confirmed that the facility was unavailable for interview.	ty did not ensure that an to promote healing. o sample residents sician's orders dent has [DIAGNOSES y impaired cognition, The MDS Practical Nurse (LPN) was pack the wound with Calcium e LPN stated that she thought the LPN was unable to ar and confirmed that there had not delivered Dakins that perhaps Resident # 18 was r stated that 6 bottles of there was no Dakins in
F 0313 Level of harm - Minimal harm or potential for actual harm Residents Affected - Few	vision and hearing. **NOTE-TERMS IN BRACKET Based on record review and staff; receive proper treatment to maint total of 29 stage two residents. (R evening since July 2014, without The finding is: Resident # 137 was admitted to th REDACTED]. The Minimum Data Set (MDS) as also documented that the resident On 6/2/14 the physician ordered a The resident was seen by the Oph CONDITION] which was visuall; that the resident refused to have catare The Ophthalmologist recommend A physician's orders [REDACTEI] pressure in open-angle [MEDICAT The Medication Administration R order for [MEDICATION NAME] in the [MEDICATION NAME] in the [MEDICATION NAME] in the stated that by looking at the MAR the order. On 9/17/14 at 10:30 AM the phare [MEDICATION NAME] for interview. On 9/17/14 at 10:30 AM the phare [MEDICATION NAME] for reserted that both eyes were to be to Only 17/14 at 11:00 AM the Direct [MEDICATION NAME] for Reserted that both eyes were to be to Only 17/14 at 11:00 AM the Direct [MEDICATION NAME] should or if it was for both. On 9/17/14 at 12:15 PM the Regis that the [MEDICATION NAME] ever asked her for clarification as	thalmologist on 6/11/14. The consult documented that the resident y significant and that [MEDICAL CONDITION] was suspected. The act surgery, to start [MEDICATION NAME] PO 1% 1 drop (git) to both eyes only. The physician also ordered [MEDICATION NAME] PO 1%, a LL CONDITION] and Ocular Hypertension 1 git to both eyes (ou). CONDITION] and Ocular Hypertension 1 git to both eyes (ou) coords (MAR) were reviewed from July 2014 through September [3] 0.005% one git into infected eye daily in the evening. It on the second floor assigned to Resident # 137 was interviewed, the medication cart or the refrigerator in the medication room for R indicated [REDACTED]. The LPN stated that she would have on the second floor was interviewed. The LPN stated that she worn NAME] was to be administered for Resident # 137. The LPN f(RN) Unit Manager to clarify the omission as to which eye the me Resident # 137 between 9/13/14 and 9/16/14 on the 3:00 PM-11:0 macy provider was interviewed. The Pharmacist stated that he had ident # 137 on 9/3/14. The Pharmacist further stated that the order	sure that residents iewed for vision in a drop (gtt) in the CTED]. I has [DIAGNOSES y impaired cognition, The MDS ar glasses. had dense [MEDICAL The consult further documented (ou) twice a day (BID), also used to reduce intraocular BID. 2014. The MARS documented the The LPN was unable to locate tesident # 137. The LPN further alled the Physician to clarify all of the properties of the would have dication was to be to the properties of the should have been clarified to not the order for the nine which eye was to be treated is interviewed. The RN stated is interviewed. The RN stated is interviewed. The RN stated in further stated that no one

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

DEPARTMENT OF HEALTH CENTERS FOR MEDICARE	Í AND HUMAN SERVICES & MEDICAID SERVICES			PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391
STATEMENT OF DEFICIENCIES AND PLAN OF CORRECTION	(XI) PROVIDER / SUPPLIER / CLIA IDENNTIFICATION NUMBER	(X2) MULTIPLE CONSTRUC A. BUILDING B. WING	TION	(X3) DATE SURVEY COMPLETED 09/17/2014
	335726		Towns and the second	
AME OF PROVIDER OF SU	JPPLIER ON AND NURSING CENTER		STREET ADDRESS, CITY,	
EACON REHABILITATIO	IN AND NURSING CENTER		140 BEACH 113TH STREE ROCKAWAY PARK, NY 1	1694
	home's plan to correct this deficien			
(X4) ID PREFIX TAG	SUMMARY STATEMENT OF I OR LSC IDENTIFYING INFOR	DEFICIENCIES (EACH DEFICI MATION)	ENCY MUST BE PRECEDED	BY FULL REGULATORY
F 0313	(continued from page 1)			
Level of harm - Minimal harm or potential for actual harm	415.12(3)(b)			
Residents Affected - Few				

PRINTED:12/18/2017 FORM APPROVED OMB NO. 0938-0391

STATEMENT OF DEPCHEER SUPPLIES (CLEAR AND WASHES) AND A COMPARED ON THE STATEMENT OF THE ST					OMB NO. 0938-0391
STREET ADDRESS, CITY, STATE, ZIP ### BEACON REHABILITATION AND NURSING CENTER #### BEACON REHABILITATION AND NURSING CENTER ###################################	DEFICIENCIES AND PLAN OF	CLIA IDENNTIFICATION NUMBER	À. BUILDING		COMPLETED
ROCKAWAY PARK, NY 11694	NAME OF PROVIDER OF SU		STREI	ET ADDRESS, CITY, STA	ATE, ZIP
(X4) ID PREFIX TAG SUMMARY STATEMENT OF DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) Keep the rate of medication errors (wrong drug, wrong dose, wrong time) to less than 5%. **NOTE- TERMS IN BRACKETS HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY** Based on observation, record review, and staff interviews during the recretification survey, the facility did not ensure that the medication error rate was less than 5%. This was evident for one of three residents observed for the administration of medications. Specifically, Resident #99 was observed to receive [MEDICATION NAME] (a diuretic medication) 20 milligrams (mg) by mouth (po) on the wrong day and [MEDICATION NAME] (an oral diabetic medication) two hours after the prescribed time resulting in an 8% medication error rate. Residents Affected - Few Residents Affected in P59 was admitted to the facility on [DATE] and has [DIAGNOSES REDACTED]. The Minimum Data Set (MDS) assessment dated [DATE] documented that the resident had a Brief Interview for Mental Status (BIMS) score of 14 indicating that the resident was cognitively intact. The current monthly physicians orders (REDACTED). The resident was also to receive [MEDICATION NAME] 10 mg pot twice a day. (BID) before meals for Diabetes Mellitus, A medication administration observation was conducted on 9/15/15 at 9:30 AM on the 3rd Floor unit where Resident #99 resided. The Medication Licensed Practical Nurse (LPN) was observed to dispense [MEDICATION NAME] 20 mg and [MEDICATION NAME] in addition to eight other medications, The LPN proceeded to sign for the [MEDICATION NAME] and [MEDICATION NAME] medications in the Electronic Medical Record (EMR) in the presence of the surveyor. Upon reconciliation of the medications for Resident #99, it was documented that the [MEDICATION NAME] had been signed for at 7:30 AM as prescribed, rather that 9:30 AM as avas administered. Additionally, it was noted that the [MEDICATION NAME] should not have been administered on 9/15/	BEACON REHABILITATION	N AND NURSING CENTER			
(X4) ID PREFIX TAG SUMMARY STATEMENT OF DEFICIENCIES (EACH DEFICIENCY MUST BE PRECEDED BY FULL REGULATORY OR LSC IDENTIFYING INFORMATION) Keep the rate of medication errors (wrong drug, wrong dose, wrong time) to less than 5%. **NOTE- TERMS IN BRACKETS HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY** Based on observation, record review, and staff interviews during the recertification survey, the facility did not ensure that the medication error rate was less than 5%. This was evident for one of three residents observed for the administration of medications. Specifically, Resident #99 was observed to receive [MEDICATION NAME] (a diuretic medication) 20 milligrams (mg) by mouth (po) on the wrong day and [MEDICATION NAME] (an oral diabetic medication) two hours after the prescribed time resulting in an 8% medication error rate. The finding is: Residents 499 was admitted to the facility on [DATE] and has [DIAGNOSES REDACTED]. The Mimimum Data Set (MDS) assessment dated [DATE] documented that the resident had a Brief Interview for Mental Status (BIMS) score of 14 indicating that the resident was cognitively intact, the current monthly physician's orders [REDACTED]. The resident was also to receive [MEDICATION NAME] 10 mg po twice a day. (BID) before meals for Diabetes Mellitus, A medication administration observation was conducted on 9/15/15 at 9:30 AM on the 3rd Floor unit where Resident #99 resided. The Medication Licensed Practical Nurse (LPN) was observed to dispense [MEDICATION NAME] 20 mg and [MEDICATION NAME] and addition to eight other medications. The LPN proceeded to sign for the [MEDICATION NAME] and [MEDICATION NAME] and addition to eight other medications for Resident #99, it was documented that the [MEDICATION NAME] as prescribed, rather that 9:30 AM as was administered. Additionally, it was noted that the [MEDICATION NAME] should not have been administered on 9/15/15. On 9/17/15 at 11/45 AM, the Rejistered Nurse (RN) Charge Nurse was interviewed and stated that the Pris should have been notifi	Contraction on the contract	h	145500000	OF THE PARTY OF TH	4
F 0332 Keep the rate of medication errors (wrong drug, wrong dose, wrong time) to less than 5%. **NOTE- TERMS IN BRACKETS HAVE BEEN EDITED TO PROTECT CONFIDENTIALITY** Based on observation, record review, and staff interviews during the recertification survey, the facility did not ensure that the medication error rate was less than 5%. This was evident for one of three residents observed for the administration of medications. Specifically, Resident #99 was observed to receive [MEDICATION NAME] (a diurctic medication) 20 milligrams (mg) by mouth (po) on the wrong day and [MEDICATION NAME] (an oral diabetic medication) two hours after the prescribed time resulting in an 8% medication error rate. The finding is: Resident #99 was admitted to the facility on [DATE] and has [DIAGNOSES REDACTED]. The Minimum Data Set (MDS) assessment dated [DATE] documented that the resident had a Brief Interview for Mental Status (BIMS) socre of 14 indicating that the resident was cognitively intact. The current monthly physician's orders [REDACTED]. The resident was also to receive [MEDICATION NAME] 10 mg po twice a day (BID) before meals for Diabetes Mellitus, A medication administration observation was conducted on 9/15/15 at 9:30 AM on the 3rd Floor unit where Resident #99 resided. The Medication Licensed Practical Nurse (LPN) was observed to dispense [MEDICATION NAME] and [MEDICATION NAME] and addition to eight other medicationss for Resident #99, it was documented that the [MEDICATION NAME] had been signed for at 7:30 AM as prescribed, rather that 9:30 AM as was administered that the [MEDICATION NAME] had been signed for at 7:30 AM as prescribed, rather that 9:30 AM as was administered that the [MEDICATION NAME] should not have been administered on 9/15/15. On 9/17/15 at 10:45 AM, the LPN Medication Nurse who completed the administration on 9/15/15 was interviewed. The LPN should have notified her when the [MEDICATION NAME] had been administered in error. The RN stated that the LPN should have been written and the Phy	×		£-1		ZELLI DECLILATODY
Level of harm - Minimal harm or potential for actual harm or potential harm or potential for actual harm or potential harm or potential for actual harm or potential for actual harm or potential harm or pot	(A4) ID PREFIX TAG			WOST DE PRECEDED DI	FULL REGULATORY
	Level of harm - Minimal harm or potential for actual harm	Keep the rate of medication erre**NOTE- TERMS IN BRACKET Based on observation, record revithat the medication error rate was administration of medications. Sp. 20 milligrams (mg) by mouth (po prescribed time resulting in an 89. The finding is: Resident #99 was admitted to the The Minimum Data Set (MDS) as (BIMS) score of 14 indicating the The current monthly physician's o day (BID) before meals for Diabetes I A medication administration observations. The Medication Licensed Practica NAME I 0 mg in addition to eight other medicat medications in the Electronic Me Upon reconciliation of the medica 7:30 AM as prescribed, rather the should not have been administer On 9/17/15 at 10:45 AM, the LPN dat when she went to administer Additionally, the LPN acknowled On 9/17/15 at 11:45 AM, the Reg notified her when the [MEDICA1 NAME] was administered two he have been written and the Physici On 9/17/15 at 12:00 PM, the Assi her when the errors occurred. The necessary inservices.	ors (wrong drug, wrong dose, wrong the SHAVE BEEN EDITED TO PROTECT and staff interviews during the recert less than 5%. This was evident for one of secifically, Resident #99 was observed to 1 on the wrong day and [MEDICATION 6 medication error rate.] facility on [DATE] and has [DIAGNOSE sessment dated [DATE] documented that the resident was cognitively intact, rders [REDACTED]. The resident was all this was all the session was conducted on 9/15/15 at 9:30. If Nurse (LPN) was observed to dispense ions, The LPN proceeded to sign for the dical Record (EMR) in the presence of the tions for Resident #99, it was documented 19:30 AM as was administered. Addition of medication Nurse who completed the act the [MEDICATION NAME] at 7:30 AM ged that the [MEDICATION NAME] at 7:30 AM ged that the [MEDICATION NAME] at 1:30 AM Mas Jack and Stered Nurse (RN) Charge Nurse was intron NAME] had been administered in ours late, the LPN should have also inforr an should have been notified.	T CONFIDENTIALITY** iffication survey, the facility of three residents observed i receive [MEDICATION N NAME] (an oral diabetic n est REDACTED]. It the resident had a Brief In lso to receive [MEDICATION OAM on the 3rd Floor unit of MEDICATION NAME] a resurveyor, of that the [MEDICATION OR HEDICATION NAME] a resurveyor, of that the [MEDICATION OR HEDICATION NAME] a resurveyor, of that the [MEDICATION OR HEDICATION NAME] a resurveyor, of that the [MEDICATION OR HEDICATION NAME] a resurveyor, or the stated that the [or of 15/15 the resident we receive the receive the receive the receive the resident wa receive the resident wa receive the receive the receive the receive the resident wa receive the receive th	y did not ensure for the for the AME] (a diuretic medication) medication) two hours after the atterview for Mental Status ON NAME] 10 mg po twice a where Resident #99 20 mg and [MEDICATION and [MEDICATION NAME] NAME] had been signed for at [MEDICATION NAME] as interviewed. The LPN stated as in the process of dressing. stered on 9/15/15. nat the LPN should have if that when the [MEDICATION if a Progress Note should he RN should have informed

LABORATORY DIRECTOR'S OR PROVIDER/SUPPLIER REPRESENTATIVE'S SIGNATURE

TITLE

(X6) DATE

Facility	City	State	Beds	Date
Boro Park Center for Rehabilitation & Health Care	Brooklyn	New York	510	6/1/2013.
Brooklyn Center for Rehabilitation and Residential Health Ca	Brooklyn	New York	215	6/1/2013
Bushwick Center for Rehabilitation and Health Care	Brooklyn	New York	225	6/1/2013
Richmond Center for Rehabilitation and Specialty HealthCare	Staten Island	New York	300	6/1/2013
Holliswood Center for Rehabilitation and Healthcare	Queens	New York	314	6/1/2013
Bronx Center for Rehabilitation & Health Care	Bronx	New York	200	6/1/2013
Hope Center for HIV and Nursing Care	Bronx	New York	66	4/1/2015
Triboro Center for Rehabilitation and Nursing	Bronx	New York	515	6/1/2013
University Center for Rehabilitation and Nursing	Bronx	New York	46	6/1/2013
Williamsbridge Center for Rehabilitation and Nursing	Bronx	New York	77	6/1/2013
Northern Metropolitan Residential Health Care Facility	Rockland County	New York	120	6/1/2013
Northern Riverview Health Care Center	Rockland County	New York	182	6/1/2013
Northern Manor Geriatric Center	Rockland County	New York	231	6/1/2013
Ellicott Center for Rehabilitation and Nursing	Buffalo	New York	160	6/1/2013
Buffalo Center for Rehabilitation and Nursing	Buffalo	New York	200	6/1/2013
Corning Center for Rehabilitation and Healthcare	Corning	New York	120	6/1/2013
Steuben Center for Rehabilitation and HealthCare	Bath	New York	105	6/1/2013
Ontario Center for Rehabilitation and Healthcare	Canandaigua	New York	98	11/1/2014
Fulton Center for Rehabilitation and Health Care	Gloversville	New York	176	6/1/2013
Essex Center for Rehabilitation and HealthCare	Elizabethtown	New York	100	1/1/2014
Warren Center for Rehabilitation and Healthcare	Queensbury	New York	80	6/1/2013
Granville Center for Rehabilitation and Nursing	Granville	New York	120	1/15/2014
Washington Center for Rehabilitation and HealthCare	Argyle	New York	122	2/1/2014
Hammonton Center for Rehabilitation and Healthcare	Hammonton	New Jersey	220	6/1/2013
Deptford Center for Rehabilitation and Healthcare	Deptford	New Jersey	240	6/1/2013

Census	Birchwood	Achieve	Beacon	Highland
16-Oct	0.948700717	0.879493088	0.983602151	0.981854839
16-Nov	0.924768519	0.875	0.980555556	0.956770833
16-Dec	0.950044803	0.936635945	0.983602151	0.9625
17-Jan	0.959453405	0.919124424	0.953225806	0.975
17-Feb	0.887400794	0.930867347	0.977380952	0.983928571
17-Mar	0.875224014	0.923963134	0.960215054	0.974495968
17-Apr	0.857638889	0.893571429	0.936111111	0.951458333
17-May	0.857750896	0.910368664	0.919623656	0.950806452
17-Jun	0.846550179	0.92047619	0.938611111	0.946666667
17-Jul	0.847670251	0.893087558	0.977956989	0.962903226
18-Aug	0.866263441	0.939170507	0.985483871	0.949697581
17-Sep	0.900925926	0.945	0.9875	0.93875
17-Oct	0.890456989	0.928110599	0.929569892	0.940826613
17-Nov	0.886805556	0.893095238	0.956666667	0.947395833



Admission

POL: 600-01

Release Date:	Category	
11/28/2016	Policy	Page 1 of 2
Original Date:	Subcategory	1 490 1 612
09/26/2003	Patient Care	

Facility Specific Addendum Attached - Review All of Policy and Addendum Pages

(Check if State Specific and/or Facility Specific Policy Addendum is attached)

POLICY

The patient is admitted without prejudice to payment source. The patient is not required to waive their rights to Medicaid or Medicare nor are they required to give assurance that they will not apply for Medicaid or Medicare.

COMPONENTS:

- 1. The patient is not treated differently based on payment source.
 - a. Room assignment is based on medical and nursing needs.
 - b. Care and services are based on individual needs.
- 2. A third person (family, surrogate, or legal representative) cannot be required to guarantee payment for the patient's stay; however, the third person may use the patient's funds to pay for care.
- 3. Medicaid-eligible patients are not required to make additional payments (e.g., deposit or a promise to pay private rates for a specified period) as a prerequisite to admission, expedited admission, or to continue the stay.
- 4. The facility discloses and provides to a patient or potential patient prior to time of admission, notice of special characteristics or service limitations of the facility.
- 5. If the facility is a composite distinct part, the facility discloses in its admission agreement its physical configuration, including the various locations that comprise the composite distinct part, and must specify the policies that apply to room changes between its different locations.
- 6. The Medicaid-eligible patient may be charged for services not covered under the state plan only if:
 - a. The patient requests and receives the service,
 - b. The patient is notified in advance that the service is not covered and the charge for the service, and
 - c. The patient's request for the service is not a condition for admission or continued stay.
- 7. Services and supplies are covered or not covered in the basic daily rates are addressed in the patient's admission agreement. A list of optional items and services with related charges will be provided upon admission.
- 8. Circumstances for terminating services or supplies may include, but are not limited to:
 - a. Patient no longer needs services or supplies
 - b. Physician has discontinued services or supplies
 - c. Patient refuses services or supplies
 - d. Services or supplies are no longer appropriate for the patient's care
 - e. Services or supplies are not covered by Medicare, Medicaid or private insurance



Admission

POL: 600-01

Release Date:	Category	
11/28/2016	Policy	Page 2 of 2
Original Date:	Subcategory	, ago = o. =
09/26/2003	Patient Care	

9. Charitable, religious, or philanthropic contributions from an organization or from a person unrelated to a Medicaid-eligible patient are acceptable as long as the contribution is not a condition of admission, expedited admission, or continued stay of a Medicaid-eligible patient.

THERAPEUTIC LEAVE/OUT-OF-CENTER ON PASS

- 1. Upon admission, the family/responsible party and patient are notified of criteria necessary for patient to leave the center on therapeutic leave/out-of-center on pass:
 - a. 72 hours notice prior to leaving the center to allow time for medication preparation:
 - 1) Obtaining prescriptions, if applicable
 - 2) Pharmacy to prepare and send patient medications appropriately labeled for pass
 - b. Failure to give the center appropriate notification can result in not obtaining the patient's medications. New prescribtions may need to be called in to family's local pharmacy by the attending physician. Family is responsible for obtaining and paying for any prescriptions.
 - c. Family is accountable for all medication including any controlled substances.
 - d. Patient's nurse and family are responsible for validating the medication count when leaving and returning to the center.
 - e. Family understands that center is responsible for the safety of the patient. Missing or unaccounted for medications or medications dosages exceding the prescribed dose will be investigated.
 - f. Missing or incorrect controlled substance counts is reportable to the local law enforcement agency and the Federal Drug Enforcement Agency.



Reference: CMS 42§483.12(d)(1)